
FORM 10-Q

UNITED STATES

SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

(MARK ONE)

1

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 FOR THE QUARTER ENDED SEPTEMBER 30, 1997 OR

[] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission file number 1-12317

NATIONAL-OILWELL, INC. (Exact name of registrant as specified in its charter)

DELAWARE

76-0475815

(State or other jurisdiction (I.R.S. Employer of incorporation or organization) Identification No.)

> 5555 SAN FELIPE HOUSTON, TEXAS 77056

(Address of principal executive offices)

(713) 960-5100

(Registrant's telephone number, including area code)

Indicate by check mark whether the registant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

YES X NO

As of November 5, 1997, 25,486,170 common shares were outstanding, assuming the exchange on a one-for-one basis of all Exchangeable Shares of Dreco Energy Services Ltd. into shares of National-Oilwell, Inc. common stock.

ITEM 1. FINANCIAL STATEMENTS

NATIONAL-OILWELL, INC. CONSOLIDATED BALANCE SHEETS (IN THOUSANDS, EXCEPT SHARE DATA)

	September 30, 1997	
ASSETS	(Unaudited)	
Current assets: Cash and cash equivalents Receivables, less allowance of \$2,915 and \$3,044 Inventories Deferred income taxes Prepaid and other current assets	\$ 27,730 180,000 177,117 2,925 5,365	139,186 128,611 4,028
	555, 157	200,000
Property, plant and equipment, net Deferred income taxes Goodwill Other assets	68,499 10,906 25,041 519 \$ 498,102 ========	6,327 1,499 \$ 352,518
LIABILITIES AND OWNERS' EQUITY		
Current liabilities:		
Current portion of long-term debt Accounts payable Customer prepayments Accrued compensation	96,482 22,908 10,204	\$ 6,100 87,461 5,889 9,778
Accrued income taxes Other accrued liabilities	10,928 22,746	3,439 11,835
other accrued manifilies	22,740	
	164,653	124,502
Long-term debt Insurance reserves Deferred income taxes Other liabilities	67,596 6,078 2,254 8,721	
	249,302	183,502
Commitments and contingencies		
Stockholders' equity: Common stock - par value \$.01; 25,328,021 shares and 23,543,717 shares issued and outstanding at September 30, 1997 and December 31, 1996	253	235
Additional paid-in capital Cumulative translation adjustment Retained earnings	198,350 (5,027) 55,224	149,497 (2,302) 21,586
	248,800	169,016
	\$ 498,102 ======	\$ 352,518 ======

The accompanying notes are an integral part of these statements.

NATIONAL-OILWELL, INC. CONSOLIDATED STATEMENTS OF OPERATIONS (UNAUDITED) (IN THOUSANDS, EXCEPT PER SHARE DATA)

		Three Months Ended September 30,		Nine Months Ended September 30,	
		1997	1996	1997	1996
Revenues		\$ 264,959	\$ 200,773	\$ 705,719	\$ 561,023
Cost of r	evenues	209,607	166,015	568,689	472,808
Gross pro	fit	55,352	34,758	137,030	88,215
Selling,	general and administrative	26,650	21,356	73,690	57,547
Special c	harge	10,660		10,660	
Operating	income	18,042	13,402	52,680	30,668
Inter	ome (expense): est and financial costs est income	(1,768) 466 (25)	(3,413) 207 619	(4,803) 1,292 (194)	(10,167) 1,131 669
Income be	fore income taxes and extraordinary loss	16,715	10,815	48,975	22,301
Provision	for income taxes	7,268	3,948	18,764	9,243
Income be	fore extraordinary loss	9,447	6,867	30,211	13,058
Extraordi	nary loss net of income tax benefit of \$376	(623)		(623)	
Net incom	e	\$ 8,824 ======	\$ 6,867 ======	\$ 29,588 ======	\$ 13,058 ======
Weighted	average shares outstanding	26,067 =======	19,499 =======	25,886 ======	19,482 =======
Income pe	r share: Before extraordinary loss Extraordinary loss	\$0.36 (0.02)		\$ 1.17 (0.03)	\$ 0.67
	Net income	\$ 0.34 ======	\$ 0.35 ======	\$ 1.14 ======	\$0.67 ======

The accompanying notes are an integral part of these statements.

	Nine Months Ended September 30,	
	1997	
Cash flow from operating activities:		
Net income	\$ 29,588	\$ 13,058
Adjustments to reconcile net income to net cash provided (used) by operating activities:		
Depreciation and amortization	10,480	6,342
Provision for losses on receivables	(395)	531 1,720
Provision for deferred income taxes	(894)	1,720
Gain on sale of assets	(2,393)	(1,698)
Foreign currency transaction (gain) loss	115	(34)
Special charge	10,660	
Extraordinary loss, net of tax benefit	623	
Changes in operating assets and liabilities: Decrease (increase) in receivables	(20, 901)	(24 621)
Decrease (increase) in inventories	(30, 091)	(34, 621)
Decrease (increase) in prepaid and other current assets	(37,057) (1 157)	(2,176) (748) (3,900)
Increase (decrease) in accounts payable	3 779	(3 900)
Increase (decrease) in other assets/liabilities, net	8,945	3,428
		3,428
Net cash provided (used) by operating activities	(9,194)	(18,098)
Cash flow from investing activities:		
Purchases of property, plant and equipment	(19,462)	(10,780)
Proceeds from sale of assets	¥,081	2,490
Acquisition of businesses, net of cash acquired	(18,045)	
Acquisition of partnership, net of cash acquired		2,490 (106,248)
Other	(1,403)	(350)
Not each provided (used) by investing estivities		
Net cash provided (used) by investing activities	(34,829)	(114,888)
Cash flow from financing activities:		
Proceeds from revolving line of credit, net	45,095	(7,321)
Payments on long-term debt	(21, 359)	
Proceeds from issuance of common stock	37,700	30,749 103,378
Acquisition debt proceeds		103,378
Net cash provided (used) by financing activities	61,436	126,806
Effect of exchange rate losses on cash	(1,544)	
Increase (decrease) in cash and equivalents	15,869	(6,180)
Cash and cash equivalents, beginning of period	13,611	21,391
Change in cash during period to conform year end	(1,750)	(5,626)
Cash and cash equivalents, end of period	\$ 27,730	\$ 9,585
	=======	=======

The accompanying notes are an integral part of these statements.

1. BASIS OF PRESENTATION

Effective January 1, 1996, National-Oilwell, Inc. acquired National-Oilwell, a general partnership between National Supply Company, Inc., a subsidiary of Armco Inc., and Oilwell, Inc., a subsidiary of USX Corporation, and subsidiaries (the "Acquisition").

Effective September 25, 1997, National-Oilwell completed a combination ("Combination") with Dreco Energy Services Ltd. ("Dreco"). As a result of the Combination, each Dreco Class "A" common share ("Dreco Common Share") outstanding was converted into a Dreco Exchangeable Share at an exchange ratio of .9159 of an Exchangeable Share for each Dreco Common Share outstanding at September 25, 1997 and, accordingly, approximately 7.2 million Exchangeable Shares were issued.

Each Exchangeable Share is intended to have substantially identical economic and legal rights as, and will ultimately be exchanged on a one-for-one basis for, a share of National-Oilwell common stock. In addition, options to purchase Dreco Common Shares were converted into options to purchase shares of National-Oilwell common stock at the same exchange ratio.

The Combination has been accounted for as a pooling-of-interests and, accordingly, the consolidated financial statements of National-Oilwell and Dreco have been combined and all prior periods have been restated to give effect to the Combination. Information concerning common stock and per share data has been restated on an equivalent share basis and assumes the exchange of all Exchangeable Shares.

National-Oilwell has a year end of December 31 and, prior to the Combination, Dreco had a year end of August 31. The restated financial statements combine the December 31, 1996 balance sheet of National-Oilwell with the November 30, 1996 balance sheet of Dreco, while reflecting the combination of the September 30, 1997 balance sheet of each company. The income statement reflects the combination of the three months ended September 30, 1997 for each company, compared to the combination of the three months ended September 30, 1996 for National-Oilwell with the three months ended August 31, 1996 for Dreco. For the 1997 nine month period, the nine months ended September 30, 1997 for National-Oilwell is combined with the six months ended May 31, 1997 and the three months ended September 30, 1997 for Dreco. For the 1996 nine month period, the nine months ended September 30, 1996 for National-Oilwell is combined with the nine months ended August 31, 1996 for Dreco. As a result of conforming reporting periods subsequent to the Combination to a calendar quarter basis, the operating results for Dreco for the month of June 1997 were included in the Consolidated Statements of Stockholders' Equity. For June 1997, Dreco recorded revenues of \$13.4 million, net income of \$917,000 and net income per share of \$0.04.

Revenues and net income before special charges and extraordinary loss of the separate companies for the period preceding the combination were as follows (in thousands):

Nine Months Ended

	September 30,1997
Revenues	
National-Oilwell	\$593,415
Dreco	112,304
	\$705,719
	=======
Net income before special charges and extraordinary loss	
National-Oilwell	\$ 25,348
Dreco	12,991
	\$ 38,339
	=======

The financial statements included herein have been prepared by the Company, without audit, pursuant to the rules and regulations of the Securities and Exchange Commission and in accordance with generally accepted accounting principles. In the opinion of management, the information furnished reflects all adjustments, all of which are of a normal, recurring nature, necessary for a fair presentation of the results of the interim periods. It is recommended that these statements be read in conjunction with the restated consolidated financial statements and notes thereto for the three years ended December 31, 1996 to be included in a Current Report on Form 8-K. No significant accounting changes have occurred during the nine months ended September 30, 1997.

2. ACQUISITIONS

On December 2, 1996, Dreco acquired 100% of the issued and outstanding shares of Vector Oil Tool Ltd. ("Vector") for consideration of 389,000 Dreco Common Shares (as adjusted for the exchange ratio in the Combination) and cash consideration of \$1,481,000. This business involves the manufacture, sale, rental and service of downhole motors and other products. The transaction was accounted for using the purchase method and did not have a material effect on the Company's consolidated financial statements. Results of operations for Vector are included, by virtue of the Combination described above, in the Company's financial statements for reporting periods beginning in 1997.

On April 25, 1997, the Company purchased the drilling controls business of Ross Hill Controls and its affiliate, Hill Graham Controls Limited for approximately \$20 million in cash. This business involves the manufacture, sale and service of electrical control systems used in conjunction with drilling operations. The transaction was accounted for under the purchase method of accounting and did not have a material effect on the Company's financial statements.

On May 15, 1997, the Company acquired by merger 100% of the common stock of PEP, Inc., a manufacturer of petroleum expendable pump products. The Company issued 400,000 shares of common stock pursuant to the transaction which was recorded in accordance with the pooling-of-interests method of accounting. The transaction did not have a material effect on the Company's historical financial statements and financial statements prior to April 1, 1997 were not restated.

3. INVENTORIES Inventories consist of (in thousands):

	September 30, 1997	December 31, 1996
Raw materials Work in process Finished goods	\$ 17,381 23,747 135,989	\$ 12,854 8,367 107,390
	\$177,117 ========	\$128,611 =======

4. STATEMENTS OF CASH FLOWS

The following information is provided to supplement the Statements of Cash Flows (in thousands):

	Nine Months End	ed September 30,
	1997	1996
Cash paid during the period for:		
Interest	\$ 2,629	\$ 6,066
Income taxes	12,225	2,429

5. LONG-TERM DEBT

On September 25, 1997, the Company entered into a new five-year unsecured \$125 million revolving credit facility (the "New Credit Facility") that was used in part to repay in full amounts outstanding under the previous credit facility as well as the seller notes payable, together with deferred interest, incurred in connection with the Acquisition. The balance of the New Credit Facility will be available for acquisitions and general corporate purposes. The New Credit Facility provides for interest at prime or LIBOR plus 0.5%, subject to adjustment based on the Company's Capitalization Ratio, as defined. The New Credit Facility contains financial covenants and ratios regarding minimum tangible net worth, maximum debt to capital and minimum interest coverage.

6. SPECIAL CHARGE

In connection with the Combination described in Note 1 above, National-Oilwell incurred one-time combination expenses of \$10.7 million (\$8.1 million net of income taxes, or \$0.31 per share) related to various professional fees and integration costs.

7. EXTRAORDINARY LOSS

The replacement of the previous credit facility by the New Credit Facility described in Note 4, resulted in the write-off of \$1.0 million (\$0.6 million after tax) in deferred financing costs related to the replaced agreement.

8. COMMON STOCK

On December 2, 1996, the Company issued 389,000 Dreco Common Shares (as adjusted for the exchange ratio in the Combination) in connection with the acquisition of Vector Oil Tool Ltd. (see Note 2). On December 10, 1996, the Company completed the sale of 1,053,000 Dreco Common Shares (as adjusted for the exchange ratio in the Combination) for total net proceeds of \$39.2 million. As these transactions were completed by Dreco after November 30, 1996, the effect is included in the September 30, 1997 combined balance sheet but not in the December 31, 1996 combined balance sheet (see Note 1).

On October 10, 1997 the Company's Board of Directors declared a 2-for-1 stock split in the form of a one-for-one stock dividend payable on November 18, 1997 to stockholders of record on November 10, 1997. The financial statements included herein have not been restated to reflect the stock split.

6

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

INTRODUCTION

The Company is a worldwide leader in the design, manufacture and sale of machinery and equipment and in the distribution of maintenance, repair and operating ("MRO") products used in oil and gas drilling and production. The Company's revenues are directly related to the level of worldwide oil and gas drilling and production activities and the profitability and cash flow of oil and gas companies and drilling contractors, which in turn are affected by current and anticipated prices of oil and gas.

Effective September 25, 1997, National-Oilwell completed a combination (the "Combination") with Dreco Energy Services Ltd. ("Dreco"). As a result of the Combination, each Dreco Class "A" common share ("Dreco Common Share") outstanding was converted into a Dreco Exchangeable Share at an exchange ratio of .9159 of an Exchangeable Share for each Dreco Common Share outstanding at September 25, 1997 and, accordingly, approximately 7.2 million Exchangeable Shares were issued. Each Exchangeable Share is intended to have substantially identical economic and legal rights as, and will ultimately be exchanged on a one-for-one basis for, a share of National-Oilwell common stock.

The Combination has been accounted for as a pooling-of-interests and, accordingly, the consolidated financial statements of National-Oilwell and Dreco have been combined and all prior periods have been restated to give effect to the Combination. Information concerning common stock and per share data has been restated on an equivalent share basis and assumes the exchange of all Exchangeable Shares.

National-Oilwell has a year end of December 31 and, prior to the Combination, Dreco had a year end of August 31. The restated financial statements combine the December 31, 1996 balance sheet of National-Oilwell with the November 30, 1996 balance sheet of Dreco, while reflecting the combination of the September 30, 1997 balance sheet of each company. The income statement reflects the combination of the three months ended September 30, 1997 for each company, compared to the combination of the three months ended September 30, 1996 for National-Oilwell with the three months ended August 31, 1996 for Dreco. For the 1997 nine month period, the nine months ended September 30, 1997 for National-Oilwell is combined with the six months ended May 31, 1997 and the three months ended September 30, 1997 for Dreco. For the 1996 nine month period, the nine months ended September 30, 1996 for National-Oilwell is combined with the nine months ended August 31, 1996 for Dreco. As a result of conforming reporting periods subsequent to the Combination to a calendar quarter basis, the operating results for Dreco for the month of June 1997 were included in the Consolidated Statements of Stockholders' Equity. For June 1997, Dreco recorded revenues of \$13.4 million, net income of \$917,000 and net income per share of \$0.04.

During 1996 and 1997, the capital structure of the Company changed significantly. In January 1996, the Company acquired the operations of National-Oilwell (the "Acquisition"), resulting in the incurrence of significant amounts of debt and related interest expense. On October 29, 1996, the Company sold 4.6 million shares of its common stock through an initial public offering (the "IPO"). Net proceeds from the IPO of approximately \$72 million were used to repay debt incurred in connection with the Acquisition. On December 10, 1996, the Company sold 1,053,000 Dreco Common Shares (as adjusted for the exchange ratio in the Combination) for proceeds of \$39.2 million. This latter transaction is reflected in the financial statements as a 1997 occurrence due to the combination of differing balance sheet dates.

In December 1996, the Company acquired the stock of Vector Oil Tool Ltd. for consideration of 389,000 Dreco Common Shares (as adjusted for the exchange ratio in the Combination) and cash consideration of \$1,481,000. Results of operations for Vector are included in the Company's financial statements for reporting periods beginning in 1997 due to the combination of differing reporting periods. During the second quarter of 1997, the Company completed the acquisition of the drilling controls business of Ross Hill Controls and its affiliate for \$20 million in cash, and acquired by merger 100% of the common stock of PEP, Inc. in exchange for 400,000 shares of National-Oilwell common stock. These transactions did not have a material effect on the Company's historical financial statements.

RESULTS OF OPERATIONS

Operating results by segment are as follows (in millions):

	Quarter Ended September 30,			
	1997	1996	1997	1996
Revenues Products and Technology Downhole Products Distribution Services Eliminations	<pre>\$ 102.3 18.1 162.4 (17.8)</pre>	10.3 134.5		25.4 371.8
Total	\$ 265.0 =======	\$ 200.8	\$ 705.7 ======	\$ 561.0 ======
Operating Income Products and Technology Downhole Products Distribution Services Corporate	\$ 15.0 7.7 8.1 (2.1)	\$ 6.9 4.5 4.6 (2.7)	18.5 19.5	8.9 12.0
Special Charge	28.7 10.7	13.4 	63.4 10.7	30.7
Total	\$ 18.0 =======	\$ 13.4 ======	\$ 52.7 ======	\$ 30.7 ======

Products and Technology

The Products and Technology segment designs and manufactures a large line of proprietary products, including drawworks, mud pumps, power swivels, electrical control systems and reciprocating pumps, as well as complete land drilling and well servicing rigs and structural components such as masts, derricks and substructures. A substantial installed base of these products results in a recurring replacement parts and maintenance business. Sales of new capital equipment can result in large fluctuations in volume between periods depending on the size and timing of the shipment of orders. This segment also provides drilling pump expendable products for maintenance of the Company's and other manufacturers' equipment.

Backlog of the Products and Technology group has grown throughout 1997 and was \$240 million at September 30, 1997, compared to \$38 million at December 31, 1996. Substantially all of the current backlog will be shipped by the end of 1998.

Revenues for the Products and Technology segment increased by \$31.8 million (45%) in the third quarter of 1997 as compared to the same quarter in 1996 due to the inclusion of \$7.6 million of revenues generated by acquisitions completed during 1997 that were not restated for 1996 and due to increased revenues from the sale of drilling capital equipment and spare parts. Operating income increased by \$8.1 million (117%) in the third quarter compared to the same quarter in 1996, with \$1.5 million of the increase due to the acquired businesses and the balance due to higher activity levels.

Products and Technology revenues increased \$46.1 million (23%) in the first nine months of 1997 as compared to the same period in 1996 due primarily to an increase in demand for drilling capital equipment and spare parts, fluid end expendable parts, and reciprocating pumps and associated parts, as well as the acquisitions noted above. Operating income for the Products and Technology segment increased \$15.1 million (91%) in the first nine months of 1997 as compared to the prior year period, representing 33% of the revenue increase. Downhole Products

National-Oilwell designs and manufactures drilling motors and specialized drilling tools for rent and sale. Rentals generally involve products that are not economical for a customer to own or maintain because of the broad range of equipment required for the diverse hole sizes and depths encountered in drilling for oil and gas. Sales generally involve products that require infrequent service, are disposable or are sold in countries where National-Oilwell does not provide repair and maintenance services.

Downhole Products revenues increased by \$7.8 million (76%) in the third quarter of 1997 and \$23.2 million (91%) in the first nine months of 1997 when compared to the same periods in 1996, primarily due to the inclusion of Vector in both periods.

Operating income for the Downhole Products segment increased by \$3.2 million (71%) in the third quarter of 1997 compared to the same quarter in 1996. Nine month operating income results for 1997 have increased by \$9.6 million (108%) over the same period in 1996. The Vector acquisition accounted for virtually all of this increase.

Distribution Services

Distribution Services revenues result primarily from the sale of MRO products from the Company's network of distribution service centers and from the sale of well casing and production tubing. These products are purchased from numerous manufacturers and vendors, including the Company's Products and Technology segment.

Distribution Services revenues during the third quarter of 1997 exceeded the comparable 1996 period by \$27.9 million. This 21% increase reflects the increased spending levels of the Company's alliance partners and other customers. Sales of MRO products, tubular products, drilling spares and fluid end expendable parts accounted for all of this increase. Operating income in the third quarter of 1997 was \$3.5 million (76%) greater than the third quarter of 1996. An increase in operating expenses offset part of the margin recorded due to the volume improvement, netting a 13% flow through of the revenue increase.

Revenues during the first nine months of 1997 increased \$85.0 million (23%) over the comparable 1996 period. Significant increases in the sales of MRO products (\$34 million), tubular goods (\$34 million), drilling spares (\$7 million) and production products (\$6 million) generated the majority of this gain. Operating income increased \$7.5 million (63%) during the first nine months of 1997 compared to the same period in 1996. A portion of the increased margin from the higher revenue levels was offset by higher operating costs associated with the addition in the second half of 1996 of operating and administrative personnel in order to better manage assets and in anticipation of the revenue growth that was achieved in 1997.

Corporate

Corporate costs during the third quarter of 1997 were \$0.6 million lower than the prior year due to the absence of an incentive compensation accrual adjustment that was recorded in the 1996 period by Dreco. Corporate costs were comparable between the nine month periods.

Special Charge

In connection with the Combination, National-Oilwell incurred one-time combination expenses of \$10.7 million (\$8.1 million net of income taxes, or \$0.31 per share) related to various professional fees and integration costs.

Interest Expense

Interest expense decreased substantially during the three months and nine months ended September 30, 1997 due to substantially lower levels of debt that resulted from the reductions made using cash proceeds from the IPO in late 1996.

Income Taxes

The income tax rate for the nine months ended September 30, 1997 of 38% was affected by certain nondeductible Combination expenses in the third quarter, offset in part by certain one-time Canadian tax credits recorded in the first two quarters by Dreco.

Extraordinary Loss

The replacement of the previous credit facility by the New Credit Facility described below, resulted in the write-off of \$1.0 million (\$0.6 million after tax) in deferred financing costs related to the replaced agreement.

LIQUIDITY AND CAPITAL RESOURCES

At September 30, 1997, the Company had working capital of \$228 million, an increase of \$60 million from December 31, 1996. During the first nine months of 1997, net of non-restated acquisitions, accounts receivable increased by \$41 million, as quarterly revenues increased steadily through this period. Inventories increased by \$49 million, due to specific build programs and in response to increasing demand for oilfield equipment and supplies, offset in part by an increase in customer prepayments of \$17 million, accounts payable of \$9 million and accrued income taxes of \$7 million.

The Company's increased business levels have required an increase in expenditures for capital equipment. Total capital expenditures were \$19.5 million during the first nine months of 1997. Additions to the downhole rental fleet and enhancements to information and inventory control systems represent a large portion of these capital expenditures. Additional capital expenditures of as much as \$10 million are anticipated in 1997 to meet the Company's operating needs, including the acquisition of a facility for approximately \$6 million that is currently being leased and further enhancements to the Company's information systems. The Company believes it has sufficient existing manufacturing capacity to meet currently anticipated demand through 1998 for its products and services. Any significantly greater increases in demand for oilfield equipment products, to the extent qualified subcontracting and outsourcing are not available, could result in additional increases in capital expenditures.

On September 25, 1997, the Company entered into a new five-year unsecured \$125 million revolving credit facility (the "New Credit Facility") that was used in part to repay in full amounts outstanding under the previous credit facility as well as the seller notes payable, together with deferred interest, incurred in connection with the Acquisition. The balance of the New Credit Facility will be available for acquisitions and general corporate purposes. The New Credit Facility provides for interest at prime or LIBOR plus 0.5%, subject to adjustment based on the Company's Capitalization Ratio, as defined. The New Credit Facility contains financial covenants and ratios regarding minimum tangible net worth, maximum debt to capital and minimum interest coverage.

The Company believes that cash generated from operations and amounts available under its revolving credit facility will be sufficient to fund operations, working capital needs, capital expenditure requirements and financing obligations. The Company also believes any significant increase in capital expenditures caused by any need to increase manufacturing capacity can be funded from operations or through debt financing. The Company intends to pursue acquisition candidates, but the timing, size or success of any acquisition effort and the related potential capital commitments cannot be predicted. The Company expects to fund future cash acquisitions primarily with cash flow from operations and borrowings, including the unborrowed portion of the New Credit Facility or new debt issuances, but may also issue additional equity in connection with acquisitions. There can be no assurance that additional financing for acquisitions will be available at terms acceptable to the Company.

RECENTLY ISSUED ACCOUNTING STANDARDS

In February 1997, the Financial Accounting Standards Board issued SFAS No. 128, "Earnings Per Share," which specifies the computation, presentation and disclosure requirements for earnings per share. SFAS No. 128 is effective for financial statements for periods ending after December 15, 1997, and earlier adoption is not permitted. Initial adoption of this standard is not expected to have a material impact on National-Oilwell's financial statements.

FORWARD-LOOKING STATEMENTS

This document, other than historical financial information, may contain forward-looking statements that involve risks and uncertainties. Readers are referred to documents filed by the Company with the Securities and Exchange Commission which identify significant risk factors which could cause actual results to differ from those contained in the forward-looking statements, including "Risk Factors" at Item 1 of the Annual Report on Form 10-K, as amended.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

A Special Meeting of Stockholders was held on September 25, 1997.

Stockholders approved the Combination Agreement dated as of May 14, 1997, as amended, between National-Oilwell and Dreco Energy Services Ltd., and the transactions contemplated thereby, by the following votes: for - 17,066,672; against - 1,015; abstained - 4,615.

Stockholders also approved and adopted a recapitalization plan pursuant to which the Amended and Restated Certificate of Incorporation authorizes 75,000,000 shares of Common Stock and one share of Special Voting Stock, by the following votes: for - 16,940,791; against - 125,035; abstained - 6,476.

There were no broker non-votes.

ITEM 5. OTHER INFORMATION

For October 1997, the first full month following the Combination, National-Oilwell reported consolidated revenues of \$89 million and net income of \$5.2 million (\$0.20 per share, assuming the exchange of all Exchangeable Shares).

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

- (a) Exhibits
 - 10.1 Loan Agreement dated as of September 25, 1997
 - 27.1 Financial Data Schedule

Restated Financial Data Schedule, Exhibit 27.1 on Form 10-Q for the Quarter Ended June 30, 1997 filed August 14, 1997

Restated Financial Data Schedule, Exhibit 27.1 on Form 10-Q for the Quarter Ended March 31, 1997 filed May 13, 1997

(b) Reports on Form 8-K

The Company has not filed any report on Form 8-K during the quarter for which this report is filed. A report on Form 8-K was filed on October 8, 1997 regarding the completion of the Combination and a report on Form 8-K will be filed on November 7, 1997 to provide a revised Management's Discussion and Analysis of Financial Condition and Results of Operations and the Company's Restated Consolidated Financial Statements on that basis.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date:	November 7, 1997	/s/ Steven W. Krablin
		Steven W. Krablin Principal Financial and Accounting Officer and Duly Authorized Signatory

EXHIBIT NO.	DESCRIPTION

10.1 Loan Agreement dated as of September 25, 1997

27.1 Financial Data Schedule

Restated Financial Data Schedule, Exhibit 27.1 on Form 10-Q for the Quarter Ended June 30, 1997 filed August 14, 1997

Restated Financial Data Schedule, Exhibit 27.1 on Form 10-Q for the Quarter Ended March 31, 1997 filed May 13, 1997

LOAN AGREEMENT

(\$100,000,000 U.S. LOAN FACILITY

AND

\$25,000,000 CANADIAN LOAN FACILITY)

DATED AS OF SEPTEMBER 25, 1997

AMONG

NATIONAL-OILWELL, L.P., AS U.S. BORROWER,

NATIONAL-OILWELL CANADA LTD. AND DRECO ENERGY SERVICES LTD., AS CANADIAN BORROWER,

> TEXAS COMMERCE BANK NATIONAL ASSOCIATION, AS U.S. AGENT AND AS A U.S. LENDER,

WELLS FARGO BANK (TEXAS), NATIONAL ASSOCIATION, AS U.S. CO-AGENT AND AS A U.S. LENDER,

THE CHASE MANHATTAN BANK OF CANADA, AS CANADIAN AGENT AND AS A CANADIAN LENDER,

THE BANK OF NOVA SCOTIA, AS CANADIAN CO-AGENT, AS A U.S. LENDER AND AS A CANADIAN LENDER

AND

THE OTHER LENDERS NOW OR HEREAFTER PARTIES HERETO

Page

1.	Defin	itions
	1.1	Certain Defined Terms
	1.2	Miscellaneous
2.	Commi	tments; Loans; BA's and Letters of Credit
	2.1	Loans and BA's
	2.2	Letters of Credit
	2.3	Certain Provisions Relating to Bankers' Acceptances 30
	2.4	Terminations, Reductions or Reallocations of Commitments 32
	2.5	Commitment Fees
	2.6	Several Obligations
	2.7	Notes
	2.8	Use of Proceeds
	2.9	Currency Fluctuations
3.	Borro	wings, Prepayments and Interest Options
	3.1	Borrowings
	3.2	Prepayments
	3.3	Interest Options
4.	Payme	nts; Pro Rata Treatment; Computations, Etc
	4.1	Payments
	4.2	Pro Rata Treatment
	4.3	Certain Actions, Notices, Etc
	4.4	Non-Receipt of Funds by Any Agent
	4.5	Sharing of Payments, Etc
5.	Condi	tions Precedent
	5.1	Initial Loans, Letters of Credit and Bankers' Acceptances 46
	5.2	All Loans, Letters of Credit and Bankers' Acceptances 47
6.	Repre	sentations and Warranties
	6.1	Organization
	6.2	Financial Statements
	6.3	Enforceable Obligations; Authorization
	6.4	Other Debt
	6.5	Litigation
	6.6	Title

6.7 Taxes 6.8 6.9 6.10 6.11 6.12 6 13 6.14 6.15 6.16 6.17 6.18 Affirmative Covenants. 7. 7.1 7.2 73 7.4 7.5 7.6 7.7 7.8 7.9 7.10 8. Negative Covenants. 8.1 8.2 8.3 Mergers, Consolidations and Dispositions of Assets 57 8.4 8.5 Nature of Business57Transactions with Related Parties57Loans and Investments58 8.6 8.7 8.8 8.9 8.10 8.11 8.12

50

50

50

58

ii

8.13

8.14

8.15

4

9.	Default	ts
	9.1	Events of Default
	9.2	Right of Setoff
	9.3	Collateral Account
	9.4	Preservation of Security for Unmatured Obligations 62
	9.5	Currency Conversion After Maturity
	9.6	Remedies Cumulative
10.	Agents	
	10.1	Appointment, Powers and Immunities
	10.2	Reliance
	10.3	Defaults
	10.4	Material Written Notices
	10.5	Rights as a Lender
	10.5	
	10.0	Non-Reliance on Agents and Other Lenders
	10.7	
	10.8 10.9	Failure to Act67Resignation or Removal of Agent67
	10.10	No Partnership
	Miccol	laneous
11.		
	11.1	Waiver
	11.2	Notices
	11.3	Expenses, Etc
	11.4	Indemnification
	11.5	Amendments, Etc
	11.6	Successors and Assigns
	11.7	Limitation of Interest
	11.8	Survival
	11.9	Captions
	11.10	Counterparts
	11.11	Venue; Governing Law
	11.12	Severability
	11.13	Tax Forms; Net Payments
	11.14	Interest Act (Canada)
	11.15	Judgment Currency
	11.16	Conflicts Between This Agreement and the Other Loan
	11.10	Documents
	11.17	Limitation on Charges; Substitute Lenders;
	TT .TI	Non-Discrimination
	11 10	Multiple Dereens Comprising Canadian Derrover
	11.18	Multiple Persons Comprising Canadian Borrower

iii

5 EXHIBITS* A-1 -- Request for Extension of Credit (U.S. Borrower) A-2 -- Request for Extension of Credit (Canadian Borrower) B -- Rate Designation Notice C -- Canadian Note D -- U.S. Note E -- Assignment and Acceptance F -- Compliance Certificate G -- Bankers' Acceptance Notice

- H -- Canadian Dollar Note I -- Subsidiaries (Showing Percentage Ownership)
- J -- Existing Borrowed Money Indebtedness
- * All Exhibits have been omitted from this filing. The company will supplementally furnish a copy of any omitted exhibit to the Securities and Exchange Commission upon request.

iv

LOAN AGREEMENT

THIS LOAN AGREEMENT is made and entered into as of September 25, 1997 (the "Effective Date"), by and among NATIONAL-OILWELL, L.P., a Delaware limited partnership (the "U.S. Borrower"); NATIONAL-OILWELL CANADA LTD., a British Columbia company, and DRECO ENERGY SERVICES LTD., an Alberta corporation (collectively, the "Canadian Borrower"); each of the lenders which is or may from time to time become a party hereto (individually, a "Lender" and, collectively, the "Lenders", which terms shall include U.S. Lenders and Canadian Lenders); TEXAS COMMERCE BANK NATIONAL ASSOCIATION ("TCB"), a national banking association, as agent for the U.S. Lenders (in such capacity, together with its successors in such capacity, the "U.S. Agent"), WELLS FARGO BANK (TEXAS), NATIONAL ASSOCIATION, as U.S. Co-Agent, THE CHASE MANHATTAN BANK OF CANADA, as agent for the Canadian Lenders (in such capacity, together with its successors in such capacity, the "Canadian Agent"), and THE BANK OF NOVA SCOTIA, as Canadian Co-Agent.

The parties hereto agree as follows:

1. Definitions.

1.1 Certain Defined Terms.

Unless a particular term, word or phrase is otherwise defined or the context otherwise requires, capitalized terms, words and phrases used herein or in the Loan Documents (as hereinafter defined) have the following meanings (all definitions that are defined in this Agreement in the singular have the same meanings when used in the plural and vice versa):

Acceptance Fee means the fee payable in Canadian Dollars to each Canadian Lender in respect of the Bankers' Acceptances accepted by such Canadian Lender computed in accordance with Section 2.3(c).

Accounts, Equipment, General Intangibles and Inventory shall have the respective meanings assigned to them in the Uniform Commercial Code enacted in the State of Texas in force on the Effective Date.

Additional Interest means the aggregate of all amounts accrued or paid pursuant to the Notes or any of the other Loan Documents (other than interest on the Notes at the Stated Rate and any Acceptance Fee) which, under applicable laws, are or may be deemed to constitute interest on the indebtedness evidenced by the Notes or the other Obligations.

Adjusted LIBOR means, with respect to each Interest Period applicable to a LIBOR Borrowing, a rate per annum equal to the quotient, expressed as a percentage, of (a) LIBOR with respect to such Interest Period divided by (b) 1.0000 minus the Eurodollar Reserve Requirement in effect on the first day of such Interest Period.

7

Affiliate means any Person controlling, controlled by or under common control with any other Person. For purposes of this definition, "control" (including "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities or otherwise.

Agents means U.S. Agent and Canadian Agent, collectively.

Agreement means this Loan Agreement, as it may from time to time be amended, modified, restated or supplemented.

Annual Financial Statements means the annual financial statements of a Person, including all notes thereto, which statements shall include a balance sheet as of the end of the applicable fiscal year and an income statement and a statement of cash flows for such fiscal year, all setting forth in comparative form the corresponding figures from the previous fiscal year, all prepared in conformity with GAAP in all material respects. Whenever an Annual Financial Statement is required by the terms of this Agreement to be audited, it shall be accompanied by the opinion of independent certified public accountants of recognized national standing, which shall state that such financial statements present fairly in all material respects the financial position of such Person and, if such Person has any Subsidiaries, its consolidated Subsidiaries as of the date thereof and the results of its operations for the period covered thereby in conformity with GAAP. If an Annual Financial Statement is not required by the terms of this Agreement to be audited, it shall be certified by the chief financial officer or other authorized officer of such Person as true, correct and complete in all material respects. All Annual Financial Statements (whether or not required to be audited) shall include unaudited consolidating financial statements for the applicable Person, in Proper Form, certified by the chief financial officer or other authorized officer of such Person as true, correct and complete in all material respects.

Applicable BA Discount Rate means, as applicable to a Bankers' Acceptance being purchased by any Canadian Lender on any day, the percentage discount rate (expressed to two decimal places and rounded upward, if necessary, to the nearest 1/100th of 1%) quoted by the Canadian Agent as that at which the Canadian Agent would, in accordance with normal practice, at or about 12:00 noon (Toronto, Ontario time), on such day, be prepared to purchase Bankers' Acceptances in an amount and having a maturity date comparable to the amount and maturity date of such Bankers' Acceptances.

Applicable Canadian Pension Legislation means, at any time, any pension legislation then applicable to the Canadian Borrower, including the Pension Benefits Act (Ontario), including all regulations made thereunder, and all rules, regulations, rulings and interpretations made or issued by any Governmental Authority having or asserting jurisdiction in respect thereof.

Applications means all applications and agreements for Letters of Credit, or similar instruments or agreements, in Proper Form, now or hereafter executed by any Person in connection with any Letter of Credit now or hereafter issued or to be issued under the terms hereof at the request of any Person. The term "Application" as used herein shall include that certain Master Commercial & Standby Letter of Credit Agreement dated executed or to be executed by and between U.S. Borrower and TCB.

Assignment and Acceptance shall have the meaning ascribed to such term in Section 11.6 hereof.

Availability Period means, for each Lender, the period from and including the Effective Date to (but not including) the Termination Date.

BA Discount Proceeds means in respect of any Bankers' Acceptance being purchased by a Canadian Lender on any day under Section 2.3, an amount (rounded to the nearest whole Canadian cent, and with one-half of one Canadian cent being rounded up) calculated on such day by multiplying:

- (A) the face amount of such Bankers' Acceptance; by
- (B) the quotient equal to one divided by the sum of one plus the product of:
 - the Applicable BA Discount Rate (expressed as a decimal) applicable to such Bankers' Acceptance; and
 - (ii) a fraction, the numerator of which is the number of days remaining in the term of such Bankers' Acceptance and the denominator of which is 365;

with such quotient being rounded up or down to the nearest fifth decimal place and .000005 being rounded up.

Bankers' Acceptance or BA means a bill of exchange denominated in Canadian Dollars drawn by the Canadian Borrower on and accepted by a Canadian Lender pursuant to Section 2.3 hereof.

Bankers' Acceptance Liabilities means, at any time and in respect of any Bankers' Acceptance, the face amount thereof if still outstanding and unpaid or, following maturity and payment thereof, the aggregate unpaid amount of all Reimbursement Obligations at that time due and payable in respect of the payment of such Bankers' Acceptance upon maturity.

Bankers' Acceptance Notice has the meaning specified in Section 2.3(a).

Bankruptcy Code means (i) the United States Bankruptcy Code, (ii) the Bankruptcy and Insolvency Act (Canada) and (iii) the Companies' Creditors Arrangement Act (Canada), as the same may be amended and together with any successor statutes.

Base Rate means, for any day, a rate per annum equal to the lesser of (a) the applicable Margin Percentage from time to time in effect plus the greater of (1) the applicable Prime Rate for that day and (2) the Federal Funds Rate for that day plus 1/2 of 1% or (b) the Ceiling Rate. If for any reason any applicable Agent shall have determined (which determination shall be conclusive and binding, absent manifest error) that it is unable to ascertain the Federal Funds Rate for any reason, the Base Rate shall, until the circumstances giving rise to such inability no longer exist, be the lesser of (a) the applicable Prime Rate plus the applicable Margin Percentage from time to time in effect or (b) the Ceiling Rate.

Base Rate Borrowing means that portion of the principal balance of the Loans at any time bearing interest at the Base Rate.

Borrowed Money Indebtedness means, with respect to any Person, without duplication, (i) all obligations of such Person for borrowed money, (ii) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (iii) all obligations of such Person under conditional sale or other title retention agreements relating to Property purchased by such Person, (iv) all obligations of such Person issued or assumed as the deferred purchase price of Property or services (excluding obligations of such Person to creditors for raw materials, inventory, services and supplies and deferred payments for services to employees and former employees incurred in the ordinary course of such Person's business), (v) all capital lease obligations of such Person, (vi) all obligations of others secured by any Lien on Property owned or acquired by such Person, whether or not the obligations secured thereby have been assumed, (vii) all contingent obligations (excluding reimbursement obligations in respect of amounts actually drawn under letters of credit or due and payable in respect of matured bankers' acceptances) of such Person in respect of outstanding letters of credit issued for the account of such Person or bankers' acceptances drawn by such Person, (viii) all guarantees of such Person, and (ix) Interest Rate Risk Indebtedness of such Person.

Borrowers means U.S. Borrower and Canadian Borrower, collectively.

Business Day means any day other than a day on which commercial banks are authorized or required to close in Houston, Texas, Calgary, Alberta or Toronto, Ontario.

Calculation Date means the last Business Day of each month.

Canadian Commitment means, as to any Canadian Lender, the obligation, if any, of such Canadian Lender to make Canadian Loans, incur or participate in Letter of Credit Liabilities relating to Canadian Letters of Credit and accept and purchase Bankers' Acceptances in an aggregate principal amount at any one time outstanding up to (but not exceeding) the amount, if any, set forth opposite such Canadian Lender's name on the signature pages hereof under the caption "Canadian

4

Commitment", or otherwise provided for in an Assignment and Acceptance Agreement (as the same may be increased or reduced from time to time pursuant to Section 2.4 hereof).

Canadian Dollars or C\$ means lawful money of Canada.

10

Canadian Dollar Notes means the Notes of Canadian Borrower evidencing the Canadian Loans denominated in Canadian Dollars, in the form of Exhibit H hereto.

Canadian Lender means each lender signatory hereto with (i) prior to the Termination Date, a Canadian Commitment and (ii) on and after the Termination Date, any outstanding Canadian Obligations.

Canadian Letter of Credit has the meaning assigned to such term in Section 2.2 hereof.

Canadian Loan means a Loan made pursuant to Section 2.1(b) hereof.

Canadian Notes means the Notes of Canadian Borrower evidencing the Canadian Loans denominated in Dollars, in the form of Exhibit C hereto.

Canadian Obligations means, as at any date of determination thereof, the sum of the following (determined without duplication): (i) the aggregate principal amount of Canadian Loans outstanding hereunder on such date, plus (ii) the aggregate amount of the Bankers' Acceptance Liabilities outstanding on such date, plus (iii) the aggregate amount of Letter of Credit Liabilities outstanding on such date relating to Canadian Letters of Credit.

Canadian Prime Loans means Loans made pursuant to Section 2.1(b) hereof which are denominated in Canadian Dollars.

Canadian Prime Rate means, on any day, as to Loans denominated in Canadian Dollars made to Canadian Borrower, the greater of (a) the annual rate of interest announced from time to time by Chase Canada as its prime rate then in effect at its Principal Office, being the reference rate used by Chase Canada for determining interest rates on commercial loans denominated in Canadian Dollars to borrowers in Canada, and (b) an annual rate of interest equal to the sum of (i) the CDOR Rate and (ii) 1.00% per annum. The Canadian Prime Rate is a reference rate and does not necessarily represent the lowest or best rate or a favored rate, and TCB, Chase Canada, each Agent and each Lender disclaims any statement, representation or warranty to the contrary. TCB, Chase Canada, any Agent or any Lender may make commercial loans or other loans at rates of interest at, above or below the Canadian Prime Rate.

Capital Expenditures means, with respect to any Person for any period, expenditures in respect of fixed or capital assets by such Person, including capital lease obligations payable during such period (to the extent not already included), which would be reflected as additions to Property, plant or equipment on a balance sheet of such Person and its consolidated Subsidiaries, if any,

prepared in accordance with GAAP; but excluding expenditures during such period for the repair or replacement of any fixed or capital asset which was destroyed or damaged, in whole or in part, to the extent financed by the proceeds of an insurance policy maintained by such Person. Capital Expenditures shall not include Permitted Investments or the assets owned by any Person acquired by way of a Permitted Investment or assets comprising substantially all of an entire business which is acquired by the applicable Person.

11

CDOR Rate means, on any day, an annual rate of interest equal to the average 30 day rate applicable to Canadian bankers' acceptances appearing on the "Reuters Screen CDOR Page" (as defined in the International Swap Dealer Association, Inc. (1991 ISDA) definitions, as modified and amended from time to time) as of 12:00 noon (Toronto, Ontario time) on such day, or if such day is not a Business Day, then on the immediately preceding Business Day; provided, however, if such rate does not appear on the Reuters Screen CDOR Page as contemplated, then the CDOR Rate on any day shall be calculated as the arithmetic mean of the 30 day rates applicable to Canadian bankers' acceptances quoted by the Canadian Lenders which are listed in Schedule I to the Bank Act (Canada) as of 12:00 noon (Toronto, Ontario time) on such day, or if such day is not a Business Day, then on the immediately preceding Business Day.

Ceiling Rate means, on any day, the maximum nonusurious rate of interest permitted for that day by whichever of applicable United States federal or Texas laws or, in the case of advances made in Canada by Canadian Lenders to Canadian Borrower, whichever of applicable Canada federal or Ontario laws (or the laws of any other jurisdiction whose usury laws are deemed to apply to the Notes or any other Loan Documents despite the intention and desire of the express choice of law provisions set forth herein) permits the higher interest rate, stated as a rate per annum. On each day, if any, that Chapter One establishes the Ceiling Rate, the Ceiling Rate shall be the "indicated rate ceiling" (as defined in Chapter One) for that day. U.S. Agent may from time to time, as to current and future balances, implement any other ceiling under Chapter One by notice to Borrowers, if and to the extent permitted by Chapter One. Without notice to Borrowers or any other Person, the Ceiling Rate shall automatically fluctuate upward and downward as and in the amount by which such maximum nonusurious rate of interest permitted by applicable law fluctuates.

Change of Control means a change resulting when any Unrelated Person or any Unrelated Persons (other than any Person that Beneficially Owns at least 10% of the aggregate voting power of all classes of Voting Stock of the Parent as of the date hereof) acting together which would constitute a Group together with any Affiliates or Related Persons thereof (in each case also constituting Unrelated Persons) shall at any time either (i) Beneficially Own more than 50% of the aggregate voting power of all classes of Voting Stock of Parent or (ii) succeed in having sufficient of its or their nominees elected to the Board of Directors of Parent such that such nominees, when added to any existing directors remaining on the Board of Directors of Parent after such election who is an Affiliate or Related Person of Such Person or Group, shall constitute a majority of the Board of Directors of Parent. As used herein (a) "Beneficially Own" means "beneficially own" as defined in Rule 13d-3 of the Securities Exchange Act of 1934, as amended, or any successor provision thereto; provided, however, that, for purposes of this definition, a Person shall not be deemed to

Beneficially Own securities tendered pursuant to a tender or exchange offer made by or on behalf of such Person or any of such Person's Affiliates until such tendered securities are accepted for purchase or exchange; (b) "Group" means a "group" for purposes of Section 13(d) of the Securities Exchange Act of 1934, as amended; (c) "Unrelated Person" means at any time any Person other than Parent or any Subsidiary of Parent and other than any trust for any employee benefit plan of Parent or any Subsidiary of Parent; (d) "Related Person" of any Person shall mean any other Person owning (1) 5% or more of the outstanding common stock of such Person or (2) 5% or more of the Voting Stock of such Person; and (e) "Voting Stock" of any Person shall mean capital stock of such Person which ordinarily has voting power for the election of directors (or persons performing similar functions) of such Person, whether at all times or only so long as no senior class of securities has such voting power by reason of any contingency.

Chapter One means Chapter One of Title 79, Texas Revised Civil Statutes, 1925, as amended.

Chase Canada means The Chase Manhattan Bank of Canada.

Code means the Internal Revenue Code of 1986, as amended, as now or hereafter in effect, together with all regulations, rulings and interpretations thereof or thereunder by the Internal Revenue Service.

Collateral means all Property, tangible or intangible, real, personal or mixed, now or hereafter subject to the Security Documents.

Commitment Fee Percentage means (i) on any day prior to October 1, 1997, 0.175% per annum and (ii) on and after October 1, 1997, the applicable per annum percentage set forth at the appropriate intersection in the table shown below, based on the Debt to Capitalization Ratio as of the last day of the then most recently ended fiscal quarter of Parent calculated by U.S. Agent as soon as practicable after receipt by U.S. Agent of all financial reports required under this Agreement with respect to such fiscal quarter (including a Compliance Certificate) (provided, however, that if the Commitment Fee Percentage is increased as a result of the reported Debt to Capitalization Ratio, such increase shall be retroactive to the date that U.S. Borrower was obligated to deliver such financial reports to U.S. Agent pursuant to the terms of this Agreement and provided further, however, that if the Commitment Fee Percentage is decreased as a result of the reported Debt to Capitalization Ratio, and such financial reports are delivered to U.S. Agent not more than ten (10) calendar days after the date required to be delivered pursuant to the terms of this Agreement, such decrease shall be retroactive to the date that U.S. Borrower was obligated to deliver such financial reports to U.S. Agent pursuant to the terms of this Agreement):

Debt to Capitalization Ratio	Commitment Fee Percentage
Greater than 30%	0.225
Greater than 20% but less than or equal to 30%	0.175
Less than or equal to 20%	0.15

7

Commitment Percentage means, as to any Lender, the percentage equivalent of a fraction the numerator of which is the amount of such Lender's U.S. Commitment or Canadian Commitment, as the case may be, and the denominator of which is the aggregate amount of the U.S. Commitments or Canadian Commitments, as the case may be, of all Lenders.

Compliance Certificate shall have the meaning given to it in Section 7.2 hereof.

Contribution Agreements means those certain Contribution Agreements dated concurrently herewith executed by and among, respectively, (i) U.S. Borrower and the Guarantors in respect of the U.S. Obligations and (ii) Canadian Borrower and the Guarantors in respect of the Canadian Obligations, as they may from time to time be amended, modified, restated or supplemented.

Controlled Group means all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control which, together with any Borrower, are treated as a single employer under Section 414 of the Code or under Applicable Canadian Pension Legislation.

Corporation means any corporation, limited liability company, partnership, joint venture, joint stock association, business trust and other business entity.

Cover for Letter of Credit Liabilities or any Bankers' Acceptance Liabilities shall be effected by paying to U.S. Agent or Canadian Agent, as the case may be, immediately available funds, to be held by U.S. Agent or Canadian Agent, as the case may be, in a collateral account maintained by U.S. Agent or Canadian Agent, as the case may be, at its Principal Office and collaterally assigned to U.S. Agent or Canadian Agent, as the case may be, as security for the applicable Obligations using documentation reasonably satisfactory to U.S. Agent or Canadian Agent, as the case may be, in the amount required by any applicable provision hereof. Such amount shall be retained by U.S. Agent or Canadian Agent, as the case may be, in such collateral account until such time as the applicable Letter of Credit shall have expired and the Reimbursement Obligations, if any, with respect thereto shall have been fully satisfied or the applicable Bankers' Acceptance shall have matured and the related Bankers' Acceptance Liabilities shall have been fully satisfied; provided, however, that at such time if a Default or Event of Default has occurred and is continuing, U.S. Agent or Canadian Agent, as the case may be, shall not be required to release such amount in such collateral account from the time of such collateral assignment until such Default or Event of Default shall have been cured or waived.

Debt to Capitalization Ratio means, as of any day, the ratio of (a) Borrowed Money Indebtedness (exclusive of the categories of Borrowed Money Indebtedness, other than obligations

8

in respect of bankers' acceptances, described in clauses (vii), (viii) and (ix) of the definition of "Borrowed Money Indebtedness" set forth in this Section 1.1) for Parent and its Subsidiaries (on a consolidated basis) as of such date to (b) Total Capitalization as of such date.

14

Default means an Event of Default or an event which with notice or lapse of time or both would, unless cured or waived, become an Event of Default.

Dollars, US\$ and \$ means lawful money of the United States of America.

Dual Lender means any Lender which has both a U.S. Commitment and a Canadian Commitment.

EBITDA means, without duplication, for any period the consolidated net earnings (excluding any extraordinary gains or losses) plus, to the extent deducted in calculating such consolidated net earnings, depreciation, amortization, other non-cash items, Interest Expense, and federal, state and provincial income tax expense and minus, to the extent added in calculating such consolidated net earnings, any non-cash items, in each case, for Parent and its Subsidiaries (on a consolidated basis).

Environmental Claim means any third party (including Governmental Authorities and employees) action, lawsuit, claim or proceeding (including claims or proceedings at common law or under the Occupational Safety and Health Act or similar laws relating to safety of employees) which seeks to impose liability for (i) noise; (ii) pollution, contamination, protection or clean-up of the air, surface water, ground water or land; (iii) solid, gaseous or liquid waste generation, handling, treatment, storage, disposal or transportation; (iv) exposure to Hazardous Substances; (v) the safety or health of employees or (vi) the manufacture, processing, distribution in commerce, use, discharge or storage of Hazardous Substances. An "Environmental Claim" includes, but is not limited to, a common law action, as well as a proceeding to issue, modify or terminate an Environmental Permit to the extent that such a proceeding attempts to redress violations of an applicable permit, license, or regulation as alleged by any Governmental Authority.

Environmental Liabilities includes all liabilities arising from any Environmental Claim, Environmental Permit or Requirements of Environmental Law under any theory of recovery, at law or in equity, and whether based on negligence, strict liability or otherwise, including but not limited to remedial, removal, response, abatement, investigative, monitoring, personal injury and damage to Property or injuries to persons, and any other related costs, expenses, losses, damages, penalties, fines, liabilities and obligations, and all costs and expenses necessary to cause the issuance, reissuance or renewal of any Environmental Permit including reasonable attorneys' fees and court costs.

Environmental Permit means any permit, license, approval or other authorization under any applicable Legal Requirement relating to pollution or protection of health or the environment, including laws, regulations or other requirements relating to emissions, discharges, releases or threatened releases of pollutants, contaminants or hazardous substances or toxic materials or wastes

into ambient air, surface water, ground water or land, or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport, or handling of pollutants, contaminants or Hazardous Substances.

ERISA means the Employee Retirement Income Security Act of 1974, as amended from time to time, and all rules, regulations, rulings and interpretations adopted by the Internal Revenue Service or the U.S. Department of Labor thereunder and, as the context may require, Applicable Canadian Pension Legislation.

Eurodollar Rate means for any day during an Interest Period for a LIBOR Borrowing a rate per annum equal to the lesser of (a) the sum of (1) the Adjusted LIBOR in effect on the first day of such Interest Period plus (2) the applicable Margin Percentage from time to time in effect and (b) the Ceiling Rate. Each Eurodollar Rate is subject to adjustments for reserves, insurance assessments and other matters as provided for in Section 3.3 hereof.

Eurodollar Reserve Requirement means, on any day, that percentage (expressed as a decimal fraction and rounded, if necessary, to the next highest one ten thousandth [.0001]) which is in effect on such day for determining all reserve requirements (including, without limitation, basic, supplemental, marginal and emergency reserves) applicable to "Eurocurrency liabilities," as currently defined in Regulation D. Each determination of the Eurodollar Reserve Requirement by any Agent shall be conclusive and binding, absent manifest error, and may be computed using any reasonable averaging and attribution method.

 $% \left({{{\rm{Event}}} \right)$ of Default shall have the meaning assigned to it in Section 9 hereof.

Exchange Rate means, on any day, (a) with respect to Canadian Dollars in relation to Dollars, the spot rate as quoted by the Bank of Canada as its noon spot rate at which Dollars are offered on such day for Canadian Dollars, and (b) with respect to Dollars in relation to Canadian Dollars, the spot rate as quoted by the Bank of Canada as its noon spot rate at which Canadian Dollars are offered on such day for Dollars.

Exchangeable Shares means the "Exchangeable Shares" of Dreco Energy Services Ltd. issued pursuant to that certain Combination Agreement dated as of May 14, 1997 executed by and between National-Oilwell, Inc. and Dreco Energy Services Ltd.

Federal Funds Rate means, for any day, a fluctuating interest rate per annum equal for such day to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any such day which is a Business Day, the average of the quotations for such day on such transactions received by U.S. Agent from three Federal funds brokers of recognized standing selected by U.S. Agent in its sole and absolute discretion.

10

Financing Statements means all such Uniform Commercial Code or Canadian provincial personal Property security financing statements as any Agent shall reasonably require, in Proper Form, duly executed by U.S. Borrower (or any other applicable Obligor) to give notice of and to perfect or continue perfection of Agents' Liens in any applicable Collateral, as any of the foregoing may from time to time be amended, modified, supplemented or restated.

Foreign Subsidiaries means Subsidiaries which are organized under the laws of a jurisdiction other than the United States of America, any State of the United States or any political subdivision thereof.

Funding Loss means, with respect to (a) any Borrower's payment of principal of a LIBOR Borrowing on a day other than the last day of the applicable Interest Period; (b) any Borrower's failure to borrow a LIBOR Borrowing or to borrow funds by way of Bankers' Acceptances on the date specified by such Borrower; (c) any Borrower's failure to make any prepayment of the Loans (other than Base Rate Borrowings and Canadian Prime Loans) on the date specified by such Borrower, or (d) any cessation of a Eurodollar Rate to apply to the Loans or any part thereof pursuant to Section 3.3, in each case whether voluntary or involuntary, any loss, expense, penalty, premium or liability actually incurred by any Lender (including but not limited to any loss or expense incurred by reason of the liquidation or reemployment of deposits or other funds acquired by any Lender to fund or maintain a Loan).

GAAP means, as to a particular Person, such United States accounting practice as, in the opinion of independent certified public accountants of recognized national standing regularly retained by such Person, conforms at the time to generally accepted accounting principles, consistently applied for all periods after the Effective Date so as to present fairly the financial condition, and results of operations and cash flows, of such Person. If any change in any accounting principle or practice is required by the Financial Accounting Standards Board, all reports and financial statements required hereunder may be prepared in accordance with such change so long as the applicable Person provides to Agents such disclosures of the impact of such change as any Agent may reasonably require. No such change in any accounting principle or practice shall, in itself, cause a Default or Event of Default hereunder (but Borrowers, Agents and Lenders shall negotiate in good faith to replace any financial covenants hereunder to the extent such financial covenants are affected by such change in accounting principle or practice).

Governmental Authority means any governmental authority of the United States of America, Canada, any State of the United States, any Province of Canada, or of any other foreign jurisdiction and any political subdivision of any of the foregoing, and any central bank, agency, department, commission, board, bureau, court or other tribunal having or asserting jurisdiction over any Agent, any Lender, any Obligor or their respective Property.

Guaranties means, collectively, (i) the Guaranties dated concurrently herewith executed by Parent and each of the current Material Subsidiaries (other than Foreign Subsidiaries) in favor of U.S. Agent, for the benefit of U.S. Lenders, (ii) the Guaranties dated concurrently herewith executed

11

by U.S. Borrower, Parent and each of the current Material Subsidiaries (other than Foreign Subsidiaries) in favor of Canadian Agent, for the benefit of Canadian Lenders and (iii) any and all other guaranties hereafter executed in favor of any Agent, for the benefit of U.S. Lenders or Canadian Lenders, relating to the Obligations, as any of them may from time to time be amended, modified, restated or supplemented.

Hazardous Substance means petroleum products, and any hazardous or toxic waste or substance defined or regulated as such from time to time by any law, rule, regulation or order described in the definition of "Requirements of Environmental Law".

Interest Coverage Ratio means, as of any day, the ratio of (a) EBITDA for the 12 months ending on the last day of the immediately preceding calendar month less Capital Expenditures by Parent and its Subsidiaries (on a consolidated basis) during such period to (b) cash Interest Expense for such period.

Interest Expense means, for any period, total interest expense (including interest expense attributable to capitalized leases and net costs under interest rate swap, collar, cap or similar agreements providing interest rate protection), determined on a consolidated basis in accordance with GAAP.

Interest Options means the Base Rate, each Eurodollar Rate and, as to the Canadian Dollar Notes only, the Canadian Prime Rate, and "Interest Option" means any of them.

Interest Payment Dates means (a) for Base Rate Borrowings and for Canadian Prime Loans, December 31, 1997 and the last day of each March, June, September and December thereafter prior to the Maturity Date, and the Maturity Date; and (b) for LIBOR Borrowings, the end of the applicable Interest Period (and if such Interest Period exceeds three months' duration, quarterly, commencing on the first quarterly anniversary of the first day of such Interest Period) and the Maturity Date.

Interest Period means, for each LIBOR Borrowing, a period commencing on the date such LIBOR Borrowing began and ending on the numerically corresponding day which is, subject to availability as set forth in Section 3.3(c)(iii), 1, 2, 3 or 6 months thereafter, as any Borrower shall elect in accordance herewith; provided, (1) unless Agents shall otherwise consent, no Interest Period with respect to a LIBOR Borrowing shall commence on a date earlier than three (3) Business Days after this Agreement shall have been fully executed; (2) any Interest Period with respect to a LIBOR Borrowing which would otherwise end on a day which is not a LIBOR Business Day shall be extended to the next succeeding LIBOR Business Day, unless such LIBOR Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding LIBOR Business Day; (3) any Interest Period with respect to a LIBOR Borrowing which begins on the last LIBOR Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last LIBOR Business Day of the appropriate calendar month; (4) no Interest Period for a Loan shall ever

12

extend beyond the Maturity Date, and (5) Interest Periods shall be selected by each Borrower in such a manner that the Interest Period with respect to any portion of the Loans which shall become due shall not extend beyond such due date.

18

Interest Rate Risk Agreement means an interest rate swap agreement, interest rate cap agreement, interest rate collar agreement or similar arrangement entered into by U.S. Borrower for the purpose of reducing U.S. Borrower's exposure to interest rate fluctuations and not for speculative purposes, approved in writing by U.S. Agent (such approval not to be unreasonably withheld), as it may from time to time be amended, modified, restated or supplemented.

Interest Rate Risk Indebtedness means all obligations and Indebtedness of U.S. Borrower with respect to the program for the hedging of interest rate risk provided for in any Interest Rate Risk Agreement.

Investment means the purchase or other acquisition of any securities or indebtedness of, or the making of any loan, advance, transfer of Property (other than transfers in the ordinary course of business) or capital contribution to, or the incurring of any liability (other than trade accounts payable arising in the ordinary course of business), contingently or otherwise, in respect of the indebtedness of, any Person.

Issuer means the issuer (or, where applicable, each issuer) of a Letter of Credit under this Agreement.

Key Agreements means all contracts, permits, licenses and other rights acquired by any Obligor or to which any Obligor is a party or by which any Obligor is bound and from time to time material to the ownership of assets or the operations of any Obligor. The term "Key Agreements" includes any document or paper evidencing, securing or otherwise relating to any Subordinated Indebtedness and also includes all collective bargaining agreements.

Legal Requirement means any law, statute, ordinance, decree, requirement, order, judgment, rule, or regulation (or interpretation of any of the foregoing) of, and the terms of any license or permit issued by, any Governmental Authority, whether presently existing or arising in the future.

Letter of Credit Liabilities means, at any time and in respect of any Letter of Credit, the sum of (i) the amount available for drawings under such Letter of Credit plus (ii) the aggregate unpaid amount of all Reimbursement Obligations at the time due and payable in respect of previous drawings made under such Letter of Credit. For the purpose of determining at any time the amount described in clause (i), in the case of any Letter of Credit payable in a currency other than Dollars or Canadian Dollars, such amount shall be converted by Agent to Dollars by any reasonable method, and such converted amount shall be conclusive and binding, absent manifest error.

Letters of Credit means the U.S. Letters of Credit and the Canadian Letters of Credit.

LIBOR means, for each Interest Period for any LIBOR Borrowing, the rate per annum (rounded upwards, if necessary, to the nearest 1/16th of 1%) equal to the average of the offered quotations appearing on Telerate Page 3750 (or if such Telerate Page shall not be available, any successor or similar service as may be selected by Agents and Borrowers) as of 11:00 a.m., Houston, Texas time (in respect of a LIBOR Borrowing relating to the U.S. Loans) or 12:00 noon, Toronto, Ontario time (in respect of a LIBOR Borrowing relating to the Canadian Loans) (or, in either case, as soon thereafter as practicable) on the day two LIBOR Business Days prior to the first day of such Interest Period for deposits in Dollars having a term comparable to such Interest Period and in an amount comparable to the principal amount of the LIBOR Borrowing to which such Interest Period relates. If none of such Telerate Page 3750 nor any successor or similar service is available, then "LIBOR" shall mean, with respect to any Interest Period for any applicable LIBOR Borrowing, the rate of interest per annum, rounded upwards, if necessary, to the nearest 1/16th of 1%, quoted by U.S. Agent at or before 11:00 a.m., Houston, Texas time (in respect of a LIBOR Borrowing relating to the U.S. Loans) or 12:00 noon, Toronto, Ontario time (in respect of a LIBOR Borrowing relating to the Canadian Loans) (or, in either case, as soon thereafter as practicable), on the date two LIBOR Business Days before the first day of such Interest Period, to be the arithmetic average of the prevailing rates per annum at the time of determination and in accordance with the then existing practice in the applicable market, for the offering to U.S. Agent or Canadian Agent, as the case may be, by one or more prime banks selected by such Agent in its sole discretion, in the London interbank market, of deposits in Dollars for delivery on the first day of such Interest Period and having a maturity equal (or as nearly equal as may be) to the length of such Interest Period and in an amount equal (or as nearly equal as may be) to the LIBOR Borrowing to which such Interest Period relates. Each determination by any Agent of LIBOR shall be conclusive and binding, absent manifest error, and may be computed using any reasonable averaging and attribution method.

LIBOR Borrowing means each portion of the principal balance of the Loans at any time bearing interest at a Eurodollar Rate.

LIBOR Business Day means a Business Day on which transactions in Dollar deposits between lenders may be carried on in the London interbank market.

Lien means any mortgage, pledge, charge, encumbrance, security interest, collateral assignment or other lien or restriction of any kind, whether based on common law, constitutional provision, statute or contract, and whether or not consensual, and shall include reservations, exceptions, encroachments, easements, rights of way, covenants, conditions, restrictions and other title exceptions.

Loans means the loans provided for by Section 2.1(a) and (b) hereof.

Loan Documents means, collectively, this Agreement, the Notes, the Bankers' Acceptances, the Bankers' Acceptance Notices, the Guaranties, the Contribution Agreements, all Applications, the Security Documents, the Notice of Entire Agreement, all instruments, certificates and agreements now or hereafter executed or delivered by any Obligor to any Agent or any Lender pursuant to any

of the foregoing or in connection with the Obligations or any commitment regarding the Obligations, and all amendments, modifications, renewals, extensions, increases and rearrangements of, and substitutions for, any of the foregoing.

Majority Lenders means, at any time while the Commitments are outstanding, Lenders having greater than 66-2/3% of the aggregate amount of Commitments, and at any other time, Lenders having greater than 66-2/3% of the aggregate amount of Obligations outstanding.

Margin Percentage means (i) on any day prior to October 1, 1997, 0.0% per annum with respect to Base Rate Borrowings and Canadian Prime Loans and 0.50% per annum with respect to LIBOR Borrowings and (ii) on and after October 1, 1997, the applicable per annum percentage set forth at the appropriate intersection in the table shown below, based on the Debt to Capitalization Ratio as of the last day of the then most recently ended fiscal quarter of Parent calculated by U.S. Agent as soon as practicable after receipt by U.S. Agent of all financial reports required under this Agreement with respect to such fiscal quarter (including a Compliance Certificate) (provided, however, that if the Margin Percentage is increased as a result of the reported Debt to Capitalization Ratio, such increase shall be retroactive to the date that U.S. Borrower was obligated to deliver such financial reports to U.S. Agent pursuant to the terms of this Agreement and provided further, however, that if the Margin Percentage is decreased as a result of the reported Debt to Capitalization Ratio, and such financial reports are delivered to U.S. Agent not more than ten (10) calendar days after the date required to be delivered pursuant to the terms of this Agreement, such decrease shall be retroactive to the date that U.S. Borrower was obligated to deliver such financial reports to U.S. Agent pursuant to the terms of this Agreement):

Debt to Capitalization Ratio	LIBOR Borrowings Margin Percentage	Canadian Prime Loans/ Base Rate Borrowings Margin Percentage
Greater than 30%	0.625	0.0
Greater than 20% but less than or equal to 30%	0.50	0.0
Less than or equal to 20%	0.375	0.0

Material Adverse Effect means a material adverse effect on the properties, business, operations, assets or condition (financial or otherwise) of Parent and the other Obligors, taken as a whole, or on the ability of the Obligors, taken as a whole, to perform their obligations under the Loan Documents.

Material Subsidiary means (i) any Subsidiary of Parent which owns assets the value of which comprises five percent (5%) or more of the book value of all of the assets of Parent and its Subsidiaries (on a consolidated basis) or having gross revenues for the immediately preceding fiscal

15

year which comprises five percent (5%) or more of the gross revenues of Parent and its Subsidiaries for such fiscal year (on a consolidated basis) and (ii) any other Subsidiary of Parent (other than a Foreign Subsidiary) which any Borrower from time to time designates as a "Material Subsidiary" by written notice to Agents.

21

Maturity Date means the maturity of the Notes and the other Obligations, September 30, 2002.

Maximum Canadian Available Amount means \$25,000,000. In connection with the application of any provision hereof using the term "Maximum Canadian Available Amount", any amounts denominated in Canadian Dollars shall be converted to Dollars using the then current Exchange Rate. The Maximum Canadian Available Amount is subject to change pursuant to Section 2.4(c) hereof.

Maximum U.S. Available Amount means \$100,000,000. The Maximum U.S. Available Amount is subject to change pursuant to Section 2.4(c) hereof.

Notes shall have the meaning assigned to such term in Section 2.7 hereof.

Notice of Entire Agreement means a notice of entire agreement, in Proper Form, executed by Borrowers, each other Obligor and Agents, as the same may from time to time be amended, modified, supplemented or restated.

Obligations means, as at any date of determination thereof, the sum of the following: (i) the aggregate principal amount of Loans outstanding hereunder on such date, plus (ii) the aggregate amount of the outstanding Letter of Credit Liabilities on such date, plus (iii) the aggregate amount of outstanding Bankers' Acceptance Liabilities on such date, plus (iv) all other outstanding liabilities, obligations and indebtedness of any Obligor under any Loan Document on such date.

Obligors means each Borrower, Parent and each Material Subsidiary.

Organizational Documents means, with respect to a United States corporation, the certificate of incorporation, articles of incorporation and bylaws of such corporation; with respect to a partnership, the partnership agreement establishing such partnership; with respect to a trust, the instrument establishing such trust and with respect to any other Person, the agreements or instruments pursuant to which such Person was formed and by which such Person is governed; in each case including any and all modifications thereof as of the date of the Loan Document referring to such Organizational Document and any and all future modifications thereof.

Parent means National-Oilwell, Inc., a Delaware corporation.

Past Due Rate means, on any day, a rate per annum equal to the lesser of (i) the Ceiling Rate for that day or (ii) the applicable Base Rate or Canadian Prime Rate, as the case may be, plus two percent (2%).

PBGC means the Pension Benefit Guaranty Corporation or any entity succeeding to any or all of its functions under ERISA and any pension commission or similar body constituted under any Applicable Canadian Pension Legislation.

Permitted Dividends means (i) dividends or distributions by a Subsidiary of a Borrower to such Borrower or redemption by a Subsidiary of any of its stock held by any Borrower and (ii) so long as no Default or Event of Default shall have occurred and be continuing (or would result therefrom), any other dividends or distributions.

Permitted Investments means: (a) readily marketable securities issued or fully guaranteed by the full faith and credit of the United States of America or of Canada with maturities of not more than one year; (b) commercial paper rated "Prime 2" by Moody's Investors Service, Inc. or "A-2" by Standard and Poor's Ratings Services or the equivalent thereof by Dominion Bond Rating Service Limited with maturities of not more than 180 days; (c) certificates of deposit or repurchase obligations issued by any U.S. or Canadian domestic bank having capital surplus of at least \$100,000,000 or by any other financial institution acceptable to Agents, all of the foregoing not having a maturity of more than one year from the date of issuance thereof; (d) money market fund the assets of which consist primarily of investments included in clauses (a), (d) money market funds (b) and (c) above; (e) loans made to Parent or U.S. Borrower or any Material Subsidiary which has executed Guaranties in favor of the Agents and loans made by a Person comprising the "Canadian Borrower" to another Person comprising the "Canadian Borrower"; (f) loans not exceeding \$20,000,000 in the aggregate at any one time outstanding to Canadian Borrower (exclusive of loans permitted under clause (e) above); (g) loans (exclusive of loans permitted under clause (e) above or clause (h) below) not exceeding \$10,000,000 in the aggregate at any one time outstanding to any Material Subsidiary the equity interests in which have been pledged to the Agents pursuant to the Security Agreements, or (h) loans (exclusive of loans permitted under clauses (e), (f) or (g) above) made to Subsidiaries so long as the aggregate unpaid principal balance of all such loans does not exceed \$3,000,000 at any time outstanding (the loans described in clauses (e), (f), (g) and (h) above shall be subject to the last sentence of Section 8.1 hereof).

Permitted Liens means each of the following: (a) artisans' or mechanics' Liens arising in the ordinary course of business, and Liens for taxes, but only to the extent that payment thereof shall not at the time be due or if due, the payment thereof is being diligently contested in good faith and adequate reserves computed in accordance with GAAP have been set aside therefor; (b) Liens in effect on the Effective Date and disclosed to the Lenders in the financial statements delivered on or prior to the Effective Date pursuant to Section 6.2 hereof or in a schedule hereto, provided that neither the Borrowed Money Indebtedness secured thereby nor the Property covered thereby shall increase after the Effective Date without the prior written consent of the Majority Lenders; (c) normal encumbrances and restrictions on title which do not secure Borrowed Money Indebtedness and which do not have a material adverse effect on the value or utility of the applicable Property;

17

(d) Liens under the Loan Documents including, without limitation, Liens securing Interest Rate Risk Indebtedness owed to one or more of the U.S. Lenders (but not to any Person which is not, at such time, a U.S. Lender); (e) Liens incurred or deposits made in the ordinary course of business (1) in connection with workmen's compensation, unemployment insurance, social security and other like laws, or (2) to secure insurance in the ordinary course of business, the performance of bids, tenders, contracts, leases, licenses, statutory obligations, surety, appeal and performance bonds and other similar obligations incurred in the ordinary course of business, not, in any of the cases specified in this clause (2), incurred in connection with the borrowing of money, the obtaining of advances or the payment of the deferred purchase price of Property; (f) attachments, judgments and other similar Liens arising in connection with court proceedings, provided that the execution and enforcement of such Liens are effectively stayed and the claims secured thereby are being actively contested in good faith with adequate reserves made therefor in accordance with GAAP; (g) Liens imposed by law, such as carriers', warehousemen's, mechanics', materialmen's and vendors' Liens, incurred in good faith in the ordinary course of business and securing obligations which are not yet due or which are being contested in good faith by appropriate proceedings if adequate reserves with respect thereto are maintained in accordance with GAAP; (h) zoning restrictions, easements, licenses, reservations, provisions, covenants, conditions, waivers, and restrictions on the use of Property, and which do not in any case singly or in the aggregate materially impair the present use or value of the Property subject to any such restriction or materially interfere with the ordinary conduct of the business of any Obligor; (i) Liens securing purchase money Borrowed Money Indebtedness permitted under Section 8.1 hereof and covering the Property so purchased; (j) capital leases and sale/leaseback transactions permitted under the other provisions of this Agreement, and (k) extensions, renewals and replacements of Liens referred to in clauses (a) through (j) of this definition; provided that any such extension, renewal or replacement Lien shall be limited to the Property or assets covered by the Lien extended, renewed or replaced and that the Borrowed Money Indebtedness secured by any such extension, renewal or replacement Lien shall be in an amount not greater than the amount of the Borrowed Money Indebtedness secured by the Lien extended, renewed or replaced.

Person means any individual, Corporation, trust, unincorporated organization, Governmental Authority or any other form of entity.

Plan means an employee pension benefit plan which is covered by Title IV of ERISA or subject to the minimum funding standards under Section 412 of the Code or any Applicable Canadian Pension Legislation and is either (a) maintained by any Borrower or any member of the Controlled Group for employees of any Borrower or any member of the Controlled Group or (b) maintained pursuant to a collective bargaining agreement or any other arrangement under which more than one employer makes contributions and to which any Borrower or any member of the Controlled Group is then making or accruing an obligation to make contributions or has within the preceding five plan years made contributions.

Prime Rate means, on any day, (a) as to Loans made to U.S. Borrower, the prime rate for that day as determined from time to time by TCB and (b) as to Loans denominated in Dollars made to Canadian Borrower, the base rate for that day for Loans denominated in Dollars quoted by Chase

18

Canada. The Prime Rate is, in each case, a reference rate and does not necessarily represent the lowest or best rate or a favored rate, and TCB, Chase Canada, each Agent and each Lender disclaims any statement, representation or warranty to the contrary. TCB, Chase Canada, any Agent or any Lender may make commercial loans or other loans at rates of interest at, above or below the Prime Rate.

Principal Office means (a) as to Obligations of U.S. Borrower, the principal office of U.S. Agent, presently located at 712 Main Street, Houston, Harris County, Texas 77002 and (b) as to Obligations of Canadian Borrower, the principal office of Canadian Agent, presently located at First Canadian Place, 100 King Street West, Suite 6900, Toronto, Ontario M5X 1A4.

Proper Form means in form and substance reasonably satisfactory to Agents.

Property means any interest in any kind of property or assets, whether real, personal or mixed, tangible or intangible.

Quarterly Dates means the last day of each March, June, September and December, provided that if any such date is not a Business Day, then the relevant Quarterly Date shall be the next succeeding Business Day.

Quarterly Financial Statements means the quarterly financial statements of a Person, which statements shall include a balance sheet as of the end of the applicable fiscal quarter and an income statement and a statement of cash flows for such fiscal quarter and for the fiscal year to date, subject to normal year-end adjustments, all setting forth in comparative form the corresponding figures as of the end of and for the corresponding fiscal quarter of the preceding year, prepared in accordance with GAAP in all material respects except that such statements are condensed and exclude detailed footnote disclosures and certified by the chief financial officer or other authorized officer of such Person as fairly presenting, in all material respects, the financial condition of such person as of such date. Quarterly Financial Statements shall also include unaudited consolidating financial statements for the applicable Person, in Proper Form, certified by the chief financial officer or other authorized officer of such Person as true, correct and complete in all material respects.

Rate Designation Date means that Business Day which is (a) in the case of Base Rate Borrowings by the U.S. Borrower, 11:00 a.m., Houston, Texas time, and, in the case of Base Rate Borrowings by the Canadian Borrower, 12:00 noon, Toronto, Ontario time, in each case on the date of such borrowing and (b) in the case of LIBOR Borrowings by the U.S. Borrower, 11:00 a.m., Houston, Texas time, and, in the case of LIBOR Borrowings by the Canadian Borrower, 12:00 noon, Toronto, Ontario time, in each case, on the date three LIBOR Business Days preceding the first day of any proposed Interest Period.

Rate Designation Notice means a written notice substantially in the form of Exhibit B.

Refunding Bankers' Acceptance has the meaning specified in Section 2.3.

Regulation D means Regulation D of the Board of Governors of the Federal Reserve System from time to time in effect and includes any successor or other regulation relating to reserve requirements applicable to member banks of the Federal Reserve System.

Regulatory Change means with respect to any Lender, any change on or after the date of this Agreement in any Legal Requirement (including, without limitation, Regulation D) or the adoption or making on or after such date of any interpretation, directive or request applying to a class of lenders including such Lender under any Legal Requirements (whether or not having the force of law) by any Governmental Authority.

Reimbursement Obligations means, as at any date, (i) the obligations of any Borrower then outstanding in respect of Letters of Credit under this Agreement, to reimburse the applicable Issuers for the amount paid by such Issuers in respect of any drawing under such Letters of Credit and (ii) the obligations of the Canadian Borrower then outstanding in respect of any Bankers' Acceptance paid by any Canadian Lender on maturity thereof. Except for Canadian Letters of Credit denominated in Canadian Dollars, Reimbursement Obligations in respect of any Letter of Credit shall at all times be payable in Dollars notwithstanding any such Letter of Credit being payable in a currency other than Dollars

Request for Extension of Credit means a request for extension of credit duly executed by the chief executive officer, chief financial officer or treasurer of U.S. Borrower or Canadian Borrower, as the case may be, or any other Person duly authorized by one of such officers, appropriately completed and substantially in the form of Exhibit A-1 (U.S. Borrower) or Exhibit A-2 (Canadian Borrower) attached hereto, as the case may be.

Requirements of Environmental Law means all requirements imposed by any law (including for example and without limitation The Resource Conservation and Recovery Act and The Comprehensive Environmental Response, Compensation, and Liability Act), rule, regulation, or order of any federal, state or local executive, legislative, judicial, regulatory or administrative agency, board or authority in effect at the applicable time which relate to (i) noise; (ii) pollution, contamination, protection or clean-up of the air, surface water, ground water or land; (iii) solid, gaseous or liquid waste generation, handling, treatment, storage, disposal or transportation; (iv) exposure to Hazardous Substances; (v) the safety or health of employees or (vi) the manufacture, processing, distribution in commerce, use, discharge or storage of Hazardous Substances.

Reset Date has the meaning specified in Section 2.9(a).

Secretary's Certificate means a certificate, in Proper Form, of the Secretary or an Assistant Secretary of a corporation as to (a) the resolutions of the Board of Directors of such corporation authorizing the execution, delivery and performance of the documents to be executed by such corporation; (b) the incumbency and signature of the officer of such corporation executing such documents on behalf of such corporation, and (c) the Organizational Documents of such corporation.

Security Agreements means (i) security agreements, each in Proper Form, executed or to be executed in favor of U.S. Agent, securing the U.S. Obligations, covering all of the issued and outstanding equity interests in any Material Subsidiary (other than Foreign Subsidiaries and other than Subsidiaries which are wholly-owned direct Subsidiaries of Foreign Subsidiaries) and 65% of the issued and outstanding equity interests (exclusive of the Exchangeable Shares) in any Material Subsidiary which is a Foreign Subsidiaries of other Foreign Subsidiaries which are direct Subsidiaries of other Foreign Subsidiaries) and (ii) security agreements, each in Proper Form, executed in favor of Canadian Agent, securing the Canadian Obligations, covering the remaining 35% of the issued and outstanding equity interests (exclusive of the Exchangeable Shares) in any Material Subsidiary which is a Foreign Subsidiary of Parent (other than Foreign Subsidiaries which are direct Subsidiaries of other Foreign Subsidiary of Parent (other than Foreign Subsidiaries) and outstanding equity interests (exclusive of the Exchangeable Shares) in any Material Subsidiary which is a Foreign Subsidiary of Parent (other than Foreign Subsidiaries which are direct Subsidiaries of other Foreign Subsidiaries), as the same may from time to time be amended, modified, restated or supplemented.

Security Documents means, collectively, the Security Agreements and any and all other security documents now or hereafter executed and delivered by any Obligor to secure all or any part of the Obligations, as any of them may from time to time be amended, modified, restated or supplemented.

Stated Rate means the effective weighted per annum rate of interest applicable to the Obligations; provided, that if on any day such rate shall exceed the Ceiling Rate for that day, the Stated Rate shall be fixed at the Ceiling Rate on that day and on each day thereafter until the total amount of interest accrued at the Stated Rate on the unpaid principal balances of the Notes and the other Obligations plus the Additional Interest equals the total amount of interest which would have accrued if there had been no Ceiling Rate. If the Obligations mature (or are prepaid) before such equality is achieved, then, in addition to the unpaid principal and accrued interest then owing pursuant to the other provisions of the Loan Documents, the applicable Borrower promises to pay on demand to the order of the holder of the applicable Obligations interest in an amount equal to the excess (if any) of (a) the lesser of (i) the total interest which would have accrued on such Obligations if the Stated Rate had been defined as equal to the Ceiling Rate from time to time in effect and (ii) the total interest which would have accrued on such Obligations if the Stated Rate were not so prohibited from exceeding the Ceiling Rate, over (b) the total interest actually accrued on such Obligations to such maturity (or prepayment) date. Without notice to any Borrower or any other Person, the Stated Rate shall automatically fluctuate upward and downward in accordance with the provisions of this definition.

Subordinated Indebtedness means all Borrowed Money Indebtedness of a Person which has been subordinated on terms and conditions satisfactory to the Majority Lenders, in their sole discretion, to all Borrowed Money Indebtedness of such Person to Lenders, whether now existing or hereafter incurred. Borrowed Money Indebtedness shall not be considered as "Subordinated Indebtedness" unless and until Agents shall have received copies of the documentation evidencing or relating to such Borrowed Money Indebtedness together with a subordination agreement, in Proper Form, duly executed by the holder or holders of such Borrowed Money Indebtedness and evidencing the terms and conditions of subordination required by the Majority Lenders. The Lenders and the

Agents acknowledge that the subordination provisions relating to the the Subordinated Indebtedness described on Exhibit J hereto which is owed to Oilwell, Inc. and National Supply Company, Inc. does not require prior approval by any Agent or any Lender.

Subsidiary means, as to a particular parent Corporation, any Corporation of which more than 50% of the indicia of equity rights (whether outstanding capital stock or otherwise) is at the time directly or indirectly owned by, such parent Corporation.

Tangible Net Worth means, without duplication, the sum of (i) capital stock, (ii) preferred stock, (iii) paid-in capital and (iv) retained earnings minus the sum of any amount shown on account of any (a) patents, patent applications, trademarks, service marks, copyrights and trade names and (b) goodwill and other intangibles, in each case for Parent and its Subsidiaries (on a consolidated basis) and determined in accordance with GAAP.

Taxes shall have the meaning ascribed to it in Section 4.1(d).

Termination Date means the earlier of (a) the Maturity Date or (b) the date specified by either Agent in accordance with Section 9.1 hereof.

Texas Credit Code means Title 79, Texas Revised Civil Statutes, 1925, as amended.

Total Canadian Exposure means, at any time and without duplication, the sum of the aggregate principal amounts of the then outstanding Canadian Loans, then outstanding Bankers' Acceptance Liabilities and then outstanding Letter of Credit Liabilities in respect of Canadian Letters of Credit, in each case expressed in Dollars using, where applicable, the then current Exchange Rate.

Total Capitalization means the sum of, without duplication, (i) Borrowed Money Indebtedness (exclusive of the categories of Borrowed Money Indebtedness, other than obligations in respect of bankers' acceptances, described in clauses (vii), (viii) and (ix) of the definition of "Borrowed Money Indebtedness" set forth in this Section 1.1) of Parent and its Subsidiaries, on a consolidated basis and (ii) preferred stock and the consolidated stockholders' equity (including paid-in capital and retained earnings) of Parent and its Subsidiaries, on a consolidated basis, determined in accordance with GAAP.

Unfunded Liabilities means, with respect to any Plan, at any time, the amount (if any) by which (a) the present value of all benefits under such Plan exceeds (b) the fair market value of all Plan assets allocable to such benefits, all determined as of the then most recent actuarial valuation report for such Plan, but only to the extent that such excess represents a potential liability of any member of the Controlled Group to the PBGC or a Plan under Title IV of ERISA or under Applicable Canadian Pension Legislation. With respect to multi-employer Plans, the term "Unfunded Liabilities" shall also include contingent liability for withdrawal liability under Section 4201 of ERISA or under Applicable Canadian Pension Legislation to all multi-employer Plans to

22

which any Borrower or any member of a Controlled Group for employees of any Borrower contributes in the event of complete withdrawal from such plans.

28

U.S. Commitment means, as to any U.S. Lender, the obligation, if any, of such U.S. Lender to make U.S. Loans and incur or participate in Letter of Credit Liabilities relating to U.S. Letters of Credit in an aggregate principal amount at any one time outstanding up to (but not exceeding) the amount, if any, set forth opposite such U.S. Lender's name on the signature pages hereof under the caption "U.S. Commitment", or otherwise provided for in an Assignment and Acceptance Agreement (as the same may be increased or reduced from time to time pursuant to Section 2.4 hereof).

U.S. Lender means each lender signatory hereto with (i) prior to the Termination Date, a U.S. Commitment and (ii) on and after the Termination Date, any outstanding U.S. Obligations.

U.S. Letter of Credit shall have the meaning assigned to such term in Section 2.2 hereof.

U.S. Loan means a Loan made pursuant to Section 2.1(a) hereof.

U.S. Notes means the Notes of U.S. Borrower evidencing the U.S. Loans, in the form of Exhibit D hereto.

U.S. Obligations means, as at any date of determination thereof, the sum of the following (determined without duplication): (i) the aggregate principal amount of U.S. Loans outstanding hereunder on such date plus (ii) the aggregate amount of the Letter of Credit Liabilities outstanding on such date relating to U.S. Letters of Credit.

1.2 Miscellaneous. The words "hereof," "herein," and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not any particular provision of this Agreement. The term "annualized" as used herein shall mean the multiplication of the applicable amount for any given period by a fraction, the numerator of which is 365 and the denominator of which is the number of days elapsed in such period.

2. Commitments; Loans; BA's and Letters of Credit.

2.1 Loans and BA's. Each Lender severally agrees, subject to all of the terms and conditions of this Agreement (including, without limitation, Sections 5.1 and 5.2 hereof), to make Loans and, in the case of Canadian Lenders, to accept and purchase Bankers' Acceptances, as follows:

(a) U.S. Loans. From time to time on or after the Effective Date and during the Availability Period, each U.S. Lender shall make loans under this Section 2.1(a) to U.S. Borrower in an aggregate principal amount at any one time outstanding (including its Commitment Percentage of all Letter of Credit Liabilities relating to U.S. Letters of Credit at such time) up to but not exceeding such U.S. Lender's Commitment Percentage of the Maximum U.S. Available Amount.

Subject to the conditions in this Agreement, any such U.S. Loan repaid prior to the Termination Date may be reborrowed pursuant to the terms of this Agreement; provided, that any and all such U.S. Loans shall be due and payable in full at the end of the Availability Period. Loans made under this Section 2.1(a) shall be made and denominated in Dollars. The aggregate of all U.S. Loans to be made by the U.S. Lenders in connection with a particular borrowing shall be equal to the lesser of (i) the unutilized portion of the U.S. Commitments or (ii) \$1,000,000 or any integral multiple of \$100,000 in excess thereof.

29

Canadian Loans. From time to time on or after the Effective Date (b) and during the Availability Period, each Canadian Lender shall make loans under this Section 2.1(b) to Canadian Borrower in an aggregate principal amount at any one time outstanding (including such Canadian Lender's Commitment Percentage of all Bankers' Acceptance Liabilities and all Letter of Credit Liabilities relating to Canadian Letters of Credit at such time) up to but not exceeding such Canadian Lender's Commitment Percentage of the Maximum Canadian Available Amount. Subject to the conditions in this Agreement, any such Canadian Loan repaid prior to the Termination Date may be reborrowed pursuant to the terms of this Agreement; provided, that any and all such Canadian Loans shall be due and payable in full at the end of the Availability Period. Loans made under this Section 2.1(b) may, at the option of Canadian Borrower, be made and denominated either in Dollars or in Canadian Dollars (but all Loans to be made under a particular borrowing must be made and denominated in the same currency). The aggregate of all Canadian Loans to be made by the Canadian Lenders in connection with a particular borrowing shall be equal to the lesser of (i) the unutilized portion of the Canadian Commitments or (ii)\$1,000,000 or any integral multiple of \$100,000 in excess thereof (if the Loans are denominated in Dollars) or C\$1,000,000 or any integral multiple of C\$100,000 in excess thereof (if the Loans are denominated in Canadian Dollars).

(c) Bankers' Acceptances. From time to time on or after the Effective Date and during the Availability Period, each Canadian Lender shall accept and purchase Bankers' Acceptances drawn on it under Section 2.3 hereof by Canadian Borrower in an aggregate principal amount at any one time outstanding (including such Canadian Lender's Commitment Percentage of all Canadian Loans outstanding at such time and all Letter of Credit Liabilities relating to Canadian Letters of Credit at such time) up to but not exceeding such Canadian Lender's Commitment Percentage of the Maximum Canadian Available Amount. No Bankers' Acceptance may be made or accepted on or after the Termination Date and all outstanding Bankers' Acceptances shall mature no later than the end of the Availability Period. Loans made by way of Bankers' Acceptances shall be made and denominated in Canadian Dollars.

(d) Borrowers, Agents and the Lenders agree that Chapter 15 of the Texas Credit Code shall not apply to this Agreement, the Notes or any other Obligation.

2.2 Letters of Credit.

30

Letters of Credit. Subject to the terms and conditions of this (a) Agreement, and on the condition that aggregate Letter of Credit Liabilities relating to U.S. Letters of Credit shall never exceed \$25,000,000 and that aggregate Letter of Credit Liabilities relating to Canadian Letters of Credit shall never exceed \$25,000,000, (i) U.S. Borrower shall have the right, in addition to U.S. Loans provided for in Section 2.1(a) hereof, to utilize the U.S. Commitments from time to time during the Availability Period by obtaining the issuance of standby letters of credit for the account of U.S. Borrower if U.S. Borrower shall so request in the notice referred to in Section 2.2(b)(i) hereof (such standby letters of credit as any of them may be amended, supplemented, extended or confirmed from time to time, being herein collectively called the "U.S. Letters of Credit") and Canadian Borrower shall have the right, in addition to Canadian Loans provided for in Section 2.1(b) hereof and Bankers' Acceptances provided for in Section 2.1(c) hereof, to utilize the Canadian Commitments from time to time during the Availability Period by obtaining the issuance of standby letters of credit for the account of Canadian Borrower if Canadian Borrower shall so request in the notice referred to in Section 2.2(b)(i) hereof (such standby letters of credit as any of them may be amended, supplemented, extended or confirmed from time to time, being herein collectively called the "Canadian Letters of Credit") and (ii) TCB agrees to issue U.S. Letters of Credit and The Bank of Nova Scotia agrees to issue Canadian Letters of Credit. The Letters of Credit will, at the request of the applicable Borrower, be issued in currencies other than those expressly provided for in this Agreement so long as the applicable Agent is reasonably satisfied that such currency is readily available in the required amounts and that such currency selection is not otherwise disadvantageous to any Agent or any Lender. Upon the date of the issuance of a Letter of Credit, the applicable Issuer shall be deemed, without further action by any party hereto, to have sold to each U.S. Lender or Canadian Lender, as the case may be, and each such U.S. Lender or Canadian Lender, as the case may be, shall be deemed, without further action by any party hereto, to have purchased from the applicable Issuer, a participation, to the extent of such Lender's Commitment Percentage, in such Letter of Credit and the related Letter of Credit Liabilities, which participation shall terminate on the earlier of the expiration date of such Letter of Credit or the Termination Date. Any Letter of Credit that shall have an expiration date after the end of the Availability Period shall be subject to Cover or backed by a standby letter of credit in form and substance, and issued by a Person, acceptable to the applicable Agent in its sole discretion. TCB or, with the prior approval of U.S. Borrower, U.S. Agent and the applicable U.S. Lender, another U.S. Lender shall be the Issuer of each U.S. Letter of Credit and The Bank of Nova Scotia or, with the prior approval of Canadian Borrower, Canadian Agent and the applicable Canadian Lender, another Canadian Lender shall be the Issuer of each Canadian Letter of Credit. Except as provided above, all U.S. Letters of Credit shall be denominated in Dollars and all Canadian Letters of Credit shall, at the option of Canadian Borrower, be denominated in either Dollars or Canadian Dollars. Fees due in respect of a U.S. Letter of Credit shall be payable in Dollars and fees due in respect of a Canadian Letter of Credit shall be payable (i) in Dollars, if such Letter of Credit is denominated in Dollars and (ii) in Canadian Dollars if such Letter of Credit is denominated in Canadian Dollars or any other currency.

(b) Additional Provisions. The following additional provisions shall apply to each Letter of Credit:

31

(i) U.S. Borrower or Canadian Borrower, as the case may be, shall give the appropriate Agent notice requesting each issuance of a Letter of Credit hereunder as provided in Section 4.3 hereof and shall furnish such additional information regarding such transaction as such Agent may reasonably request. Upon receipt of such notice, such Agent shall promptly notify each U.S. Lender or Canadian Lender, as the case may be, of the contents thereof and of such Lender's Commitment Percentage of the amount of such proposed Letter of Credit.

No U.S. Letter of Credit may be issued if after giving (ii) effect thereto the sum of (A) the aggregate outstanding principal amount of U.S. Loans plus (B) the aggregate Letter of Credit Liabilities relating to U.S. Letters of Credit would exceed the Maximum U.S. Available Amount. No Canadian Letter of Credit may be issued if after giving effect thereto the sum of (A) the aggregate outstanding principal amount of Canadian Loans plus (B) the aggregate Letter of Credit Liabilities relating to Canadian Letters of Credit plus (C) the aggregate Bankers' Acceptance Liabilities would exceed the Maximum Canadian Available Amount. On each day during the period commencing with the issuance of any Letter of Credit and until such Letter of Credit shall have expired or been terminated, the U. S. Commitment or Canadian Commitment, as the case may be, of each applicable Lender shall be deemed to be utilized for all purposes hereof in an amount equal to such Lender's Commitment Percentage of the amount then available for drawings under such Letter of Credit (and any unreimbursed drawings under such Letter of Credit).

(iii) Upon receipt from the beneficiary of any Letter of Credit of any demand for payment thereunder, the applicable Issuer shall notify the Agents and thereafter the U.S. Agent or the Canadian Agent, as the case may be, shall promptly notify the applicable Borrower and each applicable Lender as to the amount to be paid as a result of such demand and the payment date therefor. If at any time prior to the expiration date of a Letter of Credit any applicable Issuer shall have made a payment to a beneficiary of a Letter of Credit in respect of a drawing under such Letter of Credit, each applicable Lender will pay to the U.S. Agent or the Canadian Agent, as the case may be, immediately upon demand by such Issuer at any time during the period commencing after such payment until reimbursement thereof in full by the applicable Borrower, an amount equal to such Lender's U.S. Commitment Percentage or Canadian Commitment Percentage, as the case may be, of such payment, together with interest on such amount for each day from the date of demand for such payment (or, if such demand is made after 11:00 a.m. Houston, Texas time (in the case of a U.S. Letter of Credit) or 12:00 noon Toronto, Ontario time (in the case of a Canadian Letter of Credit) on such date, from the next succeeding Business Day) to the date of payment by such Lender of such amount at a rate of interest per annum equal to (i) in respect of U.S. Letters of Credit, the Federal Funds Rate, (ii) in respect of Canadian Letters of Credit which are denominated in Dollars, the Base Rate plus two percent (2%) and (iii) in respect of

Canadian Letters of Credit which are denominated in Canadian Dollars, the CDOR Rate. To the extent that it is ultimately determined that the applicable Borrower is relieved of its obligation to reimburse the applicable Issuer because of such Issuer's gross negligence or willful misconduct in determining that documents received under any applicable Letter of Credit comply with the terms thereof, the applicable Issuer shall be obligated to refund to the paying Lenders all amounts paid to such Issuer to reimburse Issuer for the applicable drawing under such Letter of Credit.

U.S. Borrower or the Canadian Borrower, as the case may (iv) be, shall be irrevocably and unconditionally obligated forthwith to reimburse the appropriate Agent, on the date on which such Agent notifies U.S. Borrower or the Canadian Borrower, as the case may be, of the date and amount of any payment by the applicable Issuer of any drawing under a Letter of Credit, for the amount paid by such Issuer upon such drawing, without presentment, demand, protest or other formalities of any kind, all of which are hereby waived. Such reimbursement may, subject to satisfaction of the conditions in Sections 5.1 and 5.2 hereof and to the Maximum U.S. Available Amount or Maximum Canadian Available Amount, as the case may be (after adjustment in the same to reflect the elimination of the corresponding Letter of Credit Liability), be made by the borrowing of Loans or, in the case of the Canadian Borrower, by the issuance, acceptance and purchase of Bankers' Acceptances. The applicable Agent will pay to each Lender such Lender's Commitment Percentage of all amounts received from U.S. Borrower or the Canadian Borrower, as the case may be, for application in payment, in whole or in part, of the Reimbursement Obligation in respect of any Letter of Credit, but only to the extent such Lender has made payment to the applicable Agent in respect of such Letter of Credit pursuant to clause (iii) above.

U.S. Borrower or the Canadian Borrower, as the case may (v) be, will pay to the appropriate Agent at the Principal Office of such Agent for the account of each applicable Lender a letter of credit fee with respect to each Letter of Credit equal to the greater of (x) \$500 or (y) the Margin Percentage applicable to LIBOR Borrowings multiplied by the daily average amount available for drawings under each Letter of Credit (and computed on the basis of the actual number of days elapsed in a year composed of 360 days), in each case for the period from and including the date of issuance of such Letter of Credit to and including the date of expiration or termination thereof, such fee to be due and payable quarterly in advance. After the occurrence of an Event of Default which has not been cured or waived, the letter of credit fee provided for the preceding sentence shall be increased by two percent (2%) per annum, such increased fee to be due and payable upon demand by the applicable Agent. In the event any Letter of Credit is drawn, that portion of the applicable letter of credit fee provided for in the preceding sentence relating to the period beyond the date of such drawing shall be credited to the applicable Borrower's Reimbursement Obligations relating thereto. The applicable Agent will pay to each applicable Lender, promptly after receiving any payment in respect of letter of credit fees referred to in this clause (v), an amount equal to the product of such Lender's U.S. Commitment Percentage or Canadian

Commitment Percentage, as the case may be, times the amount of such fees. In addition to and cumulative of the above described fees, U.S. Borrower or the Canadian Borrower, as the case may be, shall pay to the appropriate Agent, for the account of the applicable Issuer, in advance on the date of the issuance of the applicable Letter of Credit, a fee in an amount equal to 1/8% of the face amount of the applicable Letter of Credit (such fee to be retained by the applicable Issuer for its own account).

(vi) The issuance by the applicable Issuer of each Letter of Credit shall, in addition to the conditions precedent set forth in Section 5 hereof, be subject to the conditions precedent (A) that such Letter of Credit shall be in such form and contain such terms as shall be reasonably satisfactory to applicable Agent, and (B) that U.S. Borrower or the Canadian Borrower, as the case may be, shall have executed and delivered such Applications and other instruments and agreements relating to such Letter of Credit as the applicable Agent shall have reasonably requested and are not inconsistent with the terms of this Agreement. In the event of a conflict between the terms of this Agreement and the terms of any Application, the terms hereof shall control.

(vii) Each Issuer will send to U.S. Borrower or the Canadian Borrower, as the case may be, and each applicable Lender, immediately upon issuance of any Letter of Credit issued by such Issuer or any amendment thereto, a true and correct copy of such Letter of Credit or amendment.

Indemnification; Release. U.S. Borrower or the Canadian (c) Borrower, as the case may be, hereby indemnifies and holds harmless each Agent, each Lender and each Issuer from and against any and all claims and damages, losses, liabilities, costs or expenses which such Agent, such Lender or such Issuer may incur (or which may be claimed against such Agent, such Lender or such Issuer by any Person whatsoever), REGARDLESS OF WHETHER CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE OF ANY OF THE INDEMNIFIED PARTIES, in connection with the execution and delivery of any Letter of Credit or transfer of or payment or failure to pay under any Letter of Credit; provided that U.S. Borrower or the Canadian Borrower, as the case may be, shall not be required to indemnify any party seeking indemnification for any claims, damages, losses, liabilities, costs or expenses to the extent, but only to the extent, caused by (i) the willful misconduct or gross negligence of the party seeking indemnification, or (ii) the failure by the party seeking indemnification to pay under any Letter of Credit after the presentation to it of a request required to be paid under applicable law. U.S. Borrower or the Canadian Borrower, as the case may be, hereby releases, waives and discharges each Agent, each Lender and each Issuer from any claims, causes of action, damages, losses, liabilities, reasonable costs or expenses which may now exist or may hereafter arise, REGARDLESS OF WHETHER CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE OF ANY OF THE INDEMNIFIED PARTIES, by reason of or in

connection with the failure of any other Lender to fulfill or comply with its obligations to such Agent, such Lender or such Issuer, as the case may be, hereunder (but nothing herein contained shall affect any rights U.S. Borrower or the Canadian Borrower, as the case may be, may have against such defaulting Lender or may have in respect of gross negligence or willful misconduct). Nothing in this Section 2.2(c) is intended to limit the obligations of U.S. Borrower or the Canadian Borrower, as the case may be, under any other provision of this Agreement.

Additional Costs in Respect of Letters of Credit. If as a result (d) of any Regulatory Change there shall be imposed, modified or deemed applicable any tax (other than any tax based on or measured by net income), reserve, special deposit or similar requirement against or with respect to or measured by reference to Letters of Credit issued or to be issued hereunder or participations in such Letters of Credit, and the result shall be to increase the cost to any Lender of issuing or maintaining any Letter of Credit or any participation therein, or materially reduce any amount receivable by any Lender hereunder in respect of any Letter of Credit or any participation therein (which increase in cost, or reduction in amount receivable, shall be the result of such Lender's reasonable allocation of the aggregate of such increases or reductions resulting from such event), then such Lender shall notify U.S. Borrower or the Canadian Borrower, as the case may be, through the appropriate Agent (which notice shall be accompanied by a statement setting forth in reasonable detail the basis for the determination of the amount due), and within 15 Business Days after demand therefor by such Lender through such Agent, U.S. Borrower or the Canadian Borrower, as the case may be, shall pay to such Lender, from time to time as specified by such Lender, such additional amounts as shall be sufficient to compensate such Lender for such increased costs or reductions in amount. Such statement as to such increased costs or reductions in amount incurred by such Lender, submitted by such Lender to U.S. Borrower or the Canadian Borrower, as the case may be, shall be conclusive as to the amount thereof, absent manifest error, and may be computed using any reasonable averaging and attribution method. Each Lender will notify U.S. Borrower or the Canadian Borrower, as the case may be, through the appropriate Agent of any event occurring after the date of this Agreement which will entitle such Lender to compensation pursuant to this Section as promptly as practicable after any executive officer of such Lender obtains knowledge thereof and determines to request such compensation, and (if so requested by U.S. Borrower or the Canadian Borrower, as the case may be, through the appropriate Agent) will designate a different lending office of such Lender for the issuance or maintenance of Letters of Credit by such Lender or will take such other action as U.S. Borrower or the Canadian Borrower, as the case may be, may reasonably request if such designation or action is consistent with the internal policy of such Lender and legal and regulatory restrictions, can be undertaken at no additional cost, will avoid the need for, or reduce the amount of, such compensation and will not, in the sole opinion of such Lender, be disadvantageous to such Lender (provided that no such U.S. Lender shall have any obligation so to designate a different lending office which is not located in the United States of America and no such Canadian Lender shall have any obligation so to designate a different lending office which is not located in Canada).

29

35

(a) Subject to the terms and conditions hereof, each Canadian Lender severally agrees to accept and purchase Bankers' Acceptances drawn upon it by the Canadian Borrower denominated in Canadian Dollars. The Canadian Borrower shall notify the Canadian Agent by irrevocable written notice (each a "Bankers' Acceptance Notice") by 12:00 noon (Toronto, Ontario time) two (2) Business Days prior to the proposed date of any borrowing by way of Bankers' Acceptances. Each borrowing by way of Bankers' Acceptances shall be in a minimum aggregate face amount of C\$1,000,000.00 and integral multiples of C\$100,000.00 in excess thereof. The face amount of each Bankers' Acceptance shall be C\$100,000.00 or any integral multiple thereof. Each Bankers' Acceptance Notice shall be in the form of Exhibit G.

(1) Bankers' Acceptances shall be issued and shall mature on a Business Day. Each Bankers' Acceptance shall have a term of 30, 60, 90 or, if available, 180 days excluding days of grace and shall mature on or before the Maturity Date and shall be in form and substance reasonably satisfactory to each Canadian Lender.

To facilitate the acceptance of Bankers' Acceptances under (2)this Agreement, the Canadian Borrower shall, upon execution of this Agreement and from time to time as required, provide to the Canadian Agent drafts, in form satisfactory to the Canadian Agent, duly executed and endorsed in blank by the Canadian Borrower in quantities sufficient for each Canadian Lender to fulfill its obligations hereunder. In addition, the Canadian Borrower hereby appoints each Canadian Lender as its attorney to sign and endorse on its behalf, in handwriting or by facsimile or mechanical signature as and when deemed necessary by such Canadian Lender, blank forms of Bankers' Acceptances. The Canadian Borrower recognizes and agrees that all Bankers' Acceptances signed and/or endorsed on its behalf by a Canadian Lender shall bind the Canadian Borrower as fully and effectually as if signed in the handwriting of and duly issued by the proper signing officer of the Canadian Borrower. Each Canadian Lender is hereby authorized to issue such Bankers' Acceptances endorsed in blank in such face amounts as may be determined by such Canadian Lender provided that the aggregate amount thereof is equal to the aggregate amount of Bankers' Acceptances required to be accepted by such Canadian Lender. No Canadian Lender shall be responsible or liable for its failure to accept a Bankers' Acceptance if the cause of such failure is, in whole or in part, due to the failure of the Canadian Borrower to provide duly executed and endorsed drafts to the Canadian Agent on a timely basis nor shall any Canadian Lender be liable for any damage, loss or other claim arising by reason of any loss or improper use of any such instrument except loss or improper use arising by reason of the gross negligence or willful misconduct of such Canadian Lender, its officers, employees, agents or representatives. Each Canadian Lender shall maintain a record with respect to Bankers' Acceptances (i) received by it from the Canadian Agent in blank hereunder, (ii) voided by it for any reason, (iii) accepted by it hereunder, (iv) purchased by it hereunder and (v) canceled at their respective maturities. Each Canadian Lender further agrees to retain such

records in the manner and for the statutory periods provided in the various Canadian provincial or federal statutes and regulations which apply to such Canadian Lender.

(3) Drafts of the Canadian Borrower to be accepted as Bankers' Acceptances hereunder shall be duly executed by a duly authorized officer of the Canadian Borrower. Notwithstanding that any person whose signature appears on any Bankers' Acceptance as a signatory for the Canadian Borrower may no longer be an authorized signatory for the Canadian Borrower at the date of issuance of a Bankers' Acceptance, such signature shall nevertheless be valid and sufficient for all purposes as if such authority had remained in force at the time of such issuance and any such Bankers' Acceptance so signed shall be binding on the Canadian Borrower.

(4) Promptly following receipt of a Bankers' Acceptance Notice, the Canadian Agent shall so advise the Canadian Lenders and shall advise each Canadian Lender of the face amount of each Bankers' Acceptance to be accepted by it and the term thereof. The aggregate face amount of Bankers' Acceptances to be accepted by a Canadian Lender shall be determined by the Canadian Agent by reference to the respective Canadian Commitments of the Canadian Lenders, except that, if the face amount of a Bankers' Acceptance, which would otherwise be accepted by a Canadian Lender, would not be C\$100,000.00 or an integral multiple thereof, such face amount shall be increased or reduced by the Canadian Agent in its sole and unfettered discretion to the nearest integral multiple of C\$100,000.00.

(5) Each Bankers' Acceptance to be accepted by a Canadian Lender shall be accepted at such Canadian Lender's office shown on the signature pages hereof or as otherwise designated by such Canadian Lender from time to time.

(6) On the relevant borrowing date, each Canadian Lender severally agrees to purchase from the Canadian Borrower, at the face amount thereof discounted by the Applicable BA Discount Rate, any Bankers' Acceptance accepted by it and provide to the Canadian Agent, for the account of the Canadian Borrower, the BA Discount Proceeds in respect thereof after deducting therefrom the amount of the Acceptance Fee payable by the Canadian Borrower to such Canadian Lender under Section 2.3(c) in respect of such Bankers' Acceptance.

(7) Each Canadian Lender may at any time and from time to time hold, sell, rediscount or otherwise dispose of any or all Bankers' Acceptances accepted and purchased by it.

(8) The Canadian Borrower waives presentment for payment and any other defense to payment of any amounts due to a Canadian Lender in respect of a Bankers' Acceptance accepted by it pursuant to this Agreement which might exist solely by reason of such Bankers' Acceptance being held, at the maturity thereof, by such Canadian Lender in its own right and the Canadian Borrower agrees not to claim any days of grace if such

Canadian Lender as holder sues the Canadian Borrower on the Bankers' Acceptances for payment of the amount payable by the Canadian Borrower thereunder.

With respect to each Bankers' Acceptance, the Canadian Borrower, (b) prior to the occurrence and continuation of a Default, may give irrevocable telephone or written notice (or such other method of notification as may be agreed upon between the Canadian Agent and the Canadian Borrower) to the Canadian Agent at or before 12:00 noon (Toronto, Ontario time) two (2) Business Days prior to the maturity date of such Bankers' Acceptance followed by written confirmation electronically transmitted to the Canadian Agent on the same day, of the Canadian Borrower's intention to issue one or more Bankers' Acceptance on such maturity date (each a "Refunding Bankers' Acceptance") to provide for the payment of such maturing Bankers' Acceptance (it being understood that payments by the Canadian Borrower and fundings by the Canadian Lenders in respect of each maturing Bankers' Acceptance and each related Refunding Bankers' Acceptance shall be made on a net basis reflecting the difference between the face amount of such maturing Bankers' Acceptance and the BA Discount Proceeds (net of the applicable Acceptance Fee) of such Refunding Bankers' Acceptance). Any funding on account of any maturing Bankers' Acceptance must be made at or before 12:00 noon (Toronto, Ontario time) on the maturity date of such Bankers' Acceptance. If the Canadian Borrower fails to give such notice, then subject to satisfaction of the conditions in Section 5 hereof and to the Maximum Canadian Available Amount, the Canadian Borrower shall be irrevocably deemed to have requested and to have been advanced a Canadian Prime Loan in the face amount of such maturing Bankers' Acceptance on the maturity date of such Bankers' Acceptance from the Canadian Lender which accepted such maturing Bankers' Acceptance, which Canadian Prime Loan shall thereafter bear interest as such in accordance with the provisions hereof until paid in full.

(c) An Acceptance Fee shall be payable by the Canadian Borrower to each Canadian Lender in advance (in the manner specified in Section 2.3(a)(6)) in respect of, and as a condition precedent to the acceptance by such Canadian Lender of, a Bankers' Acceptance to be accepted by such Canadian Lender calculated at the rate per annum equal to the Margin Percentage applicable to LIBOR Borrowings, calculated on the face amount of such Bankers' Acceptance and computed on the basis of the number of days in the term of such Bankers' Acceptance and a year of 365 days.

2.4 Terminations, Reductions or Reallocations of Commitments.

(a) Mandatory. On the Termination Date, all Commitments shall be terminated in their entirety.

(b) Optional. U.S. Borrower or Canadian Borrower, as the case may be, shall have the right to terminate or reduce the unused portion of the U.S. Commitments or the Canadian Commitments, as the case may be, at any time or from time to time, provided that (i) U.S. Borrower or Canadian Borrower, as the case may be, shall give notice of each such termination or reduction to the appropriate Agent as provided in Section 4.3 hereof and (ii) each such partial reduction shall be in an integral multiple of \$5,000,000. Notwithstanding the foregoing, U.S. Borrower may not

32

reduce the U.S. Commitments below the then outstanding principal balance of the U.S. Obligations and Canadian Borrower may not reduce the Canadian Commitments below the then outstanding principal balance of the Canadian Obligations. No termination or reduction of the Commitments pursuant to this provision may be reinstated without the prior written approval of Agents and the Lenders.

38

Reallocations. Any Dual Lender may agree with Borrowers to (C) reallocate its existing U.S. Commitment and Canadian Commitment, so long as the sum of such U.S. Commitment and Canadian Commitment remains unchanged. In addition, with the prior written consent of all of the Dual Lenders, any U.S. Lender may agree with Borrowers to convert a portion of its U.S. Commitment into a Canadian Commitment, thereby becoming a Dual Lender, and any Canadian Lender may agree with Borrowers to convert a portion of its Canadian Commitment into a U.S. Commitment, in each case so long as (i) each Lender continues to be a U.S. Lender with a U.S. Commitment of at least \$1,000,000 and (ii) the sum of such Lender's U.S. Commitment and Canadian Commitment remains equal to the aggregate amount of such Lender's U.S. Commitment and Canadian Commitment prior to such reallocation. Borrowers shall give written notice to the Agents of any reallocation pursuant to this provision at least ten (10) Business Days prior to the effective date of any such reallocation. No Lender shall be required to agree to any such reallocation, but may do so at its option, in its sole discretion. The following conditions precedent must be satisfied prior to any such reallocation becoming effective:

(1) no Default or Event of Default shall have occurred and be continuing;

(2) if, as a result of any such reallocation, the aggregate U.S. Obligations would exceed the aggregate of all of the U.S. Commitments, then the U.S. Borrower shall, on the effective date of such reallocation, repay or prepay U.S. Loans (or provide Cover for Letter of Credit Liabilities relating to U.S. Letters of Credit) in accordance with this Agreement in an aggregate principal amount such that, after giving effect thereto, the aggregate U.S. Obligations shall not exceed the aggregate of all of the U.S. Commitments;

(3) if, as a result of any such reallocation, the Total Canadian Exposure would exceed the aggregate of all of the Canadian Commitments, then the Canadian Borrower shall, on the effective date of such reallocation, repay or prepay Canadian Loans (or provide Cover for Letter of Credit Liabilities relating to Canadian Letters of Credit or for Bankers' Acceptance Liabilities) in accordance with this Agreement in an aggregate principal amount such that, after giving effect thereto, the Total Canadian Exposure shall not exceed the aggregate of all of the Canadian Commitments;

(4) Borrowers shall have paid any amounts (or shall have provided Cover) due under Sections 2.9(c) or (d) hereof on the date of such reallocation;

(5) The Maximum Canadian Available Amount shall be adjusted to equal the sum of all of the Canadian Commitments after giving effect to such reallocation and the

Maximum U.S. Available Amount shall be adjusted to equal the sum of all of the U.S. Commitments after giving effect to such reallocation;

(6) participations by the Lenders in the outstanding Letters of Credit and the Letter of Credit Liabilities and the outstanding Loans of the Lenders shall be adjusted to give effect to such reallocation; provided, however, that in lieu of requiring any prepayment of any Bankers' Acceptances in order to make appropriate adjustments to give effect to such reallocations, Canadian Borrower shall be required to provide additional Cover for any applicable portion of the Bankers' Acceptance Liabilities;

(7) each Lender whose U.S. Commitment or Canadian Commitment shall be the subject of any reallocation shall have received from the Borrowers a fee equal to the greater of \$3,000.00 or 1/16% of the amount of the increase or decrease, as the case be, in its Canadian Commitment.

2.5 Commitment Fees.

(a) U.S. Borrower shall pay to U.S. Agent for the account of each U.S. Lender, and Canadian Borrower shall pay to Canadian Agent for the account of each Canadian Lender, commitment fees for the period from the Effective Date to and including the Termination Date at a rate per annum equal to the Commitment Fee Percentage. Such commitment fees shall be computed (on the basis of the actual number of days elapsed in a year composed of 360 days) on each day and shall be based on the excess of (x) the aggregate amount of each Lender's Commitment for such day over (y) the sum of (i) the aggregate unpaid principal balance (in Dollars) of such Lender's Note or Notes on such day plus (ii) the aggregate Letter of Credit Liabilities as to such Lender for such day plus, in the case of Canadian Lenders only, (iii) the aggregate Bankers' Acceptance Liabilities outstanding on such day. Accrued commitment fees shall be payable in arrears on the Quarterly Dates prior to the Termination Date and on the Termination Date, with any Canadian Obligations converted to Dollars at the Exchange Rate on each such date for the purposes of each such calculation.

(b) All past due fees payable under this Section shall bear interest at the Past Due Rate.

2.6 Several Obligations. The failure of any Lender to make any Loan to be made by it or to accept and purchase any Bankers' Acceptance required to be so accepted and purchased by it on the date specified therefor shall not relieve any other Lender of its obligation to make its Loan or to accept and purchase its Bankers' Acceptance on such date, but neither any Agent nor any Lender shall be responsible or liable for the failure of any other Lender to make a Loan or to accept and purchase any Bankers' Acceptance or to participate in, or co-issue, any Letter of Credit. Notwithstanding anything contained herein to the contrary, (i) if a U.S. Lender fails to make a U.S. Loan as and when required hereunder, then upon each subsequent event which would otherwise result in payments of principal being made to the defaulting U.S. Lender, the amount which would have been paid to the defaulting U.S. Lender shall be divided among the non-defaulting U.S. Lenders

34

ratably according to their respective Commitment Percentages until the Obligations of each U.S. Lender (including the defaulting U.S. Lender) are equal to such U.S. Lender's Commitment Percentage of the total U.S. Obligations and (ii) if a Canadian Lender fails to make a Canadian Loan or accept and purchase any Bankers' Acceptance as and when required hereunder, then upon each subsequent event which would otherwise result in payments of principal being made to the defaulting Canadian Lender, the amount which would have been paid to the defaulting Canadian Lender shall be divided among the non-defaulting Canadian Lenders ratably according to their respective Commitment Percentages until the Obligations of each Canadian Lender (including the defaulting Canadian Lender) are equal to such Canadian Lender's Commitment Percentage of the total Canadian Obligations.

Notes. The U.S. Loans made by each U.S. Lender shall be 2.7 evidenced by a single U.S. Note of U.S. Borrower in substantially the form of Exhibit D hereto payable to the order of such U.S. Lender in a principal amount equal to the U.S. Commitment of such U.S. Lender, and otherwise duly completed. The Canadian Loans made by each Canadian Lender which are denominated in Dollars shall be evidenced by a single Canadian Note of Canadian Borrower in substantially the form of Exhibit C hereto payable to the order of such Canadian Lender in a principal amount equal to the Canadian Commitment of such Canadian Lender, and otherwise duly completed. The Canadian Prime Loans made by each Canadian Lender shall be evidenced by a single Canadian Dollar Note of Canadian Borrower in substantially the form of Exhibit H hereto payable to the order of such Canadian Lender in a principal amount equal to two times the Canadian Commitment of such Canadian Lender, and otherwise duly completed. The promissory notes described in this Section are each, together with all renewals, extensions, modifications and replacements thereof and substitutions therefor, called a "Note" and collectively called the "Notes". Each Lender is hereby authorized by each Borrower to endorse on the schedule (or a continuation thereof) that may be attached to each Note of such Lender, to the extent applicable, the date, amount, type of and the applicable period of interest for each Loan made by such Lender to the applicable Borrower hereunder, and the amount of each payment or prepayment of principal of such Loan received by such Lender, provided, that any failure by such Lender to make any such endorsement shall not affect the obligations of any Borrower under such Note or hereunder in respect of such Loan.

2.8 Use of Proceeds. The proceeds of the Loans and of the acceptance and purchase of Bankers' Acceptances shall be used by the Borrowers for acquisitions, capital expenditures and other working capital and general corporate purposes. Neither any Agent nor any Lender shall have any responsibility as to the use of any proceeds of the Loans or of the acceptance and purchase of Bankers' Acceptances.

2.9 Currency Fluctuations.

(a) Not later than 1:00 p.m. (Houston, Texas time) on each Calculation Date, the U.S. Agent shall determine the Exchange Rate as of such Calculation Date. The Exchange Rate so determined shall become effective on the first Business Day immediately following the relevant

35

Calculation Date (a "Reset Date") and shall remain effective until the next succeeding Calculation Date.

(b) Not later than 4:00 p.m. (Houston, Texas time) on each Reset Date, the U.S. Agent shall consult with the Canadian Agent and the Agents shall determine the Total Canadian Exposure and the aggregate U.S. Obligations.

(c) If, on any Reset Date or on the date of any reallocation of the U.S. Commitments and the Canadian Commitments pursuant to Section 2.4(c) hereof, the sum of the aggregate U.S. Obligations and the Total Canadian Exposure exceeds the aggregate of all of the U.S. Commitments and the Canadian Commitments by five percent (5%) or more, then (i) the Agents shall give notice thereof to the Lenders and Borrowers and (ii) the Borrowers shall within two (2) Business Days thereafter, repay or prepay Loans (or provide Cover for Letter of Credit Liabilities or Bankers' Acceptance Liabilities) in accordance with this Agreement in an aggregate principal amount sufficient to reduce the sum of the aggregate U.S. Obligations and the Total Canadian Exposure to the aggregate of all of the U.S. Commitments and the Canadian Commitments.

(d) If, on any day prior to the Termination Date, the Total Canadian Exposure exceeds the aggregate of all of the Canadian Commitments by five percent (5%) or more, then (i) the Canadian Agent shall give notice thereof to the Canadian Borrower and the Canadian Lenders and (ii) within two (2) Business Days thereafter, the Canadian Borrower shall repay or prepay Canadian Loans (or provide Cover for Letter of Credit Liabilities relating to Canadian Letters of Credit or Bankers' Acceptance Liabilities) in accordance with this Agreement in an aggregate principal amount such that, after giving effect thereto, the Total Canadian Exposure shall not exceed the aggregate of all of the Canadian Commitments.

3. Borrowings, Prepayments and Interest Options.

41

Borrowings. The applicable Borrower shall give the applicable 3.1 Agent notice of each borrowing to be made hereunder as provided in Section 4.3 hereof and the applicable Agent shall promptly notify each applicable Lender of such request. Not later than 12:00 noon Houston, Texas time (in the case of U.S. Loans) or 1:00 p.m. Toronto, Ontario time (in the case of Canadian Loans which are not same day fundings and Bankers' Acceptances) or 2:00 p.m. Toronto, Ontario time (in the case of Canadian Loans which are same day fundings) on the date specified for each such borrowing hereunder, each applicable Lender shall make available the amount of the Loan, if any, to be made by it on such date and/or the proceeds of the acceptance and purchase of any Bankers' Acceptances, if any, to be so accepted and purchased by it on such date to the applicable Agent at its Principal Office, in immediately available funds, for the account of the applicable Borrower. Such amounts received by the applicable Agent will be held in an account maintained by the applicable Borrower with the applicable Agent. The amounts so received by the applicable Agent shall, subject to the terms and conditions of this Agreement, be made available to the applicable Borrower by wiring or otherwise transferring, in immediately available funds, such amount to an account designated by the applicable Borrower and approved by the applicable Agent.

3.2 Prepayments.

42

(a) Optional Prepayments. Except as provided in Section 3.3 hereof, each Borrower shall have the right to prepay, on any Business Day, in whole or in part, without the payment of any penalty or fee, any of the Obligations (other than Obligations relating to Bankers' Acceptances) at any time or from time to time, provided that the applicable Borrower shall give the applicable Agent notice of each such prepayment as provided in Section 4.3 hereof. Each optional prepayment on a Loan shall be in an amount equal to an integral multiple of \$3,000,000 (in respect of Loans denominated in Dollars) or C\$3,000,000 (in respect of Loans denominated in Canadian Dollars). Bankers' Acceptances may not be prepaid.

(b) Interest Payments. Accrued and unpaid interest on the unpaid principal balance of the Notes shall be due and payable on the Interest Payment Dates.

Payments and Interest on Reimbursement Obligations. Each (C) Borrower will pay to the applicable Agent for the account of each applicable Lender the amount of each Reimbursement Obligation with respect to such Borrower on the date on which the applicable Agent notifies the applicable Borrower of the date and amount of the applicable payment by the Issuer of any drawing under a Letter of Credit or payment of any Bankers' Acceptance on maturity. The amount of any Reimbursement Obligation may, if the applicable conditions precedent specified in Sections 5.1 and 5.2 hereof have been satisfied, be paid with the proceeds of Loans or, in the case of Canadian Obligations, of the acceptance and purchase or Bankers' Acceptances. Subject to Section 11.7 hereof, each Borrower will pay to the applicable Agent for the account of each applicable Lender interest on any Reimbursement Obligation at (i) at the applicable Base Rate (with respect to Reimbursement Obligations denominated in Dollars) or at the Canadian Prime Rate (with respect to Reimbursement Obligations denominated in Dollars) plus the applicable Margin Percentage from the date such Reimbursement Obligation arises until the date five (5) Business Days thereafter and (ii) at the applicable Past Due Rate thereafter until the same is paid in full.

3.3 Interest Options

Options Available. The outstanding principal balance of the (a) Canadian Dollar Notes shall bear interest at the Canadian Prime Rate and the outstanding principal balance of the other Notes shall bear interest at the applicable Base Rate; provided, that (1) all past due amounts, both principal and accrued interest, shall bear interest at the Past Due Rate, and (2) subject to the provisions hereof, each Borrower shall have the option of having all or any portion of the principal balances of its Notes (other than the Canadian Dollar Notes) from time to time outstanding bear interest at a Eurodollar Rate. The records of Agents and each of the Lenders with respect to Interest Options, Interest Periods and the amounts of Loans to which they are applicable shall be binding and conclusive, absent manifest error. Interest on the amount of each advance against the Notes shall be computed on the amount of that advance and from the date it is made. Notwithstanding anything in this Agreement to the contrary, for the full term of the Notes the interest rate produced by the aggregate of all sums paid or agreed to be paid to the holders of the Notes for the use, forbearance

or detention of the debt evidenced thereby (including all interest on the Notes at the Stated Rate plus the Additional Interest) shall not exceed the Ceiling Rate.

(b) Designation and Conversion. Each Borrower shall have the right to designate or convert its Interest Options in accordance with the provisions hereof. Provided no Event of Default has occurred and is continuing and subject to the last sentence of Section 3.3(a) and the provisions of Section 3.3(c), each Borrower may elect to have a Eurodollar Rate apply or continue to apply to all or any portion of the principal balance of its Notes (other than the Canadian Dollar Notes). Each change in Interest Options shall be a conversion of the rate of interest applicable to the specified portion of the Loans, but such conversion shall not change the respective outstanding principal balances of the applicable Notes. The Interest Options shall be designated or converted in the manner provided below:

- (i) The applicable Borrower shall give the applicable Agent telephonic notice, promptly confirmed by a Rate Designation Notice (and the applicable Agent shall promptly inform each applicable Lender thereof). Each such telephonic and written notice shall specify the amount of the Loan which is the subject of the designation, if any; the amount of borrowings into which such borrowings are to be converted or for which an Interest Option is designated; the proposed date for the designation or conversion and the Interest Period or Periods, if any, selected by the applicable Borrower. Such telephonic notice shall be irrevocable and shall be given to the applicable Agent no later than the applicable Rate Designation Date.
- No more than five (5) LIBOR Borrowings shall be in effect with respect to the U.S. Loans at any time and no more than three (3) LIBOR Borrowings shall be in effect with respect to the Canadian Loans at any time. No single LIBOR Borrowing may include both U.S. Loans and Canadian Loans.
- (iii) Each designation or conversion of a LIBOR Borrowing shall occur on a LIBOR Business Day.
- (iv) Except as provided in Section 3.3(c) hereof, no LIBOR Borrowing shall be converted to a Base Rate Borrowing or another LIBOR Borrowing on any day other than the last day of the applicable Interest Period.
- (v) Each request for a LIBOR Borrowing shall be in the amount equal to an integral multiple of \$500,000.
- (vi) Each designation of an Interest Option with respect to the U.S. Notes shall apply to all of the U.S. Notes ratably in accordance with their respective outstanding principal balances. Each designation of an Interest Option with respect to the Canadian Notes shall apply to all of the Canadian Notes ratably in accordance with their respective outstanding principal balances. If any Lender assigns an interest in any of its Notes

- when any LIBOR Borrowing is outstanding with respect thereto, then such assignee shall have its ratable interest in such LIBOR Borrowing.
- (vii) The entire outstanding principal balance of the Canadian Dollar Notes shall bear interest at the Canadian Prime Rate.
- (c) Special Provisions Applicable to LIBOR Borrowings.

Options Unlawful. If the adoption of any applicable Legal (i) Requirement after the Effective Date or any change after the Effective Date in any applicable Legal Requirement or in the interpretation or administration thereof by any Governmental Authority or compliance by any Lender with any request or directive (whether or not having the force of law) issued after the Effective Date by any central bank or other Governmental Authority shall at any time make it unlawful or impossible for any Lender to permit the establishment of or to maintain any LIBOR Borrowing, the commitment of such Lender to establish or maintain such LIBOR Borrowing shall forthwith be canceled and the applicable Borrower shall forthwith, upon demand by the applicable Agent to such Borrower, (1) convert the LIBOR Borrowing of such Lender with respect to which such demand was made to a Base Rate Borrowing; (2) pay all accrued and unpaid interest to date on the amount so converted; and (3) pay any amounts required to compensate each Lender for any additional cost or expense which any Lender may incur as a result of such adoption of or change in such Legal Requirement or in the interpretation or administration thereof and any Funding Loss which any Lender may incur as a result of such conversion. If, when any Agent so notifies any Borrower, such Borrower has given a Rate Designation Notice specifying a LIBOR Borrowing but the selected Interest Period has not yet begun, as to the applicable Lender such Rate Designation Notice shall be deemed to be of no force and effect, as if never made, and the balance of the Loans made by such Lender specified in such Rate Designation Notice shall bear interest at the Base Rate until a different available Interest Option shall be designated in accordance herewith.

(ii) Increased Cost of Borrowings. If the adoption after the Effective Date of any applicable Legal Requirement or any change after the Effective Date in any applicable Legal Requirement or in the interpretation or administration thereof by any Governmental Authority or compliance by any Lender with any request or directive (whether or not having the force of law) issued after the Effective Date by any central bank or Governmental Authority shall at any time as a result of any portion of the principal balances of the Notes being maintained on the basis of a Eurodollar Rate:

> (1) subject any Lender to any Taxes, or any deduction or withholding for any Taxes, on or from any payment due under any LIBOR Borrowing or other amount due hereunder, other than income and franchise taxes of the United States or its political subdivisions or such other jurisdiction in which the applicable Lender has its principal office or applicable lending office; or

- (2) change the basis of taxation of payments due from any Borrower to any Lender under any LIBOR Borrowing (otherwise than by a change in the rate of taxation of the overall net income of such Lender); or
- (3) impose, modify, increase or deem applicable any reserve requirement (excluding that portion of any reserve requirement included in the calculation of the applicable Eurodollar Rate), special deposit requirement or similar requirement (including, but not limited to, state law requirements and Regulation D) against assets of any Lender, or against deposits with any Lender, or against loans made by any Lender, or against any other funds, obligations or other Property owned or held by any Lender; or
- (4) impose on any Lender any other condition regarding any LIBOR Borrowing;

and the result of any of the foregoing is to increase the cost to any Lender of agreeing to make or of making, renewing or maintaining such LIBOR Borrowing, or reduce the amount of principal or interest received by any Lender, then, within 15 Business Days after demand by any Agent (accompanied by a statement setting forth in reasonable detail the applicable Lender's basis therefor), the applicable Borrower shall pay to the applicable Agent additional amounts which shall compensate each Lender for such increased cost or reduced amount. The determination by any Lender of the amount of any such increased cost, increased reserve requirement or reduced amount shall be conclusive and binding, absent manifest error. Each Borrower shall have the right, if it receives from any Agent any notice referred to in this paragraph, upon three Business Days' notice to the applicable Agent (which shall notify each affected Lender), either (i) to repay in full (but not in part) any borrowing with respect to which such notice was given, together with any accrued interest thereon, or (ii) to convert the LIBOR Borrowing which is the subject of the notice to a Base Rate Borrowing; provided, that any such repayment or conversion shall be accompanied by payment of (x) the amount required to compensate each Lender for the increased cost or reduced amount referred to in the preceding paragraph; (y) all accrued and unpaid interest to date on the amount so repaid or converted, and (z) any Funding Loss which any Lender may incur as a result of such repayment or conversion. Each Lender will notify the applicable Borrower through the applicable Agent of any event occurring after the date of this Agreement which will entitle such Lender to compensation pursuant to this Section as promptly as practicable after it obtains knowledge thereof and determines to request such compensation, and (if so requested by the applicable Borrower through the applicable Agent) will designate a different lending office of such Lender for the applicable LIBOR Borrowing or will take such other action as the applicable Borrower may reasonable request if such designation or action is consistent with the internal policy of such Lender and legal and regulatory restrictions, will avoid the need for, or reduce the amount of, such compensation and will not, in the sole opinion of such Lender, be disadvantageous to such Lender.

(iii) Inadequacy of Pricing and Rate Determination. If, for any reason with respect to any Interest Period, any Agent (or, in the case of clause 2 below, the applicable Lender) shall have

determined (which determination shall be conclusive and binding upon each Borrower, absent manifest error) that:

- (1) any Agent is unable through its customary general practices to determine any applicable Eurodollar Rate, or
- (2) any applicable Eurodollar Rate will not adequately and fairly reflect the cost to any Lender of making and maintaining such LIBOR Borrowing hereunder for any proposed Interest Period,

then the applicable Agent shall give the applicable Borrower notice thereof and thereupon, (A) any Rate Designation Notice previously given by such Borrower designating the applicable LIBOR Borrowing which has not commenced as of the date of such notice from any Agent shall be deemed for all purposes hereof to be of no force and effect, as if never given, and (B) until the applicable Agent shall notify such Borrower that the circumstances giving rise to such notice from such Agent no longer exist, each Rate Designation Notice requesting the applicable Eurodollar Rate shall be deemed a request for a Base Rate Borrowing, and any applicable LIBOR Borrowing then outstanding shall be converted, without any notice to or from any Borrower, upon the termination of the Interest Period then in effect with respect to it, to a Base Rate Borrowing.

(iv) Funding Losses. Each Borrower shall indemnify each applicable Lender against and hold each applicable Lender harmless from any Funding Loss relating to Loans to such Borrower or relating to Bankers' Acceptances requested by such Borrower. This indemnity shall survive the payment of the Notes. A certificate of any Lender (explaining in reasonable detail the amount and calculation of the amount claimed) as to any additional amounts payable pursuant to this paragraph submitted to the applicable Borrower shall be conclusive and binding upon such Borrower, absent manifest error.

(d) Funding Offices; Adjustments Automatic; Calculation Year. Any Lender may, if it so elects, fulfill its obligation as to any LIBOR Borrowing by causing a branch or affiliate of such Lender to make such Loan and may transfer and carry such Loan at, to or for the account of any branch office or affiliate of such Lender; provided, that in such event for the purposes of this Agreement such Loan shall be deemed to have been made by such Lender and the obligation of the applicable Borrower to repay such Loan shall nevertheless be to such Lender and shall be deemed held by it for the account of such branch or affiliate. Without notice to any Borrower or any other Person, each rate required to be calculated or determined under this Agreement shall automatically fluctuate upward and downward in accordance with the provisions of this Agreement. Interest at the Canadian Prime Rate or any applicable Prime Rate shall be computed on the basis of the actual number of days elapsed in a year consisting of 365 or 366 days, as the case may be. All other interest required to be calculated or determined under this Agreement shall be computed on the basis of the actual number of days elapsed in a year consisting of 360 days, unless the Ceiling Rate would thereby be exceeded, in which event, to the extent necessary to avoid exceeding the Ceiling Rate,

41

the applicable interest shall be computed on the basis of the actual number of days elapsed in the applicable calendar year in which such interest accrued.

(e) Funding Sources. Notwithstanding any provision of this Agreement to the contrary, each Lender shall be entitled to fund and maintain its funding of all or any part of the Loans in any manner it sees fit, it being understood, however, that for the purposes of this Agreement all determinations hereunder shall be made as if each Lender had actually funded and maintained each LIBOR Borrowing during each Interest Period through the purchase of deposits having a maturity corresponding to such Interest Period and bearing an interest rate equal to the Eurodollar Rate for such Interest Period.

4. Payments; Pro Rata Treatment; Computations, Etc.

4.1 Payments.

(a) Except to the extent otherwise provided herein, all payments of principal, interest, Reimbursement Obligations and other amounts to be made by any Borrower hereunder, under the Notes and under the other Loan Documents shall be made in (i) with respect to Bankers' Acceptance Liabilities and Canadian Prime Loans, Canadian Dollars and (ii) in all other cases, in Dollars, in immediately available funds, to the applicable Agent at its Principal Office (or in the case of a successor U.S. Agent, at the principal office of such successor U.S. Agent in the United States or in the case of a successor Canadian Agent, at the principal office of such successor Canadian Agent, at the principal office of such successor Canadian Agent in Canada), not later than 11:00 a.m. Houston, Texas time (in the case of any payment by the Canadian Borrower) on the date on which such payment shall become due (each such payment made after such time on such due date to be deemed to have been made on the next succeeding Business Day).

(b) Each Borrower shall, at the time of making each payment hereunder, under any Note or under any other Loan Document, specify to the applicable Agent Obligations payable by such Borrower hereunder or thereunder to which such payment is to be applied. Each payment received by any Agent hereunder, under any Note or under any other Loan Document for the account of a Lender shall be paid promptly to such Lender, in immediately available funds. If any Agent fails to send to any Lender the applicable amount by the close of business on the date any such payment is received by such Agent if such payment is received prior to 11:00 a.m. Houston, Texas time (in the case of any payment to a U.S. Lender) or 12:00 noon Toronto, Ontario time (in the case of any payment to a Canadian Lender) (or on the next succeeding Business Day with respect to payments which are received after such time), such Agent shall pay to the applicable Lender interest on such amount from such date at a rate of interest per annum equal to (i) in respect of Obligations which are denominated in Dollars, the Federal Funds Rate and (ii) in respect of Canadian Obligations which are denominated in Canadian Dollars, the CDOR Rate.

(c) If the due date of any payment hereunder or under any other Loan Document falls on a day which is not a Business Day, the due date for such payments (except as otherwise provided in

42

the definition of "Interest Period" set forth in Section 1.1 hereof) shall be extended to the next succeeding Business Day and interest shall be payable for any principal so extended for the period of such extension.

48

(d) All payments by any Borrower hereunder or under any other Loan Document shall be made free and clear of and without deduction for or on account of any present or future income, stamp, or other taxes, fees, duties, withholding or other charges of any nature whatsoever imposed by any taxing authority excluding in the case of each Lender taxes imposed on or measured by its net income or franchise taxes imposed by the jurisdiction in which it is organized or through which it acts for purposes of this Agreement (such nonexcluded items being hereinafter referred to as "Taxes"). If as a result of any change in law (or the interpretation thereof) after the date that the applicable Lender became a "Lender" under this Agreement any withholding or deduction from any payment to be made to, or for the account of, a Lender by any Borrower hereunder or under any other Loan Document is required in respect of any Taxes pursuant to any applicable law, rule, or regulation, then such Borrower will (i) pay to the relevant authority the full amount required to be so withheld or deducted; (ii) to the extent available, promptly forward to the applicable Agent an official receipt or other documentation reasonably satisfactory to such Agent evidencing such payment to such authority; and (iii) pay to the applicable Agent, for the account of each affected Lender, such additional amount or amounts as are necessary to ensure that the net amount actually received by such Lender will equal the full amount such Lender would have received had no such withholding or deduction been required. Each Lender shall determine such additional amount or amounts payable to it (which determination shall, in the absence of manifest error, be conclusive and binding on each Borrower). If a Lender becomes aware that any such withholding or deduction from any payment to be made by any Borrower hereunder or under any other Loan Document is required, then such Lender shall promptly notify the applicable Agent and the applicable Borrower thereof stating the reasons therefor and the additional amount required to be paid under this Section. Each Lender shall execute and deliver to the applicable Agent and the applicable Borrower such forms as it may be required to execute and deliver pursuant to Section 11.13 hereof. To the extent that any such withholding or deduction results from the failure of a Lender to provide a form required by Section 11.13 hereof (unless such failure is due to some prohibition under applicable Legal Requirements), the applicable Borrower shall have no obligation to pay the additional amount required by clause (iii) above. Anything in this Section notwithstanding, if any Lender elects to require payment by any Borrower of any material amount under this Section, the applicable Borrower may, within 60 days after the date of receiving notice thereof and so long as no Default shall have occurred and be continuing, elect to terminate such Lender as a party to this Agreement; provided that, concurrently with such termination the applicable Borrower shall (i) if the Agents and each of the other Lenders shall consent, pay that Lender all principal, interest and fees and other amounts owed to such Lender through such date of termination or (ii) have arranged for another financial institution approved by the Agents (such approval not to be unreasonably withheld) as of such date, to become a substitute Lender for all purposes under this Agreement in the manner provided in Section 11.6; provided further that, prior to substitution for any Lender, the applicable Borrower shall have given written notice to the Agents of such intention and the Lenders shall have

the option, but no obligation, for a period of 60 days after receipt of such notice, to increase their Commitments in order to replace the affected Lender in lieu of such substitution.

49

Pro Rata Treatment. Except to the extent otherwise provided 4.2 herein: (a) each borrowing from the Lenders under Section 2.1 hereof shall be made (x) in the case of Canadian Loans, ratably from the Canadian Lenders in accordance with their respective Canadian Commitments and (y) in the case of U.S. Loans, ratably from the U.S. Lenders in accordance with their respective U.S. Commitments; (b) each payment of commitment fees shall be made for the account of the Lenders, and each termination or reduction of the Commitments of the Lenders under Section 2.3 hereof shall be applied, pro rata, according to the Lenders' respective Commitments; (c) each payment by any Borrower of principal of or interest on the Loans or any Bankers' Acceptance shall be made to the applicable Agent for the account of the applicable Lenders pro rata in accordance with the respective unpaid principal amounts of the Loans held by or Bankers' Acceptances accepted by such Lenders, and (d) the applicable Lenders (other than the applicable Issuer) shall purchase from the applicable Issuer participations in each Letter of Credit to the extent of their respective U.S. Commitment Percentages or Canadian Commitment Percentages, as the case may be.

4.3 Certain Actions, Notices, Etc. Notices to the applicable Agent of any termination or reduction of Commitments and of borrowings and optional prepayments of Loans and requests for issuances of Letters of Credit shall be irrevocable and shall be effective only if received by the applicable Agent not later than 11:00 a.m. Houston, Texas time (in the case of U.S. Loans and U.S. Letters of Credit) or 12:00 noon Toronto, Ontario time (in the case of Canadian Loans and Canadian Letters of Credit)) on the number of Business Days prior to the date of the relevant termination, reduction, borrowing and/or prepayment specified below:

	Number of Business Days Prior Notice
Termination or Reduction of Commitments	5
Loan repayment	1
Base Rate Borrowings and Canadian Prime Loans	same day
Letter of Credit issuance	3
Selection of a Eurodollar Rate	3 LIBOR Business Days

Each such notice of termination or reduction shall specify the amount of the applicable Commitment to be terminated or reduced. Each such notice of borrowing or prepayment shall specify the amount of the Loans to be borrowed or prepaid and the date of borrowing or prepayment (which shall be a Business Day). The applicable Agent shall promptly notify the affected Lenders of the contents of each such notice.

Non-Receipt of Funds by Any Agent. Unless the applicable Agent 4.4 shall have been notified by a Lender or a Borrower (the "Payor") prior to the date on which such Lender is to make payment to such Agent of the proceeds of a Loan (or funding of a drawing under a Letter of Credit or reimbursement with respect to any drawing under a Letter of Credit or funding of a payment under a Bankers' Acceptance or reimbursement with respect to any payment under a Bankers' Acceptance) to be made by it hereunder or the applicable Borrower is to make a payment to such Agent for the account of one or more of the Lenders, as the case may be (such payment being herein called the "Required Payment"), which notice shall be effective upon receipt, that the Payor does not intend to make the Required Payment to such Agent, the applicable Agent may assume that the Required Payment has been made and may, in reliance upon such assumption (but shall not be required to), make the amount thereof available to the intended recipient on such date and, if the Payor has not in fact made the Required Payment to such Agent, the recipient of such payment (or, if such recipient is the beneficiary of a Letter of Credit, the applicable Borrower and, if such Borrower fails to pay the amount thereof to the applicable Agent forthwith upon demand, the applicable Lenders ratably in proportion to their respective Commitment Percentages) shall, on demand, pay to such Agent the amount made available by such Agent, together with interest thereon in respect of the period commencing on the date such amount was so made available by such Agent until the date Agent recovers such amount at a rate of interest per annum equal to (i) in respect of Obligations which are denominated in Dollars, the Federal Funds Rate and (ii) in respect of Canadian Obligations which are denominated in Canadian Dollars, the CDOR Rate.

4.5 Sharing of Payments, Etc. If a Lender shall obtain payment of any principal of or interest on any Loan made by it under this Agreement, on any Reimbursement Obligation or on any other Obligation then due to such Lender hereunder, through the exercise of any right of set-off (including, without limitation, any right of setoff or Lien granted under Section 9.2 hereof), banker's Lien, counterclaim or similar right, or otherwise, it shall promptly purchase from the other Lenders participations in the Loans made, or Reimbursement Obligations or other Obligations held, by the other Lenders in such amounts, and make such other adjustments from time to time as shall be equitable to the end that all the Lenders shall share the benefit of such payment (net of any expenses which may be incurred by such Lender in obtaining or preserving such benefit) pro rata in accordance with the unpaid Obligations then due to each of them. To such end all the Lenders shall make appropriate adjustments among themselves (by the resale of participations sold or otherwise) if such payment is rescinded or must otherwise be restored. Each Borrower agrees, to the fullest extent it may effectively do so under applicable law, that any Lender so purchasing a participation in the Loans made, or Reimbursement Obligations or other Obligations held, by other Lenders may exercise all rights of set-off, bankers' Lien, counterclaim or similar rights with respect to such participation as fully as if such Lender were a direct holder of Loans, Reimbursement Obligations

or other Obligations in the amount of such participation. Nothing contained herein shall require any Lender to exercise any such right or shall affect the right of any Lender to exercise, and retain the benefits of exercising, any such right with respect to any other indebtedness or obligation of any Borrower.

5. Conditions Precedent.

5.1 Initial Loans, Letters of Credit and Bankers' Acceptances. The obligation of each Lender or each Issuer to make its initial Loans or issue or participate in the initial Letter of Credit hereunder or to accept and purchase its initial Bankers' Acceptance hereunder (whichever shall first occur) is subject to the following conditions precedent, each of which shall have been fulfilled or waived to the satisfaction of Agents:

(a) Authorization and Status. Agents shall have received from the appropriate Governmental Authorities certified copies of the Organizational Documents (other than by-laws) of each Obligor, and evidence satisfactory to Agents of all action taken by each Obligor authorizing the execution, delivery and performance of the Loan Documents and all other documents related to this Agreement to which it is a party (including, without limitation, a certificate of the secretary of each such party which is a corporation setting forth the resolutions of its Board of Directors authorizing the transactions contemplated thereby and attaching a copy of its bylaws), together with such certificates as may be appropriate to demonstrate the qualification and good standing of and payment of taxes by each Obligor in the jurisdiction of its organization and in each other jurisdiction where the failure to qualify would have a Material Adverse Effect.

(b) Incumbency. Each Obligor shall have delivered to Agents a certificate in respect of the name and signature of each of the officers (i) who is authorized to sign on its behalf the applicable Loan Documents related to any Loan, Letter of Credit and Bankers' Acceptance and (ii) who will, until replaced by another officer or officers duly authorized for that purpose, act as its representative for the purposes of signing documents and giving notices and other communications in connection with any Loan, Letter of Credit and Bankers' Acceptance. Each Agent and each Lender may conclusively rely on such certificates until they receive notice in writing from the applicable Obligor to the contrary.

(c) Notes. Agents shall have received the appropriate Notes of Borrowers for each Lender, duly completed and executed.

(d) Loan Documents. Each Obligor shall have duly executed and delivered the Loan Documents to which it is a party (in such number of copies as Agents shall have requested). Each such Loan Document shall be in substantially the form furnished to the Lenders prior to their execution of this Agreement, together with such changes therein as Agents may approve.

(e) Fees and Expenses. Borrowers shall have paid to Agents all unpaid fees in the amounts previously agreed upon in writing between any Borrower and any Agent; and shall have in

46

addition paid to Agents all amounts payable under Section 11.3 hereof, on or before the date of this Agreement, except for amounts which the applicable Agent, in its sole discretion, agrees may be paid at a later date.

(f) Insurance. Borrowers shall have delivered to Agents evidence satisfactory to Agents regarding the existence of all insurance required to be maintained by each Obligor by this Agreement and the Security Documents.

(g) Security Matters. All such action as Agents shall have requested to perfect the Liens created pursuant to the Security Documents shall have been taken, including, without limitation, where applicable, the filing and recording of the Security Documents with the appropriate Governmental Authorities. Agent shall also have received evidence satisfactory to it that the Liens created by the Security Documents constitute first priority Liens, except for the exceptions expressly provided for herein or therein, including, without limitation, delivery of all applicable stock certificates (with stock powers executed in blank), Uniform Commercial Code search reports and other applicable personal property registry reports, satisfactory title evidence in form and substance acceptable to Agent, and executed releases of any prior Liens.

(h) Opinions of Counsel. Agents shall have received such opinions of counsel to Obligors as Agents shall reasonably request with respect to Obligors and the Loan Documents.

(i) Consents. Agents shall have received evidence satisfactory to the Lenders that all material consents of each Governmental Authority and of each other Person, if any, reasonably required in connection with (a) the Loans, Letters of Credit and Bankers' Acceptances and (b) the execution, delivery and performance of this Agreement and the other Loan Documents have been satisfactorily obtained.

(j) Payment of Certain Outstanding Indebtedness. Agent shall have received evidence satisfactory to Agent that all existing Indebtedness owing under the credit facility provided to U.S. Borrower and certain of its Affiliates by General Electric Credit Corporation and under the credit facility provided to Dreco Energy Service Ltd. by The Bank of Nova Scotia shall have been paid in full (or will be paid in full out of the initial advance hereunder) and that all rights to further advances under such facility shall have been terminated.

(k) Dreco Acquisition. Dreco Energy Services Ltd. shall have become a wholly-owned Subsidiary (direct or indirect) of Parent (exclusive of the Exchangeable Shares).

(1) Other Documents. Agents shall have received such other documents consistent with the terms of this Agreement and relating to the transactions contemplated hereby as Agents may reasonably request.

5.2 All Loans, Letters of Credit and Bankers' Acceptances. The obligation of each Lender to make any Loan to be made by it hereunder or to issue or participate in any Letter of Credit

47

or to accept and purchase any Bankers' Acceptance is subject to (a) the accuracy, in all material respects, on the date of such Loan or such issuance or such acceptance and purchase of all representations and warranties of each Obligor contained in this Agreement and the other Loan Documents; (b) the applicable Agent shall have received the following, all of which shall be duly executed and in Proper Form: (1) a Request for Extension of Credit as to the Loan, Letter of Credit or Bankers' Acceptance, as the case may be, by the time and on the Business Day specified under Section 4.3 hereof, (2) in the case of a Letter of Credit, an Application, and (3) such other documents as the applicable Agent may reasonably require; (c) prior to the making of such Loan or the issuance of such Letter of Credit or the acceptance and purchase of such Bankers' Acceptance, there shall have occurred no event which could reasonably be expected to have a Material Adverse Effect; (d) no Default or Event of Default shall have occurred and be continuing; (e) the making of such Loan or the issuance of such Letter of Credit or the acceptance and purchase of such Bankers' Acceptance shall not be illegal or prohibited by any Legal Requirement, and (f) Borrowers shall have paid all fees and expenses of the type described in Section 11.3 hereof and all other fees owed to any Agent or any Lender under the Loan Documents which are due and payable, in each case, prior to or on the date of such Loan or such issuance or such acceptance and purchase. The submission by any Borrower of a Request for Extension of Credit shall be deemed to be a representation and warranty that the conditions precedent to the applicable Loan or Letter of Credit or Bankers' Acceptance have been satisfied.

6. Representations and Warranties.

To induce the Lenders to enter into this Agreement and to make the Loans and issue or participate in the Letters of Credit and accept and purchase Bankers' Acceptances, U.S. Borrower and Canadian Borrower each represents and warrants (such representations and warranties to survive any investigation and the making of the Loans and the issuance of any Letters of Credit and the acceptance and purchase of any Bankers' Acceptances) to the Lenders and Agents as follows:

6.1 Organization. Each Obligor (a) is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization; (b) has all necessary power and authority to conduct its business as presently conducted, and (c) is duly qualified to do business and in good standing in the jurisdiction of its organization and in all jurisdictions in which the failure to so qualify would reasonably be expected to have a Material Adverse Effect.

6.2 Financial Statements. Borrowers have furnished to Agents: (i) audited financial statements (including a balance sheet) as to Parent which fairly present in all material respects, in accordance with GAAP, the consolidated financial condition and the results of operations of Parent as at the end of the fiscal year ended December 31, 1996, (ii) unaudited consolidating financial statements (including a balance sheet) as to Parent and as to each of the Persons comprising "Canadian Borrower" which fairly present in all material respects, in accordance with GAAP, the financial condition and the results of operations of the applicable Person, on a consolidating basis, as at the end of the fiscal year ended August 31, 1996 (as to Dreco Energy Services Ltd. and its Subsidiaries) or December 31, 1996 (as to all other applicable Persons), (iii) unaudited financial

48

statements (including a balance sheet) as to each of the Persons comprising "Canadian Borrower" which fairly present in all material respects, in accordance with GAAP, the consolidated financial condition and the results of operations of the applicable Person as at the end of the fiscal year ended August 31, 1996 (as to Dreco Energy Services Ltd. and its Subsidiaries) or December 31, 1996 (as to all other applicable Persons); (iv) unaudited financial statements (including a balance sheet) as to Parent and as to each of the Persons comprising "Canadian Borrower" which fairly present in all material respects, in accordance with GAAP (subject to year-end adjustments and the absence of notes), the consolidated financial condition and the results of operations of the applicable Person as at the end of the fiscal quarter ended May 31, 1997 (as to Dreco Energy Services Ltd. and its Subsidiaries) or June 30, 1997 (as to all other applicable Persons) and (v) unaudited consolidating financial statements (including a balance sheet) as to Parent and as to each of the Persons comprising "Canadian Borrower" which fairly present in all material respects, in accordance with GAAP, the financial condition and the results of operations of the applicable Person as at the end of the fiscal quarter ended May 31, 1997 (as to Dreco Energy Services Ltd. and its Subsidiaries) or June 30, 1997 (as to all other applicable Persons). No events, conditions or circumstances have occurred from the date that such financial statements were delivered to Agents through the Effective Date which would cause such financial statements to be misleading in any material respect. There are no material instruments or liabilities which should be reflected in such financial statements which are not so reflected. For purposes of this Section 6.2 and Section 7.2 hereof, financial statements as to each Person comprising the "Canadian Borrower" shall be prepared in accordance with generally accepted accounting principles in effect in Canada, and "GAAP" shall be deemed modified accordingly.

54

Enforceable Obligations; Authorization. The Loan Documents are 6.3 legal, valid and binding obligations of each applicable Obligor, enforceable in accordance with their respective terms, except as may be limited by bankruptcy, insolvency and other similar laws and judicial decisions affecting creditors' rights generally and by general equitable principles. The execution, delivery and performance of the Loan Documents (a) have all been duly authorized by all necessary action; (b) are within the power and authority of each applicable Obligor; (c) do not and will not contravene or violate any Legal Requirement applicable to any applicable Obligor or the Organizational Documents of any applicable Obligor, the contravention or violation of which would reasonably be expected to have a Material Adverse Effect; (d) do not and will not result in the breach of, or constitute a default under, any material agreement or instrument by which any Obligor or any of its Property may be bound, and (e) do not and will not result in the creation of any Lien upon any Property of any Obligor, except in favor of U.S. Agent as expressly contemplated therein. All necessary permits, registrations and consents for such making and performance have been obtained. Except as otherwise expressly stated in the Security Documents, the Liens of the Security Documents will constitute valid and perfected first priority Liens on the Property described therein, subject to no other Liens whatsoever except Permitted Liens.

6.4 Other Debt. No Obligor is in default in the payment of any other Borrowed Money Indebtedness or under any agreement, mortgage, deed of trust, security agreement or lease to which it is a party and which default would reasonably be expected to have a Material Adverse Effect.

6.5 Litigation. There is no litigation or administrative proceeding, to the knowledge of any executive officer of any Obligor, pending or threatened against, nor any outstanding judgment, order or decree against, any Obligor before or by any Governmental Authority which does or would reasonably be expected to have a Material Adverse Effect. No Obligor is in default with respect to any judgment, order or decree of any Governmental Authority where such default would have a Material Adverse Effect.

6.6 Title. Each Obligor has good and marketable title to the Collateral, if any, pledged (or purported to be pledged) by such Obligor pursuant to the Security Documents, free and clear of all Liens (other than Permitted Liens upon proceeds of Collateral, but not the original Collateral).

6.7 Taxes. Each Obligor has filed all material tax returns required to have been filed and paid all taxes shown thereon to be due, except those for which extensions have been obtained and those which are being contested in good faith.

6.8 Regulations G, U and X. None of the proceeds of any Loan or proceeds from the acceptance and purchase of Bankers' Acceptances will be used for the purpose of purchasing or carrying directly or indirectly any margin stock or for any other purpose that would constitute this transaction a "purpose credit" within the meaning of Regulations G, U and X of the Board of Governors of the Federal Reserve System, as any of them may be amended from time to time.

6.9 Subsidiaries. As of the Effective Date, Parent has no Subsidiaries other than as set forth on Exhibit I hereto. The percentage of the issued and outstanding equity interests in each applicable Subsidiary which is owned by Parent or one or more of its Subsidiaries is set forth on Exhibit I hereto.

6.10 No Untrue or Misleading Statements. No representation or warranty made by any Obligor in any Loan Document or in any document, instrument or other writing furnished to the Lenders by or on behalf of any Obligor in connection with the transactions contemplated in any Loan Document does or will contain any untrue material statement of fact or will omit to state any such fact (of which any executive officer of any Obligor has knowledge) necessary to make the representations, warranties and other statements contained herein or in such other document, instrument or writing not misleading in any material respect.

6.11 ERISA. With respect to each Plan, each Borrower and each member of the Controlled Group have fulfilled their obligations, including obligations under the minimum funding standards of ERISA and the Code and are in compliance in all material respects with the provisions of ERISA and the Code. No event has occurred which could result in a liability of any Borrower or any member of the Controlled Group to the PBGC or a Plan (other than to make contributions in the ordinary course) that would reasonably be expected to have a Material Adverse Effect. There have not been any nor are there now existing any events or conditions that would cause any Lien provided under ERISA to attach to any Property of any Borrower or any member of the Controlled Group.

50

Unfunded Liabilities as of the date hereof do not exceed \$1,000,000. No "prohibited transaction" has occurred with respect to any Plan.

56

6.12 Investment Company Act. No Obligor is an investment company within the meaning of the Investment Company Act of 1940, as amended, or, directly or indirectly, controlled by or acting on behalf of any Person which is an investment company, within the meaning of said Act.

6.13 Public Utility Holding Company Act. No Obligor is an "affiliate" or a "subsidiary company" of a "public utility company," or a "holding company," or an "affiliate" or a "subsidiary company" of a "holding company," as such terms are defined in the Public Utility Holding Company Act of 1935, as amended.

6.14 Solvency. None of U.S. Borrower, Canadian Borrower, any other Obligor, U.S. Borrower and its Subsidiaries, on a consolidated basis, or Canadian Borrower and its Subsidiaries, on a consolidated basis, is "insolvent," as such term is used and defined in (i) the Bankruptcy Code and (ii) the fraudulent conveyance statutes of the State of Texas or of any jurisdiction in which any of the Collateral may be located.

6.15 Fiscal Year. The fiscal year of each Obligor (other than Dreco Energy Services Ltd. and its Subsidiaries) ends on December 31. The fiscal year of Dreco Energy Services Ltd. currently ends on August 31, but by December 31, 1997 the fiscal year end of Dreco Energy Services Ltd. and its Subsidiaries will be changed to December 31.

6.16 Compliance. Each Obligor is in compliance with all Legal Requirements applicable to it, except to the extent that the failure to comply therewith would not reasonably be expected to have a Material Adverse Effect.

Environmental Matters. Each Obligor has, to the best knowledge 6.17 of their respective executive officers, obtained and maintained in effect all Environmental Permits (or the applicable Person has initiated the necessary steps to transfer the Environmental Permits into its name or obtain such permits), the failure to obtain which would reasonably be expected to have a Material Adverse Effect. Each Obligor and its Properties, business and operations have been and are, to the best knowledge of their respective executive officers, in compliance with all applicable Requirements of Environmental Law and Environmental Permits the failure to comply with which would reasonably be expected to have a Material Adverse Effect. Each Obligor and its Properties, business and operations are not subject to any (A) Environmental Claims or (B), to the best knowledge of their respective executive officers (after making reasonable inquiry of the personnel and records of their respective Corporations), Environmental Liabilities, in either case direct or contingent, arising from or based upon any act, omission, event, condition or circumstance occurring or existing on or prior to the date hereof which would reasonably be expected to have a Material Adverse Effect. None of the officers of any Obligor has received any notice of any violation or alleged violation of any Requirements of Environmental Law or Environmental Permit or any Environmental Claim in connection with its Properties, liabilities, condition (financial or otherwise), business or operations

which would reasonably be expected to have a Material Adverse Effect. Neither U.S. Borrower nor Canadian Borrower knows of any event or condition with respect to currently enacted Requirements of Environmental Laws presently scheduled to become effective in the future with respect to any of the Properties of any Obligor which would reasonably be expected to have a Material Adverse Effect, for which the applicable Obligor has not made good faith provisions in its business plan and projections of financial performance.

6.18 Collateral Covered. As of the Effective Date, the Collateral covered by the Security Documents constitutes all of the issued and outstanding equity interests in all of the Subsidiaries of Parent.

7. Affirmative Covenants.

U.S. Borrower and Canadian Borrower each covenants and agrees with Agents and the Lenders that prior to the termination of this Agreement it will do or cause to be done, and cause each other Obligor (unless limited by the language of the applicable provision to less than all of the Obligors) to do or cause to be done, each and all of the following:

7.1 Taxes, Existence, Regulations, Property, Etc. At all times (a) pay when due all material taxes and governmental charges of every kind upon it or against its income, profits or Property, unless and only to the extent that the same shall be contested diligently in good faith and adequate reserves in accordance with GAAP have been established therefor; (b) do all things necessary to preserve its existence, qualifications, rights and franchises in all jurisdictions where such failure to qualify would reasonably be expected to have a Material Adverse Effect; (c) comply with all applicable Legal Requirements (including without limitation Requirements of Environmental Law) in respect of the conduct of its business and the ownership of its Property, the noncompliance with which would reasonably be expected to have a Material Adverse Effect; and (d) cause its Property to be protected, maintained and kept in good repair and make all replacements and additions to such Property as may be reasonably necessary to conduct its business properly and efficiently.

Financial Statements and Information. Furnish to Agents and each 7.2 Lender each of the following: (a) as soon as available and in any event within 90 days after the end of each applicable fiscal year, beginning with the fiscal year ending on December 31, 1997, audited Annual Financial Statements of Parent and unaudited Annual Financial Statements of each Person comprising "Canadian Borrower", (b) as soon as available and in any event within 45 days after the end of each fiscal quarter of each applicable fiscal year, Quarterly Financial Statements of Parent and of each Person comprising "Canadian Borrower"; (c) concurrently with the financial statements provided for in Subsections 7.2(a) and (b) hereof, such schedules, computations and other information, in reasonable detail, as may be required by Agents to demonstrate compliance with the covenants set forth herein or reflecting any non-compliance therewith as of the applicable date, all certified and signed by an authorized officer of U.S. Borrower as true and correct in all material respects to the best knowledge of such officer and, commencing with the quarterly financial

statement prepared as of September 30, 1997, a compliance certificate ("Compliance Certificate") in the form of Exhibit F hereto, duly executed by such authorized officer; (d) by December 31 of each fiscal year, U.S. Borrower's and Canadian Borrower's annual business plans for the next fiscal year (including their balance sheets and income and cash flow projections for such fiscal year) (Agents and Lenders acknowledge that such business plans shall be subject to review and approval by the Board of Directors of the applicable Person); (e) promptly upon their becoming publicly available, each financial statement, report, notice or definitive proxy statements sent by any Obligor to shareholders generally and each regular or periodic report and each registration statement, prospectus or written communication (other than transmittal letters) in respect thereof filed by any Obligor with, or received by any Obligor in connection therewith from, any securities exchange or the Securities and Exchange Commission or any successor agency, and (f) such other information relating to the condition (financial or otherwise), operations, prospects or business of any Obligor as from time to time may be reasonably requested by any Agent. Each delivery of a financial statement pursuant to this Section 7.2 shall constitute a restatement of the representations contained in the last two sentences of Section 6.2.

7.3 Financial Tests. Have and maintain:

(a) Tangible Net Worth - Tangible Net Worth for Parent and its Subsidiaries (on a consolidated basis) of not less than (1) at the end of each fiscal quarter ending after the date hereof through and including December 31, 1997, an amount equal to 85% of Tangible Net Worth of Parent and its Subsidiaries, on a consolidated basis, as of June 30, 1997 and (2) at the end of each fiscal quarter ending thereafter, the amount referred to in clause (1) above plus 50% of the Net Income (if positive) of Parent and its Subsidiaries, on a consolidated basis (if positive), for the period from the Effective Date through the last day of the fiscal quarter ending immediately prior to the date of such calculation plus 100% of the net proceeds realized from the issuance of any equity securities by Parent or its Subsidiaries during that period.

(b) Debt to Capitalization Ratio - a Debt to Capitalization Ratio of not greater than 40% at the end of each fiscal quarter ending after the date hereof.

(c) Interest Coverage Ratio - an Interest Coverage Ratio of not less than 3.00 to 1.00 at all times.

7.4 Inspection. Permit each Agent and each Lender upon 3 days' prior notice (unless a Default or an Event of Default has occurred which is continuing, in which case no prior notice is required) to inspect its Property, to examine its files, books and records, except privileged communication with legal counsel and classified governmental material, and make and take away copies thereof, and to discuss its affairs with its officers and accountants, all during normal business hours and at such intervals and to such extent as any Agent may reasonably desire.

53

7.5 Further Assurances. Promptly execute and deliver, at the expense of U.S. Borrower or Canadian Borrower, as the case may be, any and all other and further instruments which may be reasonably requested by any Agent to cure any defect in the execution and delivery of any Loan Document in order to effectuate the transactions contemplated by the Loan Documents, and in order to grant, preserve, protect and perfect the validity and priority of the security interests created by the Security Documents.

7.6 Books and Records. Maintain books of record and account which permit financial statements to be prepared in accordance with GAAP.

7.7 Insurance. U.S. Borrower and Canadian Borrower will (and will cause each other Obligor to) maintain insurance (including business interruption insurance) with such insurers, on such of its Property, with responsible companies in such amounts, with such deductibles and against such risks as are usually carried by owners of similar businesses and properties in the same general areas in which the applicable Obligor operates or as any Agent may otherwise reasonably require, and furnish each Agent satisfactory evidence thereof promptly upon request. These insurance provisions are cumulative of the insurance provisions of the Security Documents. Each Agent shall be provided with a certificate of the insurer that the insurance required by this Section may not be canceled, reduced or affected in any material manner without thirty (30) days' prior written notice to Agents.

7.8 Notice of Certain Matters. Give Agents written notice of the following promptly after any executive officer of U.S. Borrower or Canadian Borrower shall become aware of the same:

(a) the issuance by any court or governmental agency or authority of any injunction, order or other restraint prohibiting, or having the effect of prohibiting, the performance of this Agreement, any other Loan Document, or the making of the Loans or the acceptance and purchase of Bankers' Acceptances or the initiation of any litigation, or any claim or controversy which would reasonably be expected to result in the initiation of any litigation, seeking any such injunction, order or other restraint;

(b) the filing or commencement of any action, suit or proceeding, whether at law or in equity or by or before any court or any Governmental Authority involving claims with a specified amount in excess of \$3,000,000 against any Obligor or which may reasonably be expected to result in a Default hereunder; and

(c) any Event of Default or Default, specifying the nature and extent thereof and the action (if any) which is proposed to be taken with the respect thereto.

Borrowers will also notify Agents in writing at least 30 days prior to the date that any Obligor changes its name or the location of its chief executive office or principal place of business or the place where it keeps its books and records.

54

Capital Adequacy. If any Lender shall have determined that the 7.9 adoption after the Effective Date or effectiveness after the Effective Date (whether or not previously announced) of any applicable law, rule, regulation or treaty regarding capital adequacy, or any change therein after the Effective Date, or any change in the interpretation or administration thereof after the Effective Date by any Governmental Authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any Lender with any request or directive after the Effective Date regarding capital adequacy (whether or not having the force of law) of any such Governmental Authority, central bank or comparable agency has or would have the effect of reducing the rate of return on such Lender's capital as a consequence of its obligations hereunder, under the Letters of Credit, the Notes or other Obligations held by it to a level below that which such Lender could have achieved but for such adoption, change or compliance (taking into consideration such Lender's policies with respect to capital adequacy) by an amount deemed by such Lender to be material, then from time to time, upon satisfaction of the conditions precedent set forth in this Section, after demand by such Lender (with a copy to the appropriate Agent) as provided below, pay (subject to Section 11.7 hereof) to such Lender such additional amount or amounts as will compensate such Lender for such reduction. The certificate of any Lender setting forth such amount or amounts as shall be necessary to compensate it and the basis thereof and reasons therefor shall be delivered as soon as practicable to U.S. Borrower or Canadian Borrower, as the case may be, and shall be conclusive and binding, absent manifest error. U.S. Borrower or Canadian Borrower, as the case may be, shall pay the amount shown as due on any such certificate within fifteen (15) Business Days after the delivery of such certificate. In preparing such certificate, a Lender may employ such assumptions and allocations of costs and expenses as it shall in good faith deem reasonable and may use any reasonable averaging and attribution method.

ERISA Information and Compliance. Promptly furnish to Agents: 7.10 (i) immediately upon receipt, a copy of any notice of complete or partial withdrawal liability under ERISA and any notice from the PBGC under ERISA of an intent to terminate or appoint a trustee to administer any Plan, (ii) if requested by any Agent, promptly after the filing thereof with the United States Secretary of Labor or the PBGC or the Internal Revenue Service or any Governmental Authority having jurisdiction under Applicable Canadian Pension Legislation, copies of each annual and other report with respect to each Plan or any trust created thereunder, (iii) immediately upon becoming aware of the occurrence of any "reportable event," as such term is used under ERISA for which the disclosure requirements have not been waived, or of any "prohibited transaction," as such term is defined in Section 4975 of the Code, in connection with any Plan or any trust created thereunder, a written notice signed by the President or the principal financial officer of the applicable Borrower or the applicable member of the Controlled Group specifying the nature thereof, what action the applicable Borrower or the applicable member of the Controlled Group is taking or proposes to take with respect thereto, and, when known, any action taken by the PBGC, the Internal Revenue Service, the Department of Labor or any other applicable Governmental Authority with respect thereto, (iv) promptly after the filing or receiving thereof by any Borrower or any member of the Controlled Group of any notice of the institution of any proceedings or other actions which may result in the termination of any Plan, and (v) each request for waiver of the funding standards or extension of the amortization periods required by ERISA or Section 412 of the Code promptly after the request is

55

submitted by Borrower or any member of the Controlled Group to the Secretary of the Treasury, the Department of Labor, the Internal Revenue Service or any other applicable Governmental Authority. To the extent required under applicable statutory funding requirements, each Borrower will fund, or will cause the applicable member of the Controlled Group to fund, all current service pension liabilities as they are incurred under the provisions of all Plans from time to time in effect, and comply with all applicable provisions of ERISA, in each case, except to the extent that failure to do the same would not reasonably be expected to have a Material Adverse Effect. Each Borrower covenants that it shall and shall cause each member of the Controlled Group to (1) make contributions to each $\ensuremath{\mathsf{Plan}}$ in a timely manner and in an amount sufficient to comply with the contribution obligations under such Plan and the minimum funding standards requirements of ERISA; (2) prepare and file in a timely manner all notices and reports required under the terms of ERISA including but not limited to annual reports; and (3) pay in a timely manner all required PBGC premiums, in each case, except to the extent that failure to do the same would not reasonably be expected to have a Material Adverse Effect.

8. Negative Covenants.

U.S. Borrower and Canadian Borrower each covenants and agrees with Agents and the Lenders that prior to the termination of this Agreement it will not, and will not suffer or permit any other Obligor to, do any of the following:

8.1 Borrowed Money Indebtedness. Create, incur, suffer or permit to exist, or assume or guarantee, directly or indirectly, or become or remain liable with respect to any Borrowed Money Indebtedness, whether direct, indirect, absolute, contingent or otherwise, except the following: (a) indebtedness under this Agreement and the other Loan Documents and indebtedness secured by Liens permitted by Section 8.2 hereof; (b) Borrowed Money Indebtedness existing on the date of this Agreement and listed on Exhibit J attached hereto, and subject to Section 8.10 hereof, all renewals, extensions and replacements (but not increases other than increases in the Subordinated Indebtedness described on Exhibit J hereto which is owed to Oilwell, Inc. and National Supply Company, Inc. by reason of the addition of accrued interest in accordance with the present terms of the documentation evidencing such Subordinated Indebtedness) of any of the foregoing; (c) Interest Rate Risk Indebtedness; (d) purchase money Borrowed Money Indebtedness (or Borrowed Money Indebtedness under capital leases) to acquire (or lease) Equipment in the ordinary course of business not exceeding, in the aggregate for all such Borrowed Money Indebtedness for Parent and its Subsidiaries (on a consolidated basis), \$3,000,000 outstanding at any one time, and (e) other Borrowed Money Indebtedness not exceeding, in the aggregate for Parent and each of its Subsidiaries, at any one time outstanding, an amount equal to 12-1/2% of the Tangible Net Worth of Parent and its Subsidiaries (on a consolidated basis) as of the last day of the immediately preceding fiscal quarter. Without limiting the foregoing, Borrowers agree that all Borrowed Money Indebtedness owing by Parent to any of its Subsidiaries or owing by any Subsidiary of Parent to Parent or to any other Subsidiary of Parent shall be subordinated in a manner acceptable to the Majority Lenders and pursuant to documentation in Proper Form.

56

8.2 Liens. Create or suffer to exist any Lien upon any of its Property now owned or hereafter acquired, or acquire any Property upon any conditional sale or other title retention device or arrangement or any purchase money security agreement; or in any manner directly or indirectly sell, assign, pledge or otherwise transfer any of its Accounts or General Intangibles; provided, however, that any Obligor may create or suffer to exist Permitted Liens.

8.3 Contingent Liabilities. Directly or indirectly guarantee the performance or payment of, or purchase or agree to purchase, or assume or contingently agree to become or be secondarily liable in respect of, any obligation or liability of any other Person except for (a) the endorsement of checks or other negotiable instruments in the ordinary course of business; (b) obligations disclosed to Agents in the financial statements delivered on or prior to the Effective Date pursuant to Section 6.2 hereof (but not increases of such obligations after the Effective Date); (c) guaranties of performance (but not of any Borrowed Money Indebtedness) by a Subsidiary provided in the ordinary course of business, and (d) those liabilities permitted under Section 8.1 hereof.

8.4 Mergers, Consolidations and Dispositions of Assets. In any single transaction or series of transactions, directly or indirectly: (a) liquidate or dissolve; (b) be a party to any merger or consolidation unless and so long as (i) no Default or Event of Default has occurred that is then continuing, (ii) immediately thereafter and giving effect thereto, no event will occur and be continuing which constitutes a Default, (iii) an Obligor is the surviving Person; (iv) the surviving Person ratifies and assumes each Loan Document to which any party to such merger was a party, and (v) Agents are given at least 30 days' prior written notice of such merger or consolidation; (c) sell, convey or lease all or any part of its assets, except for sales and leases of Inventory in the ordinary course of business and replacement of other Property in the ordinary course of business, or (d) except for Liens in favor of Agents as contemplated herein, pledge, transfer or otherwise dispose of any equity interest in (or voting rights in respect of) any Subsidiary of Parent or any indebtedness of any Subsidiary of Parent or issue or permit any Subsidiary of Parent to issue any additional equity interest.

8.5 Redemption, Dividends and Distributions. At any time: (a) redeem, retire or otherwise acquire, directly or indirectly, any equity interest in any Obligor or (b) make any distributions of any Property or cash to the owner of any of the equity interests in any Obligor other than Permitted Dividends. Notwithstanding the foregoing, the holders of the Exchangeable Shares may exchange the Exchangeable Shares for shares of National-Oilwell, Inc. in accordance with the terms of the Combination Agreement described in the definition of "Exchangeable Shares" in Article I hereof.

8.6 Nature of Business. Change the nature of its business or enter into any business which is substantially different from the business in which it is presently engaged.

8.7 Transactions with Related Parties. Enter into any transaction or agreement with any officer, director or holder of any equity interest in any Obligor (or any Affiliate of any such Person) unless the same is upon terms substantially similar to those obtainable from wholly unrelated sources

57

(to the best knowledge of the executive officers of the applicable Obligor or Affiliate, after making reasonable inquiry of the personnel and records of the applicable Obligor or Affiliate).

8.8 Loans and Investments. Make any loan, advance, extension of credit or capital contribution to, or make or, except as permitted by Sections 8.4 or 8.9 hereof, have any Investment in, any Person, or make any commitment to make any such extension of credit or Investment, except (a) Permitted Investments and (b) normal and reasonable advances in the ordinary course of business to officers and employees.

Subsidiaries. Form, create or acquire any Subsidiary, except 8.9 that Parent (or any of its Subsidiaries) may form, create or acquire a whollyowned Subsidiary so long as (a) immediately thereafter and giving effect thereto, no event will occur and be continuing which constitutes a Default; (i) if such Subsidiary is a Material Subsidiary but is not a Foreign Subsidiary, such Subsidiary shall execute and deliver to each Agent a Guaranty in substantially the same form as the Guaranties executed concurrently herewith; (ii) if such Subsidiary is a Material Subsidiary but is not Foreign Subsidiary, the applicable owner(s) of the equity interests in such Subsidiary shall execute and deliver to U.S. Agent such Security Documents as U.S. Agent may reasonably require in order to create a valid, perfected, first priority Lien upon all of the issued and outstanding equity interests in such Subsidiary, (iii) if such Subsidiary is a Material Subsidiary and is Foreign Subsidiary and is not a wholly-owned direct Subsidiary of another Foreign Subsidiary, the applicable owner(s) of the equity interests in such Subsidiary shall execute and deliver to U.S. Agent such Security Documents as U.S. Agent may reasonably require in order to create a valid, perfected, first priority Lien upon 65% of the issued and outstanding equity interests in such Subsidiary and shall execute and deliver to Canadian Agent such Security Documents as Canadian Agent may reasonably require in order to create a valid, perfected, first priority Lien upon the remaining 35% of the issued and outstanding equity interests in such Subsidiary and (iv) no Foreign Subsidiary may form, create or acquire a Subsidiary which is not a Foreign Subsidiary, and (b) Agents are given at least 30 days' prior written notice of such formation, creation or acquisition.

8.10 Key Agreements. Terminate or agree to the termination of any Key Agreement or amend, modify or obtain or grant a waiver of any provision of any of the Key Agreements if such action would reasonably be expected to have a Material Adverse Effect.

8.11 Organizational Documents. Amend, modify, restate or supplement any of its Organizational Documents if such action would reasonably be expected to materially and adversely affect any Collateral or Obligation or the ability of any Obligor to perform its Obligations under any Loan Document, unless such action shall be consented to in writing by Agents.

8.12 Unfunded Liabilities. Incur any Unfunded Liabilities after the Effective Date or allow any Unfunded Liabilities in excess of \$1,000,000, in the aggregate, to arise or exist.

8.13 Subordinated Indebtedness. Except as expressly permitted in writing by the Majority Lenders, Borrowers will not amend, modify or obtain or grant a waiver of any provision of any

58

document or instrument evidencing any Subordinated Indebtedness or purchase, redeem, retire or otherwise acquire for value, deposit any monies with any Person with respect to or make any payment or prepayment of the principal of or any other amount owing in respect of, any Subordinated Indebtedness. Notwithstanding the foregoing, the Subordinated Indebtedness listed on Exhibit J hereto payable to Oilwell, Inc. and National Supply Company, Inc. may be prepaid at any time, without any necessity of notice to or consent by any Agent or any Lender.

8.14 Acquisitions. Acquire any real Property or any material personal Property after the Effective Date with respect to which the aggregate consideration for a single transaction would exceed \$50,000,000.

8.15 Prohibitions of Liens or Subsidiary Dividends. Enter into any agreement or contract which limits or restricts in any way (i) the granting of Liens by any Obligor securing any of the Obligations or (ii) any dividends or distributions by any Subsidiary of any Borrower to such Borrower or to another Subsidiary of such Borrower.

9. Defaults.

64

9.1 Events of Default. If any one or more of the following events (herein called "Events of Default") shall occur, then either Agent may (and at the direction of the Majority Lenders, shall) do any or all of the following: (1) without notice to U.S. Borrower, Canadian Borrower or any other Person, declare the Commitments terminated (whereupon the Commitments shall be terminated) and/or accelerate the Termination Date to a date as early as the date of termination of the Commitments; (2) terminate any Letter of Credit allowing for such termination, by sending a notice of termination as provided therein and require the applicable Borrower to provide Cover for outstanding Letters of Credit; (3) declare the principal amount then outstanding of and the unpaid accrued interest on the Loans and Reimbursement Obligations and all fees and all other amounts payable hereunder, under the Notes and under the other Loan Documents to be forthwith due and payable, whereupon such amounts shall be and become immediately due and payable, without notice (including, without limitation, notice of acceleration and notice of intent to accelerate), presentment, demand, protest or other formalities of any kind, all of which are hereby expressly waived by U.S. Borrower and Canadian Borrower; provided that in the case of the occurrence of an Event of Default with respect to any Obligor referred to in clause (f), (g) or (h) of this Section 9.1, the Commitments shall be automatically terminated and the principal amount then outstanding of and unpaid accrued interest on the Loans and the Reimbursement Obligations and all fees and all other amounts payable hereunder, under the Notes and under the other Loan Documents shall be and become automatically and immediately due and payable, without notice (including, without limitation, notice of acceleration and notice of intent to accelerate), presentment, demand, protest or other formalities of any kind, all of which are hereby expressly waived by U.S. Borrower and Canadian Borrower, and (4) exercise any or all other rights and remedies available to any Agent or any Lenders under the Loan Documents, at law or in equity:

(a) Payments - (i) any Obligor shall fail to make any payment or required prepayment of any installment of principal on the Loans payable under this Agreement or the other Loan Documents when due or (ii) any Obligor fails to make any payment or required prepayment of any Reimbursement Obligation or any interest with respect to the Loans or any other fee or amount under this Agreement or the other Loan Documents when due and, in the case of clause (ii) only, such failure to pay continues unremedied for a period of five days; or

(b) Other Obligations - any Obligor shall default in the payment when due of any principal of or interest on any Borrowed Money Indebtedness having an outstanding principal amount of at least \$3,000,000 (other than the Loans and Reimbursement Obligations) and such default shall continue beyond any applicable period of grace; or any event or condition shall occur which results in the acceleration of the maturity of any such Borrowed Money Indebtedness or enables (or, with the giving of notice or lapse of time or both, would enable) the holder of any such Borrowed Money Indebtedness or any Person acting on such holder's behalf to accelerate the maturity thereof and such event or condition shall not be cured within any applicable period of grace; or

(c) Representations and Warranties - any representation or warranty made or deemed made by or on behalf of any Obligor in this Agreement or any other Loan Document or in any certificate furnished or made by any Obligor to Agents or the Lenders in connection herewith or therewith shall prove to have been incorrect, false or misleading in any material respect as of the date thereof or as of the date as of which the facts therein set forth were stated or certified or deemed stated or certified; or

(d) Affirmative Covenants - (i) default shall be made in the due observance or performance of any of the covenants or agreements contained in Section 7.3 hereof, (ii) default shall be made in the due observance or performance of any of the covenants or agreements contained in Sections 7.2, 7.4, 7.7 or 7.8 hereof and, in each case, such default continues unremedied for a period of 20 days after notice thereof is given by any Agent to U.S. Borrower or to Canadian Borrower, or (iii) default is made in the due observance or performance of any of the other covenants and agreements contained in Section 7 hereof or any other affirmative covenant of any Obligor contained in this Agreement or any other Loan Document and such default continues unremedied for a period of 30 days after notice thereof is given by any Agent to U.S. Borrower or to Canadian Borrower; or

(e) Negative Covenants - default is made in the due observance or performance by U.S. Borrower or Canadian Borrower of any of the covenants or agreements contained in Section 8 of this Agreement or of any other negative covenant of any Obligor contained in this Agreement or any other Loan Document; or

(f) Involuntary Bankruptcy or Receivership Proceedings - a receiver, receiver-manager, interim receiver, monitor, conservator, liquidator or trustee of any Obligor or of any

60

of its Property is appointed by the order or decree of any court or agency or supervisory authority having jurisdiction; or any Obligor is adjudicated bankrupt or insolvent; or any of such Person's Property is sequestered by court order and such order remains in effect for more than 60 days; or a petition is filed against any Obligor under any state or federal bankruptcy, reorganization, arrangement, insolvency, readjustment or debt, dissolution, liquidation or receivership law or any jurisdiction, whether now or hereafter in effect, and is not dismissed within 60 days after such filing; or

(g) Voluntary Petitions or Consents - any Obligor commences a voluntary case or other proceeding or order seeking liquidation, reorganization, arrangement, insolvency, readjustment of debt, dissolution, liquidation or other relief with respect to itself or its debts or other liabilities under any bankruptcy, insolvency or other similar law now or hereafter in effect or seeking the appointment of a trustee, receiver, liquidator, custodian or other similar official of it or any substantial part of its Property, or consents to any such relief or to the appointment of or taking possession by any such official in an involuntary case or other proceeding commenced against it, or fails generally to, or cannot, pay its debts generally as they become due or takes any corporate action to authorize or effect any of the foregoing; or

(h) Assignments for Benefit of Creditors or Admissions of Insolvency - any Obligor makes an assignment for the benefit of its creditors, or admits in writing its insolvency (including any admission of its inability to pay its debts generally as they become due), or consents to the appointment of a receiver, receiver-manager, interim receiver, monitor, trustee, or liquidator of such Obligor or of all or any substantial part of its Property; or

 (i) Undischarged Judgments - a final non-appealable judgment or judgments for the payment of money exceeding, in the aggregate, \$1,000,000 (exclusive of amounts covered by insurance) is rendered by any court or other governmental body against any Obligor and such Obligor does not discharge the same or provide for its discharge in accordance with its terms, or procure a stay of execution thereof within 30 days from the date of entry thereof; or

(j) Security Documents - any Security Document for any reason ceases to create a valid, perfected, first priority Lien on any of the Collateral purported to be covered thereby and securing that portion of the Obligations which is therein designated as being secured, or any Obligor (or any other Person who may have granted or purported to grant such Lien) will so state in writing or U.S. Agent shall cease to have a valid, perfected, first priority Lien upon all of the issued and outstanding equity interests in and to all Subsidiaries (other than Foreign Subsidiaries) of Parent and upon 65% of the issued and outstanding equity interests in and to Foreign Subsidiaries of Parent securing the U.S. Obligations or Canadian Agent shall cease to have a valid, perfected, first priority Lien upon the remaining 35% of the issued and outstanding equity interests in and to Foreign Subsidiaries of Parent securing the Canadian Obligations; or

(k) Ownership Change or Encumbrance - Parent shall cease to own, directly or indirectly, all of the issued and outstanding equity interests in each of the Borrowers and each Material Subsidiary or any Change of Control shall occur.

9.2 Right of Setoff. Upon the occurrence and during the continuance of any Event of Default, each Lender is hereby authorized at any time and from time to time, without notice to any Obligor (any such notice being expressly waived by U.S. Borrower, Canadian Borrower and the other Obligors), to setoff and apply any and all deposits (general or special, time or demand, provisional or final but excluding the funds held in accounts clearly designated as escrow or trust accounts held by U.S. Borrower, Canadian Borrower or any other Obligor for the benefit of Persons which are not Affiliates of any Obligor), whether or not such setoff results in any loss of interest or other penalty, and including without limitation all certificates of deposit, at any time held, and any other funds or Property at any time held, and other indebtedness at any time owing by such Lender to or for the credit or the account of U.S. Borrower, Canadian Borrower or any other Obligor against any and all of the Obligations irrespective of whether or not such Lender or any Agent will have made any demand under this Agreement, the Notes or any other Loan Document. Should the right of any Lender to realize funds in any manner set forth hereinabove be challenged and any application of such funds be reversed, whether by court order or otherwise, the Lenders shall make restitution or refund to U.S. Borrower or Canadian Borrower or the applicable other Obligor, as the case may be, pro rata in accordance with their U.S. Commitments or Canadian Commitments, as the case may be. Each Lender agrees to promptly notify U.S. Borrower, Canadian Borrower and Agents after any such setoff and application, provided that the failure to give such notice will not affect the validity of such setoff and application. The rights of Agents and the Lenders under this Section are in addition to other rights and remedies (including without limitation other rights of setoff) which Agents or the Lenders may have. This Section is subject to the terms and provisions of Sections 4.5 and 11.7 hereof. Any amounts realized under this Section which constitute an asset of Canadian Borrower shall only be applied to the payment of Canadian Obligations.

9.3 Collateral Account. U.S. Borrower hereby agrees, in addition to the provisions of Section 9.1 hereof, that upon the occurrence and during the continuance of any Event of Default, it shall, if requested by any Agent or by the Majority Lenders (through any Agent), pay to U.S. Agent an amount in immediately available funds equal to the then aggregate amount available for drawings under all outstanding U.S. Letters of Credit, which funds shall be held by U.S. Agent as Cover. Canadian Borrower hereby agrees, in addition to the provisions of Section 9.1 hereof, that upon the occurrence and during the continuance of any Event of Default, it shall, if requested by any Agent or by the Majority Lenders (through any Agent), pay to Canadian Agent an amount in immediately available funds equal to the sum of the then aggregate amount available for drawings under all outstanding Canadian Letters of Credit plus the unpaid principal balance of all outstanding Bankers' Acceptances, which funds shall be held by Canadian Agent as Cover.

9.4 Preservation of Security for Unmatured Obligations. In the event that, following (i) the occurrence of an Event of Default and the exercise of any rights available to any Agent or any Lender under the Loan Documents, and (ii) payment in full of the principal amount then outstanding

62

of and the accrued interest on the Loans and Reimbursement Obligations and fees and all other amounts payable hereunder and under the Loan Documents and all other amounts secured by the Security Documents, any Letters of Credit or Bankers' Acceptances shall remain outstanding and undrawn upon, the applicable Agent shall be entitled to hold (and each Borrower and each other Obligor hereby grants and conveys to Agent a security interest in and to) all cash or other Property ("Proceeds of Remedies") realized or arising out of the exercise of any rights available under the Loan Documents, at law or in equity, including, without limitation, the proceeds of any foreclosure, as collateral for the payment of any amounts due or to become due under or in respect of such Letters of Credit and/or such Bankers' Acceptances. Such Proceeds of Remedies shall be held for the ratable benefit of the U.S. Lenders or the Canadian Lenders, as the case may be. The rights, titles, benefits, privileges, duties and obligations of the applicable Agent with respect thereto shall be governed by the terms and provisions of this Agreement and, to the extent not inconsistent with this Agreement, the applicable Security Documents. The applicable Agent may, but shall have no obligation to, invest any such Proceeds of Remedies in such manner as such Agent, in the exercise of its sole discretion, deems appropriate. Such Proceeds of Remedies shall be applied to Reimbursement Obligations arising in respect of any such Letters of Credit, the payment of any Lender's obligations under any such Letter of Credit and/or the Obligations relating to any such Bankers' Acceptance when such Letter of Credit is drawn upon or such Bankers' Acceptance matures, as the case may be. Nothing in this Section shall cause or permit an increase in the maximum amount of the Obligations permitted to be outstanding from time to time under this Agreement. Any amounts realized under this Section which constitute an asset of Canadian Borrower shall only be applied to the payment of Canadian Obligations.

9.5 Currency Conversion After Maturity. At any time following the occurrence of an Event of Default and the acceleration of the maturity of the Obligations owed to the Canadian Lenders hereunder, the Canadian Lenders shall be entitled to convert, with two (2) Business Days' prior notice to Canadian Borrower, any and all or any part of the then unpaid and outstanding LIBOR Borrowings and Base Rate Borrowings of the Canadian Borrower to Canadian Prime Loans. Any such conversion shall be calculated so that the resulting Canadian Prime Loans shall be the equivalent on the date of conversion of the amount of Dollars so converted. Any accrued and unpaid interest denominated in Dollars at the time of any such conversion shall be similarly converted to Canadian Dollars, and such Canadian Prime Loans and accrued and unpaid interest thereon shall thereafter bear interest in accordance with the terms hereof.

9.6 Remedies Cumulative. No remedy, right or power conferred upon any Agent or any Lender is intended to be exclusive of any other remedy, right or power given hereunder or now or hereafter existing at law, in equity, or otherwise, and all such remedies, rights and powers shall be cumulative.

10. Agents.

68

10.1 Appointment, Powers and Immunities. Each U.S. Lender hereby irrevocably appoints and authorizes U.S. Agent to act as its agent hereunder, under the U.S. Letters of Credit and under

the other Loan Documents with such powers as are specifically delegated to U.S. Agent by the terms hereof and thereof, together with such other powers as are reasonably incidental thereto. Each Canadian Lender hereby irrevocably appoints and authorizes Canadian Agent to act as its agent hereunder, under the Canadian Letters of Credit and under the other Loan Documents with such powers as are specifically delegated to Canadian Agent by the terms hereof and thereof, together with such other powers as are reasonably incidental thereto. Any Loan Documents executed in favor of any Agent shall be held by such Agent for the ratable benefit of the applicable Lenders. None of the Agents ("Agents" as used in this Section 10 shall include reference to their Affiliates and their own and their Affiliates' respective officers, shareholders, directors, employees and agents) (a) shall have any duties or responsibilities except those expressly set forth in this Agreement, the Letters of Credit, and the other Loan Documents, and shall not by reason of this Agreement or any other Loan Document be a trustee or fiduciary for any Lender; (b) shall be responsible to any Lender for any recitals, statements, representations or warranties contained in this Agreement, the Letters of Credit or any other Loan Document, or in any certificate or other document referred to or provided for in, or received by any of them under, this Agreement, the Letters of Credit or any other Loan Document, or for the value, validity, effectiveness, genuineness, enforceability, execution, filing, registration, collectibility, recording, perfection, existence or sufficiency of this Agreement, the Letters of Credit, or any other Loan Document or any other document referred to or provided for herein or therein or any Property covered thereby or for any failure by any Obligor or any other Person to perform any of its obligations hereunder or thereunder, or shall have any duty to inquire into or pass upon any of the foregoing matters; (c) shall be required to initiate or conduct any litigation or collection proceedings hereunder or under the Letters of Credit or any other Loan Document except to the extent requested and adequately indemnified by the Majority Lenders; (d) shall be responsible for any mistake of law or fact or any action taken or omitted to be taken by it hereunder or under the Letters or Credit or any other Loan Document or any other document or instrument referred to or provided for herein or therein or in connection herewith or therewith, including, without limitation, pursuant to its own negligence, except for its own gross negligence or willful misconduct; (e) shall be bound by or obliged to recognize any agreement among or between any Borrower and any Lender to which such Agent is not a party, regardless of whether such Agent has knowledge of the existence of any such agreement or the terms and provisions thereof; (f) shall be charged with notice or knowledge of any fact or information not herein set out or provided to such Agent in accordance with the terms of this Agreement or any other Loan Document; (g) shall be responsible for any delay, error, omission or default of any mail, telegraph, cable or wireless agency or operator, and (h) shall be responsible for the acts or edicts of any Governmental Authority. Any Agent may employ agents and attorneys-in-fact and none of the Agents shall be responsible for the negligence or misconduct of any such agents or attorneys-in-fact selected by it with reasonable care. Without in any way limiting any of the foregoing, each Lender acknowledges that none of the Agents (nor any Issuer) shall have greater responsibility in the operation of the Letters of Credit than is specified in the Uniform Customs and Practice for Documentary Credits (1993 Revision, International Chamber of Commerce Publication No. 500). In any foreclosure proceeding concerning any Collateral, each holder of an Obligation if bidding for its own account or for its own account and the accounts of other Lenders is prohibited from including in the amount of its bid an amount to be applied as a credit against the Obligations held by it or the Obligations held by the other Lenders; instead, such

69

holder must bid in cash only. However, in any such foreclosure proceeding, (i) U.S. Agent may (but shall not be obligated to) submit a bid for all U.S. Lenders (including itself) in the form of a credit against the U.S. Obligations, and U.S. Agent or its designee may (but shall not be obligated to) accept title to such collateral for and on behalf of all U.S. Lenders and (ii) Canadian Agent may (but shall not be obligated to) submit a bid for all Canadian Lenders (including itself) in the form of a credit against the Canadian Obligations, and Canadian Agent or its designee may (but shall not be obligated to) accept title to such collateral for and on behalf of all Canadian Lenders

Reliance. Each Agent shall be entitled to rely upon any 10.2 certification, notice or other communication (including any thereof by telephone, telex, telegram or cable) believed by it to be genuine and correct and to have been signed or sent by or on behalf of the proper Person or Persons, and upon advice and statements of legal counsel (which may be counsel for any Borrower), independent accountants and other experts selected by such None of the Agents shall be required in any way to determine the Agent. identity or authority of any Person delivering or executing the same. As to any matters not expressly provided for by this Agreement, the Letters of Credit, or any other Loan Document, each Agent shall in all cases be fully protected in acting, or in refraining from acting, hereunder and thereunder in accordance with instructions of the Majority Lenders, and any action taken or failure to act by U.S. Agent pursuant thereto shall be binding on all of the U.S. Lenders and any action taken or failure to act by Canadian Agent pursuant thereto shall be binding on all of the Canadian Lenders. Pursuant to instructions of the Majority Lenders, the Agents shall have the authority to execute releases of the Security Documents on behalf of the Lenders without the joinder of any Lender. If any order, writ, judgment or decree shall be made or entered by any court affecting the rights, duties and obligations of any Agent under this Agreement or any other Loan Document, then and in any of such events such Agent is authorized, in its sole discretion, to rely upon and comply with such order, writ, judgment or decree which it is advised by legal counsel of its own choosing is binding upon it under the terms of this Agreement, the relevant Loan Document or otherwise; and if such Agent complies with any such order, writ, judgment or decree, then it shall not be liable to any Lender or to any other Person by reason of such compliance even though such order, writ, judgment or decree may be subsequently reversed, modified, annulled, set aside or vacated.

10.3 None of the Agents shall be deemed to have knowledge Defaults. of the occurrence of a Default (other than the non-payment of principal of or interest on Loans or Reimbursement Obligations) unless such Agent has received notice from a Lender or a Borrower specifying such Default and stating that such notice is a "Notice of Default." In the event that any Agent receives such a Notice of Default, such Agent shall give prompt notice thereof to the Lenders (and shall give each Lender prompt notice of each such non-payment). Each Agent shall (subject to Section 10.7 hereof) take such action with respect to such Notice of Default as shall be directed by the Majority Lenders and within its rights under the Loan Documents and at law or in equity, provided that, unless and until an Agent shall have received such directions, such Agent may (but shall not be obligated to) take such action, or refrain from taking such action, permitted hereby with respect to such Notice of Default as it shall deem advisable in the best interests of the Lenders and within its rights under the Loan Documents, at law or in equity.

10.4 Material Written Notices. In the event that any Agent receives any written notice of a material nature from any Borrower or any Obligor under the Loan Documents, such Agent shall promptly inform each of the Lenders thereof.

71

10.5 Rights as a Lender. With respect to its Commitments and the Obligations, each of TCB and Chase Canada, in its capacity as a Lender hereunder, shall have the same rights and powers hereunder as any other Lender and may exercise the same as though it were not acting in its agency capacity, and the term "Lender" or "Lenders" shall, unless the context otherwise indicates, include each Agent in its individual capacity. Each Agent may (without having to account therefor to any Lender) accept deposits from, lend money to and generally engage in any kind of banking, trust, letter of credit, agency or other business with any Borrower (and any of their Affiliates) as if it were not acting as an Agent, and each Agent may accept fees and other consideration from any Borrower (in addition to the fees heretofore agreed to between any Borrower and any Agent) for services in connection with this Agreement or otherwise without having to account for the same to the Lenders.

Indemnification. The Canadian Lenders and the U.S. Lenders, 10.6 respectively, agree to indemnify Canadian Agent and U.S. Agent, respectively (to the extent not reimbursed under Section 2.2(c), Section 11.3 or Section 11.4 hereof, but without limiting the obligations of any Borrower under said Sections 2.2(c), 11.3 and 11.4), ratably in accordance with the sum of the applicable Lenders' respective Commitments, for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind and nature whatsoever, REGARDLESS OF WHETHER CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE OF ANY INDEMNIFIED PARTIES, which may be imposed on, incurred by or asserted against the applicable Agent in any way relating to or arising out of this Agreement, the Letters of Credit or any other Loan Document or any other documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby (including, without limitation, the costs and expenses which any Borrower is obligated to pay under Sections 2.2(c), 11.3 and 11.4 hereof, interest, penalties, attorneys' fees and amounts paid in settlement, but excluding, unless a Default has occurred and is continuing, normal administrative costs and expenses incident to the performance of its agency duties hereunder) or the enforcement of any of the terms hereof or thereof or of any such other documents; provided that no Lender shall be liable for any of the foregoing to the extent they arise from the gross negligence or willful misconduct of the party to be indemnified. The obligations of the Lenders under this Section 10.6 shall survive the termination of this Agreement and the repayment of the Obligations.

10.7 Non-Reliance on Agents and Other Lenders. Each Lender agrees that it has received current financial information with respect to each Borrower and each other Obligor and that it has, independently and without reliance on any Agent or any other Lender and based on such documents and information as it has deemed appropriate, made its own credit analysis of each Borrower and each other Obligor and decision to enter into this Agreement and that it will, independently and without reliance upon any Agent or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own analysis and decisions in taking or not taking action under this Agreement or any of the other Loan Documents. None of the Agents

shall be required to keep itself informed as to the performance or observance by any Obligor of this Agreement, the Letters of Credit or any of the other Loan Documents or any other document referred to or provided for herein or therein or to inspect the properties or books of any Obligor. Except for notices, reports and other documents and information expressly required to be furnished to the Lenders by an Agent hereunder, under the Letters of Credit or the other Loan Documents, none of the Agents shall have any duty or responsibility to provide any Lender with any credit or other information concerning the affairs, financial condition or business of any Obligor (or any of their affiliates) which may come into the possession of any Agent.

10.8 Failure to Act. Except for action expressly required of an Agent hereunder, under the Letters of Credit or under the other Loan Documents, each Agent shall in all cases be fully justified in failing or refusing to act hereunder and thereunder unless it shall receive further assurances to its satisfaction by the Lenders of their indemnification obligations under Section 10.6 hereof against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action.

Resignation or Removal of Agent. Subject to the appointment and 10.9 acceptance of a successor U.S. Agent or Canadian Agent, as the case may be, provided below, U.S. Agent and Canadian Agent, respectively, may resign at as any time by giving notice thereof to the U.S. Lenders and the Canadian Lenders, respectively, and to U.S. Borrower and Canadian Borrower, respectively. Any Agent may be removed at any time with or without cause by the Majority Lenders; provided, that such Agent shall continue as U.S. Agent or Canadian Agent, as the case may be, until such time as any successor shall have accepted appointment hereunder as U.S. Agent or Canadian Agent, as the case may be. Upon any such resignation or removal, (i) the Majority Lenders without the consent of Borrower shall have the right to appoint a successor U.S. Agent or Canadian Agent, as the case may be, so long as such successor U.S. Agent or Canadian Agent, as the case may be, is also a Lender at the time of such appointment and (ii) the Majority Lenders shall have the right to appoint a successor U.S. Agent or Canadian Agent, as the case may be, that is not a Lender at the time of such appointment so long as Borrower's consent to such appointment (which consent shall not be unreasonably withheld). If no successor U.S. Agent or Canadian Agent, as the case may be, shall have been so appointed by the Majority Lenders and accepted such appointment within 30 days after the retiring U.S. Agent's or Canadian Agent's, as the case may be, giving of notice of resignation or the Majority Lenders' removal of the retiring U.S. Agent or Canadian Agent, as the case may be, then the retiring Agent may, on behalf of the applicable Lenders, appoint a successor U.S. Agent or Canadian Agent, as the case may be, without the necessity of any consent on the part of any Borrower or any Lender. Any successor U.S. Agent shall be a bank which has an office in the United States and a combined capital and surplus of at least \$250,000,000 and any successor Canadian Agent shall be a bank which has an office in Canada and a combined capital and surplus of at least C\$250,000,000. Upon the acceptance of any appointment as U.S. Agent or Canadian Agent, as the case may be, hereunder by a successor Agent, such successor Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Agent and the retiring Agent shall be discharged from its duties and obligations hereunder and under any other Loan Documents. Such successor Agent shall promptly specify by notice to Borrowers its

67

Principal Office referred to in Section 3.1 and Section 4 hereof. After any retiring Agent's resignation or removal hereunder as an Agent, the provisions of this Section 10 shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as Agent.

10.10 No Partnership. Neither the execution and delivery of this Agreement nor any of the other Loan Documents nor any interest the Lenders, Agents or any of them may now or hereafter have in all or any part of the Obligations shall create or be construed as creating a partnership, joint venture or other joint enterprise between the Lenders or among the Lenders and any Agent. The relationship between the Lenders, on the one hand, and any Agent, on the other, is and shall be that of principals and agent only, and nothing in this Agreement or any of the other Loan Documents shall be construed to constitute any Agent as trustee or other fiduciary for any Lender or to impose on any Agent any duty, responsibility or obligation other than those expressly provided for herein and therein.

11. Miscellaneous.

73

11.1 Waiver. No waiver of any Default or Event of Default shall be a waiver of any other Default or Event of Default. No failure on the part of any Agent or any Lender to exercise and no delay in exercising, and no course of dealing with respect to, any right, power or privilege under any Loan Document shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or privilege thereunder preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The remedies provided in the Loan Documents are cumulative and not exclusive of any remedies provided by law or in equity.

11.2 Notices. All notices and other communications provided for herein (including, without limitation, any modifications of, or waivers or consents under, this Agreement) shall be given or made by telex, telegraph, telecopy (confirmed by mail), cable or other writing and telexed, telecopied, telegraphed, cabled, mailed or delivered to the intended recipient at the "Address for Notices" specified below its name on the signature pages hereof (or provided for in an Assignment and Acceptance); or, as to any party hereto, at such other address as shall be designated by such party in a notice (given in accordance with this Section) (i) as to any Borrower, to Agents, (ii) as to U.S. Agent, to U.S. Borrower and to each U.S. Lender, (iii) as to Canadian Agent, to Canadian Borrower and to each Canadian Lender, (iv) as to any U.S. Lender, to U.S. Borrower and Agents and (v) as to any Canadian Lender, to Canadian Borrower and Agents. Except as otherwise provided in this Agreement, all such notices or communications shall be deemed to have been duly given when (a) transmitted by telex or telecopier or delivered to the telegraph or cable office, (b) personally delivered (c) one Business Day after deposit with an overnight mail or delivery service, postage prepaid or (d) three Business Days' after deposit in a receptacle maintained by the United States Postal Service or Canada Post, as the case may be, postage prepaid, registered or certified mail, return receipt requested, in each case given or addressed as aforesaid.

Expenses, Etc. Whether or not any Loan is ever made or any 11.3Bankers' Acceptances ever accepted and purchased or any Letter of Credit ever issued, Borrowers shall pay or reimburse within 10 days after written demand (a) any Agent for paying the reasonable fees and expenses of legal counsel to such Agent, together with the reasonable fees and expenses of each local counsel to such Agent, in connection with the preparation, negotiation, execution and delivery of this Agreement (including the exhibits and schedules hereto), the Security Documents and the other Loan Documents and the making of the Loans and the acceptance and purchase of Bankers' Acceptances and the issuance of Letters of Credit hereunder, and any modification, supplement or waiver of any of the terms of this Agreement, the Letters of Credit or any other Loan Document; (b) any Agent for any Lien search fees, collateral audit fees, appraisal fees, survey fees, environmental study fees, and title insurance costs and premiums; (c) any Agent for reasonable out-of-pocket expenses incurred in connection with the preparation, documentation, administration and syndication of any of the Loan Documents (including, without limitation, the advertising, marketing, printing, publicity, duplicating, mailing and similar expenses) or any of the Obligations; (d) any Agent or any Lender for paying all transfer, stamp, documentary or other similar taxes, assessments or charges levied by any governmental or revenue authority in respect of this Agreement, any Letter of Credit or any other Loan Document or any other document referred to herein or therein; (e) any Agent for paying all costs, expenses, taxes, assessments and other charges incurred in connection with any filing, registration, recording or perfection of any security interest contemplated by this Agreement, any Security Document or any document referred to herein or therein, and (f) following the occurrence and during the continuation of an Event of Default, any Lender or any Agent for paying all amounts reasonably expended, advanced or incurred by such Lender or such Agent to satisfy any obligation of any Obligor under this Agreement or any other Loan Document, to protect the Collateral, to collect the Obligations or to enforce, protect, preserve or defend the rights of the Lenders or Agents under this Agreement or any other Loan Document, including, without limitation, fees and expenses incurred in connection with such Lender's or such Agent's participation as a member of a creditor's committee in a case commenced under the Bankruptcy Code or other similar law, fees and expenses incurred in connection with lifting the automatic stay prescribed in Section 362 of the Bankruptcy Code and fees and expenses incurred in connection with any action pursuant to Section 1129 of the Bankruptcy Code and all other customary outof-pocket expenses incurred by such Lender or such Agent in connection with such matters, together with interest thereon at the Past Due Rate applicable to U.S. Loans on each such amount from the due date of payment until the date of reimbursement to such Lender or such Agent.

74

11.4 Indemnification. Borrowers, jointly and severally, shall indemnify each Agent, each Lender and each affiliate thereof and their respective directors, officers, employees and agents from, and hold each of them harmless against, any and all losses, liabilities, claims or damages to which any of them may become subject, REGARDLESS OF WHETHER CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE OF ANY INDEMNIFIED PARTIES, insofar as such losses, liabilities, claims or damages arise out of or result from any (i) actual or proposed use by any Borrower of the proceeds of any extension of credit (whether a Loan, a Bankers' Acceptance or a Letter of Credit) by any Lender hereunder; (ii) breach by any Obligor of this Agreement or any other Loan Document; (iii) violation by any Obligor of any Legal Requirement, or (iv) investigation,

litigation or other proceeding relating to any of the foregoing, and Borrowers, jointly and severally, shall reimburse each Agent, each Lender, and each Affiliate thereof and their respective directors, officers, employees and agents, upon demand for any reasonable expenses (including reasonable legal fees) incurred in connection with any such investigation or proceeding; provided, however, that none of the Borrowers shall have any obligations pursuant to this Section with respect to any losses, liabilities, claims, damages or expenses incurred by the Person seeking indemnification by reason of the gross negligence or willful misconduct of that Person or with respect to any disputes between or among any and all of Agents, Lenders and Issuers. Nothing in this Section is intended to limit the obligations of any Borrower under any other provision of this Agreement. Each Agent and each Lender, respectively, shall indemnify Borrowers and hold Borrowers harmless from and against the gross negligence or willful misconduct of such Agent or such Lender, as the case may be. Nothing in this Section shall render Canadian Borrower liable in respect of the U.S. Obligations.

Amendments, Etc. No amendment or modification of this Agreement, 11.5 the Notes or any other Loan Document shall in any event be effective against any Borrower unless the same shall be agreed or consented to in writing by the applicable Borrower. No amendment, modification or waiver of any provision of this Agreement, the Notes or any other Loan Document, nor any consent to any departure by any Borrower therefrom, shall in any event be effective against the Lenders unless the same shall be agreed or consented to in writing by the Majority Lenders, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, that no amendment, modification, waiver or consent shall, unless in writing and signed by each Lender affected thereby, do any of the following: (a) increase any Commitment of any of the Lenders (or reinstate any termination or reduction of the Commitments) or subject any of the Lenders to any additional obligations; (b) reduce the principal of, or interest on, any Loan, Reimbursement Obligation, fee or other amount due hereunder; (c) postpone or extend the Maturity Date, the Termination Date, the Availability Period or any scheduled date fixed for any payment of principal of, or interest on, any Loan, Reimbursement Obligation, fee or other sum to be paid hereunder or waive any Event of Default described in Section 9.1(a) hereof; (d) change the percentage of any of the Commitments or of the aggregate unpaid principal amount of Obligations, or the percentage of Lenders, which shall be required for the Lenders or any of them to take any action under this Agreement; (e) change any provision contained in Sections 2.2(c), 7.9, 11.3 or 11.4 hereof or this Section 11.5, or (f) release any Person from liability under a Guaranty or release all or substantially all of the security for the Obligations or release Collateral (exclusive of Collateral with respect to which any Agent is obligated to provide a release pursuant to this Agreement or any of the other Loan Documents or by law) in any one (1) calendar year ascribed an aggregate value on the most recent financial statements of the applicable Borrower delivered to Agents in excess of \$1,000,000. Notwithstanding anything in this Section 11.5 to the contrary, no amendment, modification, waiver or consent shall be made with respect to Section 10 without the consent of U.S. Agent to the extent it affects U.S. Agent, as U.S. Agent or Canadian Agent to the extent it affects Canadian Agent, as Canadian Agent.

70

76

This Agreement shall be binding upon and inure to the benefit of (a) Borrowers, Agents and the Lenders and their respective successors and assigns; provided, however, that no Borrower may assign or transfer any of its rights or obligations hereunder without the prior written consent of all of the Lenders, and any such assignment or transfer without such consent shall be null and void. Each Lender may sell participations to any Person in all or part of any Loan or Bankers' Acceptance, or all or part of its Notes, Commitments or interests in Letters of Credit or Bankers' Acceptances, in which event, without limiting the foregoing, the provisions of the Loan Documents shall inure to the benefit of each purchaser of a participation; provided, however, the pro rata treatment of payments, as described in Section 4.2 hereof, shall be determined as if such Lender had not sold such participation. No Lender that sells one or more participations to any Person shall be relieved by virtue of such participation from any of its obligations to Borrowers under this Agreement. In the event any Lender shall sell any participation, such Lender shall retain the sole right and responsibility to enforce the obligations of Borrowers hereunder, including, without limitation, the right to approve any amendment, modification or waiver of any provision of this Agreement other than amendments, modifications or waivers with respect to (i) any fees payable hereunder to the Lenders, (ii) the amount of principal or the rate of interest payable on, or the dates fixed for the scheduled repayment of principal of, any of the Obligations and (iii) the release of the Liens on all or substantially all of the Collateral.

Each U.S. Lender may assign to one or more U.S. Lenders or any (b) other Person all or a portion of its interests, rights and obligations under this Agreement; provided, however, that (i) the aggregate amount of the Commitments of the assigning U.S. Lender subject to each such assignment shall in no event be less than \$10,000,000 and (ii) other than in the case of an assignment to another U.S. Lender (that is, at the time of the assignment, a party hereto) or to an Affiliate of such U.S. Lender or to a Federal Reserve Bank, Agents and, so long as no Event of Default shall have occurred and be continuing, U.S. Borrower must each give its prior written consent, which consents shall not be unreasonably withheld. Each Canadian Lender may assign to one or more Canadian Lenders or any other Person all or a portion of its interests, rights and obligations under this Agreement; provided, however, that (i) the aggregate amount of the Commitments of the assigning Canadian Lender subject to each such assignment shall in no event be less than \$10,000,000 and (ii) other than in the case of an assignment to another Canadian Lender (that is, at the time of the assignment, a party hereto) or to an Affiliate of such Canadian Lender, Agents and, so long as no Event of Default shall have occurred and be continuing, Canadian Borrower must each give its prior written consent, which consents shall not be unreasonably withheld. As a condition precedent to any such assignment, the parties to each such assignment shall execute and deliver to the applicable Agent, for its acceptance an Assignment and Acceptance in the form of Exhibit E hereto (each an "Assignment and Acceptance") with blanks appropriately completed, together with any Note or Notes subject to such assignment and a processing and recording fee of \$3,000 paid by the assignee (for which Borrowers will have no liability). Upon such execution, delivery and acceptance, from and after the effective date specified in each Assignment and Acceptance, (A) the assignee thereunder shall be a party hereto and, to the extent provided in such Assignment and Acceptance,

have the rights and obligations of a Lender hereunder and (B) the Lender thereunder shall, to the extent provided in such Assignment and Acceptance, be released from its obligations under this Agreement (and, in the case of an Assignment and Acceptance covering all or the remaining portion of an assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto except in respect of provisions of this Agreement which survive payment of the Obligations and termination of the Commitments). Notwithstanding anything contained in this Agreement to the contrary, any Lender may at any time assign all or any portion of its rights under this Agreement and the other Loan Documents as collateral to a Federal Reserve Bank; provided that no such assignment shall release such Lender from any of its obligations hereunder.

77

By executing and delivering an Assignment and Acceptance, the (C) assignor and assignee thereunder confirm to and agree with each other and the other parties hereto as follows: (i) other than the representation and warranty that it is the legal and beneficial owner of the interest being assigned thereby free and clear of any adverse claim, the applicable assignor makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with this Agreement or any of the other Loan Documents or the execution, legality, validity, enforceability, genuineness, sufficiency or value of this Agreement or any of the other Loan Documents or any other instrument or document furnished pursuant thereto; (ii) the applicable assignor makes no representation or warranty and assumes no responsibility with respect to the financial condition of any Borrower or the performance or observance by any Borrower of any of its obligations under this Agreement or any of the other Loan Documents or any other instrument or document furnished pursuant hereto; (iii) the applicable assignee confirms that it has received a copy of this Agreement, together with copies of the financial statements referred to in Section 6.2 hereof and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into such Assignment and Acceptance; (iv) the applicable assignee will, independently and without reliance upon any Agent, the applicable assignor or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement and the other Loan Documents; (v) the applicable assignee appoints and authorizes U.S. Agent or Canadian Agent, as the case may be, to take such action as agent on its behalf and to exercise such powers under this Agreement and the other Loan Documents as are delegated to such Agent by the terms hereof, together with such powers as are reasonably incidental thereto; and (vi) the applicable assignee agrees that it will perform in accordance with their terms all obligations that by the terms of this Agreement and the other Loan Documents are required to be performed by it as a Lender.

(d) The entries in the records of each applicable Agent as to each Assignment and Acceptance delivered to it and the names and addresses of the Lenders and the Commitments of, and principal amount of the Obligations owing to, each Lender from time to time shall be conclusive, in the absence of manifest error, and Borrowers, Agents and the Lenders may treat each Person the name of which is recorded in the books and records of the applicable Agent as a Lender hereunder for all purposes of this Agreement and the other Loan Documents.

(e) Upon the applicable Agent's receipt of an Assignment and Acceptance executed by an assigning Lender and the assignee thereunder, together with any Note or Notes subject to such assignment and the written consent to such assignment (to the extent consent is required), such Agent shall, if such Assignment and Acceptance has been completed with blanks appropriately filled, (i) accept such Assignment and Acceptance, (ii) record the information contained therein in its records and (iii) give prompt notice thereof to the applicable Borrower. Within five Business Days after receipt of notice, the applicable Borrower, at its own expense, shall execute and deliver to the applicable Agent new Notes payable to the order of such assignee in the appropriate amounts and, if the assigning Lender has retained Commitments hereunder, new Notes to the order of the assigning Lender in the appropriate amounts. Such new Notes shall be dated the effective date of such Assignment and Acceptance and shall otherwise be in the forms required hereunder.

(f) Any Lender may, in connection with any assignment or participation or proposed assignment or participation pursuant to this Section 11.6, disclose to the assignee or participant or proposed assignee or participant, any information relating to any Borrower furnished to such Lender by or on behalf of any Borrower.

11.7 Limitation of Interest. Borrowers and the Lenders intend to strictly comply with all applicable federal and Texas laws, including applicable usury laws (or the usury laws of any jurisdiction, including Canada, whose usury laws are deemed to apply to the Notes or any other Loan Documents despite the intention and desire of the parties to apply the usury laws of the State of Texas). Accordingly, the provisions of this Section 11.7 shall govern and control over every other provision of this Agreement or any other Loan Document which conflicts or is inconsistent with this Section, even if such provision declares that it controls. As used in this Section, the term "interest" includes the aggregate of all charges, fees, benefits or other compensation which constitute interest under applicable law, provided that, to the maximum extent permitted by applicable law, (a) any non-principal payment shall be characterized as an expense or as compensation for something other than the use, forbearance or detention of money and not as interest, and (b) all interest at any time contracted for, reserved, charged or received shall be amortized, prorated, allocated and spread, in equal parts during the full term of the Obligations. In no event shall Borrowers or any other Person be obligated to pay, or any Lender have any right or privilege to reserve, receive or retain, (a) any interest in excess of the maximum amount of nonusurious interest permitted under the laws of the State of Texas or the applicable laws (if any) of the United States or of any other applicable jurisdiction, or (b) total interest in excess of the amount which such Lender could lawfully have contracted for, reserved, received, retained or charged had the interest been calculated for the full term of the Obligations at the Ceiling Rate. The daily interest rates to be used in calculating interest at the Ceiling Rate shall be determined by dividing the applicable Ceiling Rate per annum by the number of days in the calendar year for which such calculation is being made. None of the terms and provisions contained in this Agreement or in any other Loan Document (including, without limitation, Section 9.1 hereof) which directly or indirectly relate to interest shall ever be construed without reference to this Section 11.7, or be construed to create a contract to pay for the use, forbearance or detention of money at an interest rate in excess of the Ceiling Rate. If the term of any Obligation is shortened by reason of acceleration of maturity as a result of any Default or by any

73

other cause, or by reason of any required or permitted prepayment, and if for that (or any other) reason any Lender at any time, including but not limited to, the stated maturity, is owed or receives (and/or has received) interest in excess of interest calculated at the Ceiling Rate, then and in any such event all of any such excess interest shall be canceled automatically as of the date of such acceleration, prepayment or other event which produces the excess, and, if such excess interest has been paid to such Lender, it shall be credited pro tanto against the then-outstanding principal balance of the applicable Borrower's obligations to such Lender, effective as of the date or dates when the event occurs which causes it to be excess interest, until such excess is exhausted or all of such principal has been fully paid and satisfied, whichever occurs first, and any remaining balance of such excess shall be promptly refunded to its payor.

11.8 Survival. The obligations of Borrowers under Sections 2.2(c), 2.2(d), 7.9, 11.3 and 11.4 hereof and all other obligations of Borrowers in any other Loan Document (to the extent stated therein), the obligations of each Issuer under the last sentence of Section 2.2(b)(iii) and the obligations of the Lenders under Section 10.5 and 11.7 hereof, shall, notwithstanding anything herein to the contrary, survive the repayment of the Loans and Reimbursement Obligations and the termination of the Commitments and the Letters of Credit.

11.9 Captions. Captions and section headings appearing herein are included solely for convenience of reference and are not intended to affect the interpretation of any provision of this Agreement.

11.10 Counterparts. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same agreement and any of the parties hereto may execute this Agreement by signing any such counterpart.

11.11 Venue; Governing Law. This Agreement and (except as therein provided) the other Loan Documents are performable in Harris County, Texas, which shall be a proper place of venue for suit on or in respect thereof. Each Borrower irrevocably agrees that any legal proceeding in respect of this Agreement or the other Loan Documents shall be brought in the district courts of Harris County, Texas or the United States District Court for the Southern District of Texas, Houston Division (collectively, the "Specified Courts"). Each Borrower hereby irrevocably submits to the nonexclusive jurisdiction of the state and federal courts of the State of Texas. Each Borrower hereby irrevocably waives, to the fullest extent permitted by law, any objection which it may now or hereafter have to the laying of venue of any suit, action or proceeding arising out of or relating to any Loan Document brought in any Specified Court, and hereby further irrevocably waives any claims that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum. Each Borrower further (1) agrees to designate and maintain an agent for service of process in the State of Texas in connection with any such suit, action or proceeding upon request by any Agent and to deliver to Agents evidence thereof and (2) irrevocably consents to the service of process out of any of the aforementioned courts in any such suit, action or proceeding by the mailing of copies thereof by certified mail, return receipt requested, postage prepaid, to such Borrower at its address as provided in this Agreement or as otherwise provided by applicable law.

Nothing herein shall affect the right of any Agent or any Bank to commence legal proceedings or otherwise proceed against any Borrower in any jurisdiction or to serve process in any manner permitted by applicable law. Each Borrower agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. THIS AGREEMENT AND (EXCEPT AS THEREIN PROVIDED) THE OTHER LOAN DOCUMENTS SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE APPLICABLE LAWS OF THE STATE OF TEXAS AND THE UNITED STATES OF AMERICA FROM TIME TO TIME IN EFFECT; PROVIDED, HOWEVER, THAT, EXCEPT AS MAY BE REQUIRED UNDER APPLICABLE LAWS, THE USURY LAWS OF THE STATE OF TEXAS OR THE UNITED STATES OF AMERICA SHALL NOT APPLY TO ADVANCES MADE IN CANADA BY CANADIAN LENDERS TO CANADIAN BORROWER, BUT RATHER THE USURY LAWS OF THE PROVINCE OF ONTARIO AND THE FEDERAL LAWS OF CANADA APPLICABLE THEREIN SHALL GOVERN IN SUCH CONTEXT.

11.12 Severability. Whenever possible, each provision of the Loan Documents shall be interpreted in such manner as to be effective and valid under applicable law. If any provision of any Loan Document shall be invalid, illegal or unenforceable in any respect under any applicable law, the validity, legality and enforceability of the remaining provisions of such Loan Document shall not be affected or impaired thereby.

11.13 Tax Forms; Net Payments.

Each U.S. Lender which is organized under the laws of a (a) jurisdiction outside the United States shall, on the day of the initial borrowing from each such U.S. Lender hereunder and from time to time thereafter if requested by U.S. Borrower or U.S. Agent, provide U.S. Agent and U.S. Borrower with the forms prescribed by the Internal Revenue Service of the United States certifying as to such U.S. Lender's status for purposes of determining exemption from United States withholding taxes with respect to all payments to be made to such U.S. Lender hereunder or other documents satisfactory to such U.S. Lender, U.S. Borrower and U.S. Agent indicating that all payments to be made to such U.S. Lender hereunder are not subject to United States withholding tax or are subject to such tax at a rate reduced by an applicable tax treaty. Unless U.S. Borrower and U.S. Agent shall have received such forms or such documents indicating that payments to such U.S. Lender hereunder are not subject to United States withholding tax or are subject to such tax at a rate reduced by an applicable tax treaty, U.S. Borrower and U.S. Agent shall be entitled to withhold taxes from such payments at the applicable statutory rate.

(b) Except as provided in Section 11.13(a), (i) all payments (whether of principal, interest, fees, reimbursements or otherwise) by Borrowers under this Agreement and the other Loan Documents shall be made without set-off or counterclaim and shall be made free and clear of and without deduction for any present or future tax, levy, impost or any other charge, if any, of any nature whatsoever now or hereafter imposed by any taxing authority and (ii) if the making of such payments by any Borrower is prohibited by law unless such a tax, levy, impost or other charge is deducted or

75

withheld therefrom, the applicable Borrower shall pay to the relevant Agent, on the date of each such payment, such additional amounts (without duplication of any amounts required to be paid by the applicable Borrower hereunder) as may be necessary in order that net amounts received by the Lenders after such deduction or withholding shall equal the amounts which would have been received if such deduction or withholding were not required. The applicable Borrower shall confirm that all applicable taxes, if any, imposed on this Agreement or transactions hereunder shall have been properly and legally paid by it to the appropriate taxing authorities by sending official tax receipts or notarized copies of such receipts to the relevant Agent within 30 days after payment of any applicable tax. Notwithstanding the foregoing, in no event shall the compensation payable under this Section (to the extent, if any, constituting interest under applicable laws) together with all amounts constituting interest under applicable laws and payable in connection with this Agreement and the other Loan Documents exceed the Ceiling Rate.

(c) Each Canadian Lender is a resident of Canada for purposes of the Income Tax Act (Canada).

11.14 Interest Act (Canada). Whenever interest is calculated on the basis of a year of 360 or 365 days, for the purposes of the Interest Act (Canada), the yearly rate of interest which is equivalent to the rate payable hereunder is the rate payable multiplied by the actual number of days in the year and divided by 360 or 365, as the case may be. All interest will be calculated using the nominal rate method and not the effective rate method and the deemed reinvestment principle shall not apply to such calculations.

11.15 Judgment Currency. The obligation of each Borrower to make payments on any Obligation to the Lenders or to any Agent hereunder in any currency (the "first currency") shall not be discharged or satisfied by any tender or recovery pursuant to any judgment expressed in or converted into any other currency (the "second currency") except to the extent to which such tender or recovery shall result in the effective receipt by the applicable Lender or the applicable Agent of the full amount of the first currency payable, and accordingly the primary obligation of each Borrower shall be enforceable as an alternative or additional cause of action for the purpose of recovery in the second currency of the amount (if any) by which such effective receipt shall fall short of the full amount of the full currency payable and shall not be affected by a judgment being obtained for any other sum due hereunder.

11.16 Conflicts Between This Agreement and the Other Loan Documents. In the event of any conflict between the terms of this Agreement and the terms of any of the other Loan Documents, the terms of this Agreement shall control.

11.17 Limitation on Charges; Substitute Lenders; Non-Discrimination. Anything in Sections 3.3(c) or 7.9 notwithstanding:

76

(1) No Borrower shall be required to pay to any Lender reimbursement with regard to any costs or expenses described in such Sections, unless such Lender notifies the applicable Borrower of such costs or expenses within 90 days after the date paid or incurred;

(2) none of the Lenders shall be permitted to pass through to any Borrower charges and costs under such Sections on a discriminatory basis (i.e., which are not also passed through by such Lender to other customers of such Lender similarly situated where such customer is subject to documents providing for such pass through); and

if any Lender elects to pass through to any Borrower any (3)material charge or cost under such Sections or elects to terminate the availability of LIBOR Borrowings for any material period of time, the applicable Borrower may, within 60 days after the date of such event and so long as no Default shall have occurred and be continuing, elect to terminate such Lender as a party to this Agreement; provided that, concurrently with such termination such Borrower shall (i) if Agents and each of the other Lenders shall consent, pay that Lender all principal, interest and fees and other amounts owed to such Lender through such date of termination or (ii) have arranged for another financial institution approved by Agents (such approval not to be unreasonably withheld) as of such date, to become a substitute Lender for all purposes under this Agreement in the manner provided in Section 11.6; provided further that, prior to substitution for any Lender, the applicable Borrower shall have given written notice to Agents of such intention and the Lenders shall have the option, but no obligation, for a period of 60 days after receipt of such notice, to increase their Commitments in order to replace the affected Lender in lieu of such substitution.

11.18 Multiple Persons Comprising Canadian Borrower. The parties recognize and acknowledge that "Canadian Borrower" is comprised of more than one Person. All duties and obligations imposed on "Canadian Borrower" hereunder or under any of the other Loan Documents shall be the joint and several duties and obligations of each Person comprising "Canadian Borrower", notwithstanding the manner in which any of such obligations shall be allocated among the Persons comprising "Canadian Borrower" for the purposes of such Persons' internal records and accounts. Whenever this Agreement or any of the other Loan Documents requires the execution and delivery of any document or instrument by "Canadian Borrower", such requirement may be satisfied by execution and delivery by any single one of the Persons comprising "Canadian Borrower", and execution and delivery of any document or instrument by any single one of the Persons comprising "Canadian Borrower" shall be binding upon each of the Persons comprising "Canadian Borrower".

77

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the Effective Date.

83

NATIONAL-OILWELL, L.P., a Delaware limited partnership, as U.S. Borrower Its Sole General Partner, NOW Oilfield By: Services, Inc., a Delaware corporation By: /s/ DANIEL L. MOLINARO -----Name: Daniel L. Molinaro -----Title: Treasurer -----NATIONAL-OILWELL CANADA LTD., a British Columbia company, as Canadian Borrower By: /s/ DANIEL L. MOLINARO Name: Daniel L. Molinaro -----Title: Assistant Treasurer -----DRECO ENERGY SERVICES LTD., an Alberta corporation, as Canadian Borrower By: /s/ PAUL M. NATION -----Name: Paul M. Nation -----Title: Assistant Secretary -----

Address for Notices:

National-Oilwell, Inc. 5555 San Felipe, 16th Floor Houston, Texas 77056 Attention: Mr. Daniel L. Molinaro Telecopy No.: (713) 960-5212

with a copies to:

Dreco Energy Services Ltd. Suite 1340 Weber Centre 5555 Calgary Trail Edmonton, Alberta T6H 5P9 CANADA Attention: Mr. Robert H. Gillard Telecopy No.: (403) 944-8256

and

National-Oilwell Canada Ltd. 2300 Bow Valley Square III 255 Fifth Avenue SW Calgary, Alberta T2P 3G6 CANADA Attention: A. B. Moore Telecopy No.: (403) 262-9160

65	TEXAS COMMERCE BANK NATIONAL ASSOCIATION, as U.S. Agent and as a U.S. Lender		
	By: /s/ MONA M. FOCH		
	Mona M. Foch, Vice President		
	Address for Notices:		
U.S. Commitment:	712 Main Street Houston, Texas 77002		
\$15,000,000	Attention: Manager, Structured Finance - Oil Service		
	Telecopy No.: (713) 216-6710		

Canadian Commitment:

\$0

as Canadian Agent and as a Canadian Lender /s/ CHRISTINE CHAN By: -----Name: Christine Chan -----Title: Vice President -----Address for Notices: U.S. Commitment: First Canadian Place 100 King Street West, Suite 6900 Toronto, Ontario M5X 1A4 \$0 Attention: Ms. Christine Chan Telecopy No.: (416) 216-4161 Canadian Commitment:

THE CHASE MANHATTAN BANK OF CANADA,

\$10,000,000

WELLS FARGO BANK (TEXAS), NATIONAL ASSOCIATION, as U.S. Co-Agent and as a U.S. Lender By: /s/ FRANK W. SCHAGEMAN Name: Frank W. Schageman Title: Vice President, Senior Relationship Mgr. Address for Notices: U.S. Commitment: 1000 Louisiana 3rd Floor \$20,000,000 Attention: Mr. Frank Schageman Telecopy No.: (713) 250-7912

Canadian Commitment:

\$0

as Canadian Co-Agent and as a Canadian Lender By: /s/ MIKE JACKSON -----Name: Mike Jackson Title: Vice President and Office Head -----By: /s/ JEFF CEBRYK Name: Jeff Cebryk Title: Relationship Manager ------U.S. Commitment: Address for Notices: \$0 International Banking Division-Loan Accounting 14th Floor 44 King Street West Canadian Commitment: Toronto, Ontario CANADA M5H 1H1 \$15,000,000 Attention: Assistant Manager with a copies to: The Bank of Nova Scotia Corporate Banking Calgary Suite #3820, 700-2nd Street S.W. Calgary, Alberta CANADA T2P 2N7 Attention: Vice President and to: 1100 Louisiana Suite 3000 Houston, Texas 77002

THE BANK OF NOVA SCOTIA,

88

Attention: Mr. Mark Ammerman Telecopy No.: (713) 752-2425

89	THE BANK OF NOVA SCOTIA, as a U.S. Lender			
	By: /s/ F. C. H. ASHBY			
	Name: F. C. H. Ashby			
	Title: Senior Manager - Loan Operations			
U.S. Commitment:				
\$5,000,000	Address for Notices:			
	600 Peachtree Street, Suite 2700 Atlanta, Georgia 30308			
Canadian Commitment:	Attention: Mr. Cleve Bushey			
\$0	with a copy to:			
	1100 Louisiana Suite 3000 Houston, Texas 77002 Attention: Mr. Mark Ammerman Telecopy No.: (713) 752-2425			

THE FUJI BANK, LIMITED HOUSTON AGENCY
By: /s/ KENICHI TATARA
Name: Kenichi Tatara
Title: Vice President & Manager
Address for Notices:
U.S. Commitment:
The Fuji Bank, Limited
1221 McKinney, Suite 410
Houston, Texas 77010
Attention: Mr. Mark E. Polasek
Telecopy No.: (713) 759-0048
Canadian Commitment:

85

\$0

51	THE BANK OF NEW YORK
	By: /s/ ALAN F. LYSTER, JR.
	Name: Alan F. Lyster, Jr.
	Title: Vice President
	Address for Notices:
U.S. Commitment:	One Wall Street 22nd Floor
\$15,000,000	New York, New York 10286 Attention: Mr. Alan Lyster Telecopy No.: (212) 635-6534
Canadian Commitment:	

\$0

32	CORESTATES BANK, N.A.		
	By: /s/ SCOTT HUFFMAN		
	Name: Scott Huffman		
	Title: Vice President		
	Address for Notices:		
U.S. Commitment:	5847 San Felipe, N.A. Suite 1050		
\$15,000,000	Houston, Texas 77057 Attention: Mr. Scott Huffman Telecopy No.: (713) 954-0930		
Canadian Commitment:			

\$0

CREDIT LYONNAIS NEW YORK BRANCH By: /s/ PASCAL POUBELLE Name: Pascal Poubelle Title: Executive Vice President Address for Notices: U.S. Commitment: U.S. Commitment: 1000 Louisiana Suite 5360 \$15,000,000 Houston, Texas 77002 Attention: Mr. Tom Byargeon Telecopy No.: (713) 751-0307 Canadian Commitment:

\$0

5 1,000

9-MOS		6-MOS		3-M0S	3-MOS	
DEC-31	DEC-31-1997		DEC-31-1997		DEC-31-1997	
JAN	1-01-1997	JAN-01	-1997	JAN-01-1997	7	
c c	SEP-30-1997	JUN-3	30-1997	MAR-31-19	MAR-31-1997	
	27,730		47,891		44,385	
	Θ		0		0	
	182,915	10	165,232		147,030	
	2,915		2,955		2,967	
	177,117		152,970		137,499	
3	393,137	371,0	371,024		337,886	
	104,315		96,594		80,876	
	35,816	32	32,434		28,889	
	498,102	468	3,763	415, 429	9	
164,65		141,523		113,596		
	67,596		69,268		56,568	
	0	Θ		0		
	0		Θ		Θ	
	253		253		250	
	248,547		239,333	225	5,735	
498,102		468,763		415,429		
	705,719		440,760		206,670	
7	705,719	440,	760	206,670		
	568,689)	359,082		168,030	
	568,689	35	59,082	168,03	30	
	Θ		0	Θ		
	(395)	-	108	(10)		
2	1,803	3,034		1,452		
	48,975	32,	, 266	15,096		
	18,674		11,496	5,3	397	
30,	211	20,770		9,699		
	Θ		Θ	6	Ð	
	623		Θ	Θ		
	0		0		Θ	
	29,588		20,770		599	
	1.14		.81		38	
	1.14		.81	. 3	38	